



ALTERATIONS IN THE STATES AND CONTENTS OF CONSCIOUSNESS: EMPIRICAL AND THEORETICAL ASPECTS

Valdas Noreika

University of Turku

Faculty of Social Sciences Department of Psychology Centre for Cognitive Neuroscience

Supervised by

Professor Antti Revonsuo Centre for Cognitive Neuroscience Department of Psychology University of Turku Turku, Finland

Professor Heikki Hämäläinen Centre for Cognitive Neuroscience Department of Psychology University of Turku Turku, Finland Research fellow Katja Valli Centre for Cognitive Neuroscience Department of Psychology University of Turku Turku, Finland

Reviewed by

Professor George Mashour Department of Anesthesiology Department of Neurosurgery University of Michigan Ann Arbor, Michigan, USA Professor Mark Blagrove Department of Psychology Swansea University Swansea, Wales, UK

Opponent

Professor George Mashour Department of Anesthesiology Department of Neurosurgery University of Michigan Ann Arbor, Michigan, USA

Cover art: Mikalojus Konstantinas Čiurlionis, Ramybė [Stillness], fragment, 1903/1904

The originality of this thesis has been checked in accordance with the University of Turku quality assurance system using the Turnitin OriginalityCheck service.

ISBN 978-951-29-5865-8 (PRINT) ISBN 978-951-29-5866-5 (PDF) ISSN 0082-6987 Painosalama Oy - Turku, Finland 2014

ABSTRACT

The main purpose of the present doctoral thesis is to investigate subjective experiences and cognitive processes in four different types of altered states of consciousness: naturally occurring dreaming, cognitively induced hypnosis, pharmacologically induced sedation, and pathological psychosis. Both empirical and theoretical research is carried out, resulting in four empirical and four theoretical studies. The thesis begins with a review of the main concepts used in consciousness research, the most influential philosophical and neurobiological theories of subjective experience, the classification of altered states of consciousness, and the main empirical methods used to study consciousness alterations. Next, findings of the original studies are discussed, as follows. Phenomenal consciousness is found to be dissociable from responsiveness, as subjective experiences do occur in unresponsive states, including anaesthetic-induced sedation and natural sleep, as demonstrated by post-awakening subjective reports. Two new tools for the content analysis of subjective experiences and dreams are presented, focusing on the diversity, complexity and dynamics of phenomenal consciousness. In addition, a new experimental paradigm of serial awakenings from non-rapid eye movement sleep is introduced, which enables more rapid sampling of dream reports than has been available in previous studies. It is also suggested that lucid dreaming can be studied using transcranial brain stimulation techniques and systematic analysis of pre-lucid dreaming. For blind judges, dreams of psychotic patients appear to be indistinguishable from waking mentation reports collected from the same patients, which indicates a close resemblance of these states of mind. However, despite phenomenological similarities, dreaming should not be treated as a uniform research model of psychotic or intact consciousness. Contrary to this, there seems to be a multiplicity of routes of how different states of consciousness can be associated. For instance, seemingly identical time perception distortions in different alterations of consciousness may have diverse underlying causes for these distortions. It is also shown that altered states do not necessarily exhibit impaired cognitive processing compared to a baseline waking state of consciousness: a case study of time perception in a hypnotic virtuoso indicates a more consistent perceptual timing under hypnosis than in a waking state. The thesis ends with a brief discussion of the most promising new perspectives for the study of alterations of consciousness.

Keywords: altered state of consciousness, baseline state of consciousness, content analysis, distortions of time perception, dreaming, hypnosis, perception of duration, phenomenal consciousness, psychosis, sedation, sleep, stream of consciousness, subjective experience, wakefulness.

TABLE OF CONTENTS

ACKNOWLEDGEMENTS AND MEMORIES							
LIST OF ORIGINAL PUBLICATIONS							
LIST OF ABBREVIATIONS							
1.	INTR	ITRODUCTION					
2.	CON	ISCIOUSNESS15					
	2.1.	The key concepts and distinctions	15 15 16 16				
		2.1.5. Antered vs. baseline states of consciousness					
	2.2.	Fundamental properties of consciousness	19 19				
	2.3.	The hard problem of subjectivity	21				
	2.4.	 Philosophical theories of the mind-body problem	22 22				
	2.5.	Neurocognitive theories of consciousness	26 26 27 28 29				
3.	ALTE	RED STATES OF CONSCIOUSNESS	31				
		Definitions and classifications					
		Dreaming: a naturally occurring alteration of consciousness					
	3.3.	Hypnosis: a cognitively induced alteration of consciousness					
	3.4. 2 F	Sedation: a pharmacologically induced alteration of consciousness					
	3.5. 3.6.	Psychosis: a pathological alteration of consciousness Unconscious states of mind					
		Interactions between different alterations of consciousness					
	5.7.						

4.	EMP	IRICAL METHODS OF STUDYING ALTERED STATES OF CONSCIOUSNESS	41
	4.1.	Facing the obstacles: introspection, reportability, baseline, and sampling	41
	4.2.	Content analysis of subjective reports	43
	4.3.	Behavioural methods to study alterations of consciousness	44
	4.4.	Neuroimaging of altered states of consciousness	45
5.		S OF THE PRESENT THESIS	47
6.	ORIC	SINAL STUDIES	48
	6.1.	General overview	48
	6.2.	Study I: Consciousness lost and found: Subjective experiences in an unresponsive state (Noreika et al., 2011) 6.2.1. Methods 6.2.2. Results	49
	6.3.	Study II: Early-night serial awakenings as a new paradigm for studies on NREM dreaming (Noreika et al., 2009) 6.3.1. Methods	51 51
	6.4.	Study III: Dream bizarreness and waking thought in schizophrenia (Noreika et al., 2010a) 6.4.1. Methods 6.4.2. Results	54
	6.5.	Study IV: Dreaming and waking experiences in schizophrenia: How should the (dis)continuity hypotheses be approached empirically? (Noreika, 2011) 6.5.1. Literature review	55
	6.6.	Study V: How to integrate dreaming into a general theory of consciousness – A critical review of existing positions and suggestions for future research (Windt & Noreika, 2011)	57
	6.7.	 Study VI: New perspectives for the study of lucid dreaming: From brain stimulation to philosophical theories of self-consciousness (Noreika et al., 2010b) 6.7.1. Literature review 6.7.2. Original proposal. 	60
	6.8.	Study VII: Perception of short time scale intervals in a hypnotic virtuoso (Noreika et al., 2012) 6.8.1. Methods	61 61
	6.9.	Study VIII: Variability of duration perception: From natural and induced alterations to psychiatric disorders (Noreika et al., 2014)	63
		α	

		6.9.1. Literature review	63		
		6.9.2. Original proposal	64		
7.	DISC	CUSSION	66		
	7.1.	Methodological advancements	67		
	7.2.	Theoretical advancements	69		
	7.3.	The future perspectives for studying alterations of consciousness	71		
8.	CON	ICLUSIONS	74		
9.	REFI	ERENCES	75		
ORIGINAL PUBLICATIONS					

This work was mostly carried out at the Centre for Cognitive Neuroscience (CCN), Department of Psychology, University of Turku, Finland, during 2005-2010. Several manuscripts and revisions were submitted during my research visit in 2010-2011 at the Oxford Centre for Human Brain Activity, University of Oxford, UK. Finally, the dissertation manuscript was written in 2012-2013 at the Medical Research Council - Cognition and Brain Sciences Unit, Cambridge, UK, where I currently work. So many places and years have made it challenging to acknowledge all those who have helped me; genuine apologies to those not mentioned in these acknowledgements.

I want to express my sincere gratitude to the supervisors, Prof. Antti Revonsuo, Prof. Heikki Hämäläinen, and Dr. Katja Valli. I met Antti back in 2001 when I was taking part in an exchange program at the University of Skövde, Sweden. Antti was giving a course on the philosophy of cognitive neuroscience and consciousness. Inspired by his interdisciplinary approach, I asked Antti about the possibility of joining his lab and, after a very encouraging reply, I came to Turku, first, for a Master project and later for a PhD research. I was always astonished by Antti's ability to bridge so easily between different discourses of consciousness research: from the analysis of subjective reports, to the neural correlates and the metaphysics of consciousness. Antti, your ideas and guidance laid the foundations of my dissertation, and the intellectual freedom cultivated within your group allowed me to form my own research interests. You will remain my science and philosophy hero for years to come! I want to show my appreciation to Heikki, who approved and supported my unusual research topics. While being a Director of the CCN, Heikki developed an intellectually stimulating environment and high quality laboratory facilities – thank you for all your help! While Antti set the strategic goals of my research and Heikki provided institutional support, it was Katja who literally put my hands on research, including electrode application, scoring of sleep stages, content analysis, academic English and so on. I value you as a distinguished scientist and as a good friend, and I am very grateful for the time spent in the lab and your office, as well as for the whale watching experience we had after the dream conference in Canada!

While working at the University of Turku, I was surrounded by the most welcoming and encouraging people from our Consciousness Research Group. I owe sincere thanks to my close friend Levente Móró – our (sometimes too) long discussions in the shared office continuously challenged my convictions, and I had to change many of them! Leve also helped me numerous times with his technical and programming skills, and collected and analysed data for Study I. I am grateful to Niina Salminen-Vaparanta who introduced me to the world of transcranial magnetic stimulation and visual consciousness. It was so nice when you, your husband Pasi Vaparanta and Levente visited us in the hospital to see our newborn daughter! My warmest thanks go to Dr. Valtteri Arstila, who did not allow me to forget my background in philosophy, forced me to look deeper into assumptions of my research, and helped me to carry out Study VII. I love your cheerful spirit and positivity! I also owe thanks to Dr. Sakari Kallio, who introduced me to the mysterious world of hypnosis and contributed enormously to the Study VII. Dr. Mika Koivisto helped many times with statistics and also gave me a lesson in how to play floorball – I am still learning it! Pilleriin Sikka always made me feel that it is me, not her,

who is the most awesome person. I do confirm here that it is you! Dr. Henry Railo was always very sharp in his scientific comments and occasionally made pretty nice shots on the pool table – thank you for all those moments! I am also grateful to other present or past members of research group: Nils Sandman, Heidi Sederholm, Panu Helle, Jarno Tuominen, Dr. Mirja Johanson, and Dr. Sari Laatu. I very much enjoyed our group meetings, especially the ones at the Villa Rauhala with a Finnish sauna near the sea!

I received enormous support from my co-authors, for which I am deeply grateful. I thank warmly Hetti Lahtela, a Master student, who helped collect sleep data and carried out content analysis of dream reports for Study II. It was my first experimental project with many technical challenges and unexpected situations, such as being unable to open and then close the frozen car doors when we had to drive home in the middle of a sleepless lab night during Finish winter! Dr. Leila Jylhänkangas was another very hard working contributor who collected most of the data for Study I. Your attitude towards interview surveys, data transciption, and content analysis raised the standards for my other studies, thank you for that! Study I would not have been possible without our anaesthesiology colleagues from the Turku University Hospital, namely, Dr. Kimmo Kaskinoro, Dr. Riku Aantaa, and Prof. Harry Scheinin. You introduced me to the astonishing puzzle of the sedated brain! My warmest thanks go also to Dr. Juha Markkula and Katriina Seppälä, who collected and content-analysed data for Study III, which inspired me to continue conducting psychiatrically-oriented research of consciousness and cognition. Dr. Till Wagner helped me to formalize some of the ideas on variability of interval timing, which were laid out in Study VIII. It was a very simple maths for you, but a very big help for me. My special thanks go to Dr. Jennifer Windt, an accomplished philosopher and a close friend, who helped to shape my own theoretical position on many aspects of dreaming and consciousness. After our endless discussions and conference trips, we once made a list of papers to write. Somehow, we managed to complete Studies V and VI of the present thesis, but let's not forget - that is just 20% of our original plan! Our conference trips were often accompanied by Dr. Dr. Nicolas Langlitz, an anthropologist and historian of science, who showed me new perspectives of how to look at science – thank you for that! I want to express my sincere thanks to Dr. Bigna Lenggenhager and Dr. Ahmed Karim, who contributed to Study VI. We did not manage to induce lucid dreaming with transcranial stimulation, but at least we proposed it, and someone else did! I thank warmly Prof. John Wearden for modelling data for Study VII.

I shared the space with many more students and colleagues working at the CCN, including Tiina Kontto, Lisa Svartsjö, Mila Oravecz, Dr. Monika Intaitė, Dr. Victor Vorobyev, Anna Kouvo, Tiina Tuominen, Dr. Riia Kivimäki, Dr. Vladas Valiulis, and Teemu Laine - the CCN environment was so inspiring and relaxed because of you! Teemu also often helped me with technical issues – thank you for this! Special thanks goes to my close friend Dr. Myoung Soo Kwon, who was always there since my Masters project. Often the only foreigners, we understood each other very well at work as well as outside the university. Myoung, I can still feel a fire in my mouth caused by the "mildly spicy" Korean food you cooked us! My sincere thanks go to Maria Ek, Outi Nieminen, Terttu Kauppinen, and Minna Varjonen for providing administrative office support – I would have been completely lost without all of you! I am also grateful to Prof. Pekka Niemi and Prof. Jukka Hyönä, the Heads of the Department, who helped me to move forwards with my studies. My heartfelt thanks go to Dr. Christine Falter, a special friend and colleague, who invited me for a one year research visit to Oxford and also contributed to Studies VII and VIII. Christine introduced me to the research of (ab)normal development, which eventually delayed my PhD submission, but nevertheless it was worth it! Christine, I miss the time we shared the office space and had plenty of time to exchange ideas and I am happy we still continue some projects! I am also very grateful to Prof. Kia Nobre, who made my Oxford visit possible, and Dr. Tiina Parviainen and Dr. Jennifer Swettenham with whom I shared many memorable moments.

I want to show appreciation to Dr. Tristan Bekinschtein who employed me to work in his project on the transitions of consciousness and showed many new ways of how to study subjectivity. Moreover, his witty jokes from the "Doctor Noreika" series kept me moving forwards with my PhD manuscript. Tristan, I am happy to announce that we need a new series of jokes! My warmest thanks go to Tristan's group members and friends: Andrés Canales-Johnson, Dr. Srivas Chennu, Stanimira Georgieva, Moos Peeters, Dr. Paola Finoia, and Dr. Corinne Bareham. You provided an amazing intellectual environment for finalizing my dissertation. Corinne also helped me by translating these acknowledgements from the Lithuanian-English to the British-English.

I want to express my sincere thanks to the reviewers of this thesis, Prof. George Mashour and Prof. Mark Blagrove, as well as anonymous reviewers of the published articles. Their comments and suggestions improved my work tremendously. Thanks also to all those who participated in these studies and gave their precious time and efforts – it is not fun to be awakened eight times from your sleep in just three hours! I also thank Louise Morgan, the main editor of this thesis.

My PhD research was financially supported by the National Doctoral Programme of Psychology, the Signe & Ane Gyllenberg Foundation, the Academy of Finland, the Centre for International Mobility (CIMO), the Oskar Öflund Foundation, the Volkswagen Foundation, and the Turku University Foundation. I am deeply grateful to these institutions.

I wish to express my gratitude to my friends from the church in Turku, Kristina and Dr. Juha Mikkonen, and Kirsi and Tomi Jalkanen. Thank you for reminding me that there is more to the meaning of life than my eyes can see.

The most important comes at the end. I am deeply indebted and grateful to my family, especially my wife Audrone Noreikiene, who was unfortunately left alone during many long evenings and weekends that I spent on my manuscripts. You helped me throughout my doctoral studies, from being my first research participant to choosing a suit for the PhD defence! Our children are the everlasting source of joy. Many thanks to our daughter Emilija Noreikaite, who kept asking impatiently "When will you finish your dissertation?", and to our son Linas Noreika, who is now two months old and already smiles when I say "Your daddy will soon be a doctor!". I owe more than I can express in words to my parents Zita and Antanas Noreikos who supported me in all choices of studies that I made, drifting between philosophy, neurobiology and psychology. I never had to doubt how much you love me. My only regret of carrying out PhD research in Turku was that I could not see you as often as I wanted to. Well, at least, Estonians invented Skype. I wish to express my gratefulness also to my cousin Dr.

Donatas Noreika. Whilst sharing the room during our undergraduate studies, a perfect match occurred between my theory-oriented philosophy and your data-oriented psychology. Our heightened discussions would often turn into shouting, with only one step left before the plates would fly. Science and knowledge definitely deserve that.

Cambridge, 9 September 2014

Valdas Noreika

LIST OF ORIGINAL PUBLICATIONS

This thesis is based on the following publications, which are referred to in the text by their Roman numerals.

Empirical studies

- I Noreika, V., Jylhänkangas, L., Móró, L., Valli, K., Kaskinoro, K., Aantaa, R., Scheinin, H., & Revonsuo, A. (2011a). Consciousness lost and found: Subjective experiences in an unresponsive state. *Brain and Cognition*, 77, 327–334.^a
- II Noreika, V., Valli, K., Lahtela, H., & Revonsuo, A. (2009). Early-night serial awakenings as a new paradigm for studies on NREM dreaming. *International Journal of Psychophysiology*, 74, 14–18.^a
- III Noreika, V., Valli, K., Markkula, J., Seppälä, K., & Revonsuo, A. (2010a). Dream bizarreness and waking thought in schizophrenia. *Psychiatry Research*, *178*, 562–564.^a

Theoretical studies

- IV Noreika, V. (2011). Dreaming and waking experiences in schizophrenia: How should the (dis)continuity hypotheses be approached empirically? *Consciousness and Cognition, 20*, 349–352.^a
- Windt, J. M., & Noreika, V. (2011). How to integrate dreaming into a general theory of consciousness A critical review of existing positions and suggestions for future research. *Consciousness and Cognition, 20,* 1091-1107.^a
- VI Noreika, V., Windt, J. M., Lenggenhager, B., & Karim, A. A. (2010b). New perspectives for the study of lucid dreaming: From brain stimulation to philosophical theories of selfconsciousness. *International Journal of Dream Research*, *3*, 36–45.^b

Empirical study

VII Noreika, V., Falter, C. M., Arstila, V., Wearden, J. H., & Kallio, S. (2012). Perception of short time scale intervals in a hypnotic virtuoso. *International Journal of Clinical and Experimental Hypnosis, 60*, 318–337.^c

Theoretical study

VIII Noreika, V., Falter, C. M., & Wagner, T. (2014). Variability of duration perception: From natural and induced alterations to psychiatric disorders. V. Arstila & D. Lloyd (Eds.), *Subjective Time: The Philosophy, Psychology, and Neuroscience of Temporality* (pp. 529-555). Cambridge, MA: MIT Press.^d

^a Printed with the permission of the Elsevier.

^b Open access journal.

^c Printed with the permission of the Taylor & Francis.

^d Printed with the permission of the MIT Press.

LIST OF ABBREVIATIONS

ASC	altered state of consciousness
BIS index	bispectral index
EEG	electroencephalography
EMG	electromyography
ENSA	early-night serial awakenings
EOG	electrooculography
fMRI	functional magnetic resonance imaging
DLPFC	dorsolateral prefrontal cortex
LOR	loss of responsiveness
LSD	lysergic acid diethylamide
Μ	mean
MEG	magnetoencephalography
NREM sleep	non-rapid eye movement sleep
O-PC-Dreams	Orlinsky's Modified Scale for Perceptual Complexity of Dreams
PCI	perturbational complexity index
PSG	polysomnography
REM sleep	rapid eye movement sleep
ROR	regaining of responsiveness
SCSB	static charge sensitive bed
SD	standard deviation
SEDA-Coding	Subjective Experiences During Anaesthesia Coding System
tDCS	transcranial direct current stimulation
TMS	transcranial magnetic stimulation

1. INTRODUCTION

Human individuals are unique in the momentary and lifelong collection of their contents of consciousness, or subjective experiences, which include – but are not limited to – sensations, feelings, thoughts, and beliefs. Typically, the contents of consciousness are complex and dynamic, with an extremely wide range of possible experiences that continuously adapt to internal and external stimuli as well as to other subjective experiences. While in standard social situations we are relatively successful in guessing subjective experiences of other individuals by drawing inferences from their behaviour and affective display (Ames, 2005), it proves to be very difficult to measure the contents of consciousness with scientific methods. The task becomes even more complicated when a participant undergoes an alteration of consciousness, such as during the transition from waking consciousness to sleep dreaming. The most challenging task is a scientific assessment of subjective experiences of an individual who loses capacity to respond in an altered state of consciousness (ASC), such as during sleep or general anaesthesia. Nevertheless, any general theory of consciousness that happen under different circumstances.

The problem of the detection of consciousness can be addressed with objective methods, such as the functional magnetic resonance imaging (fMRI), with subjective methods, such as analysis of verbal reports and questionnaires, or with a combination of both types of methods. For instance, analysis of neurophysiological data can make it possible to distinguish which of the two competing images participants perceive in the binocular rivalry paradigm (Haynes & Rees, 2005), or to evaluate if an unresponsive vegetative state patient follows mental imageryrelated instructions (Owen et al., 2006). In a recent study, machine-learning-based analysis of fMRI data even enabled prediction above chance level of the visual contents of non-rapid eye movement (NREM) sleep dreaming (Horikawa et al., 2013). Yet, despite an invaluable progress, studies using objective methods are limited by the predictability of the contents of consciousness, i.e., they rely on the recognition of neural activity patterns that emerge in association with the standardized contents of consciousness. For instance, a recent study on dream content detection relied on the neural similarity between relatively simple perceptual experiences in waking and sleep onset (Horikawa et al., 2013), which may not hold for more intense and unpredictable rapid eye movement (REM) sleep dreaming or other unresponsive states of consciousness with complex spontaneously developing subjective experiences. Furthermore, neuroimaging-driven prediction is not explanation, while the ultimate goal of the science of consciousness is the explanation of how subjective experiences arise from brain activity.

Compared to the objective methods, content analysis of subjective reports enables a very detailed assessment of naturally and spontaneously occurring subjective experiences, including, but not limited to, such diverse contents of consciousness as Self and body, visual experiences, perception of sound, temporal and spatial distortions, and so on. Arguably, a detailed mapping of human phenomenology in different states of consciousness will eventually provide a solid ground for research into the neural basis of consciousness. While subjective methods alone are not able to provide sufficient insight into the neural mechanisms

of consciousness, a combination of both objective and subjective methods will probably become one of the most promising avenues of consciousness research in the near future. Nevertheless, it is not straightforward to combine subjective and objective methods, and they are still being developed largely separately. Likewise, this dissertation will focus mostly on the application and development of subjective methods, which will be used to describe the phenomenal contents of ASCs.

The main goal of this doctoral thesis was to investigate subjective experiences and cognitive processes in four different alterations of consciousness: dreaming, hypnosis, sedation, and psychosis. Individual studies focused on the development of new experimental paradigms and content analysis methods, systematic comparison of different alterations of consciousness, and contribution to the theoretical issues underlying research of ASCs. In the following sections, after reviewing the main concepts used in consciousness research, the most influential philosophical and neurobiological theories of subjective experience, the classification of ASCs, and the main empirical methods to study consciousness alterations, each of the original studies will be presented in detail.

2. CONSCIOUSNESS

Consciousness science is a multidisciplinary field in the intersection between philosophy, experimental psychology, neuropsychology, and cognitive neuroscience. Given a large number of competing theories and approaches, it is natural that many of the key concepts used in consciousness research, such as *awareness, access consciousness* or *unconsciousness*, are often used in different and partly controversial meanings. To minimize conceptual confusion, these terms should be carefully defined, searching for the most accurate match between our semantics and the natural kinds of human mind.

2.1. The key concepts and distinctions

2.1.1. Phenomenal consciousness

When scientists and philosophers discuss the hard problem of consciousness (Chalmers, 1995; see Section 2.4), they primarily mean *phenomenal consciousness*, which is the most fundamental term in contemporary research of subjective experiences. The concept of phenomenal consciousness refers to the presence of a subjective aspect in a wide range of mental functions, including perception, emotion, memory, attention, language, decision making, and control of motor output (Block, 1995; Revonsuo, 2006). When we say a word, touch a tablecloth or remember what we have seen during the last trip abroad, these acts are usually accompanied by internal experiences, which are subjective and available only to us. As Nagel (1974) phrased, an organism is conscious if 'there is something it is like to *be* that organism' (p. 436), that is, if it has experiences that bear distinctive subjective qualities – often referred to as *qualia* (Lewis, 1929) – for the organism (see also Section 2.3).

2.1.2. Sensory awareness

Phenomenal consciousness is sometimes equated with awareness, however, the latter concept is narrower and refers to a special type of phenomenal consciousness: the presence of sensory experiences triggered by real physical stimuli of which we thereby become aware. In particular, each of the classical sensory systems, i.e., vision, hearing, olfaction, gustation and somatosensory system, has not only unique receptive fields, cellular receptors, central pathways and cortical networks (Møller, 2003), but also unique sensory qualities of experience. We not only detect a leaf falling from a tree, but also have a subjective awareness of that leaf, awareness that does not seem necessary for the detection itself. Naturally, phenomenal consciousness within each sensory system is often conceptualized as visual awareness (Koivisto et al., 2006), auditory awareness (Gutschalk et al., 2008), olfactory awareness (Li et al., 2010), gustatory awareness (Verhagen, 2007), or somatosensory awareness (Schubert et al., 2006). Notably, phenomenal consciousness as a broader term also includes, for instance, internally generated hallucinatory experiences, or subjective experiences related to non-sensory mental functions, such as semantic processing. That is, awareness is a type of phenomenal consciousness triggered by external physical stimuli or internal physiological processes in a case of interoceptive awareness (Critchley et al., 2004),

whereas *phenomenal consciousness* refers to the subjective experience itself, regardless of what triggered it, external or internal stimuli, thoughts, or memories.

2.1.3. Access consciousness, reflective consciousness, cognitive accessibility

Subjective experiences can be attended to and information about them distributed within a broader neural network for further information processing, leading to various cognitive and behavioural effects. This implicates that phenomenal consciousness can be described functionally. Functional interactions between phenomenal consciousness and broader cognitive systems guiding rational behaviour have been termed as access consciousness (Block, 1995). Block (1995) argued for a double dissociation between phenomenal consciousness and access consciousness, i.e., subjective experiences might exist without being broadcasted to the executive processes, whereas functional causal roles of access consciousness may be achieved without the presence of subjective experiences. Revonsuo (1995a, 2006) pointed out that such a definition of access consciousness allows a subjectively unconscious person to be called conscious, leading to a conceptual contradiction. Instead, a theoretically more coherent approach may be a separation of two levels of consciousness: the first fundamental level would be phenomenal consciousness and the second higher level would consist of cognitive operations over selected subjective experiences held in working memory (Revonsuo, 2006). Farthing (1992) termed the second higher-order level as reflective consciousness, although this term may be somewhat too narrow, as the concept of reflective consciousness is sometimes described in terms of abstract thinking and verbalization of subjective experiences (Bartolomeo et al., 2008), whereas cognitive operations over phenomenal consciousness also include more basic cognitive processes that are not necessarily verbalized, such as selective attention or difference detection (Revonsuo, 2006). In more recent studies, Block (2007, 2011) toned down the arguments regarding dissociation between phenomenal and access consciousness, and instead of these terms started using the concepts of phenomenology and cognitive accessibility, the latter of which is defined as the broadcasting of subjective experiences within the central processing system. The concepts of phenomenology and cognitive accessibility (Block, 2007) seem to be compatible with the distinction between phenomenal and reflective consciousness (Farthing, 1992; Revonsuo, 2006).

2.1.4. Contents vs. state of consciousness

Subjective experiences are sometimes termed as *phenomenal contents of consciousness*, which refers to the variety of experiences that are contained or occur in phenomenal consciousness (Revonsuo, 2006). Most of the phenomenal contents of consciousness are also *representational contents*, i.e., they seem to be representing specific aspects of the physical world (Chalmers, 2000). Even though not all subjective experiences have veridical physical referent, for example, some bizarre dream experiences or hallucinations may be unrelated to the actual physical world, they still typically do appear for an experiential subject as representing external reality. The totality of phenomenal contents forms the *state of consciousness*, i.e., an overall pattern of subjective psychological functioning (Tart, 1972). Alongside a theoretical distinction between the presence of specific contents of consciousness and the presence of an overall state of consciousness, two independent neurophysiological

research programs are currently developing in the empirical science of consciousness. Research on the phenomenal contents of consciousness focuses on the neural mechanisms of specific subjective experiences, for example, detection of masked visual targets (Mathewson et al., 2009), while ignoring other simultaneously occurring contents of consciousness. By contrast, neuroimaging research of the state of consciousness typically contrasts the conscious state with the loss or the absence of consciousness in conditions such as regaining of consciousness after its loss due to the administration of anaesthetic agents (Långsjö et al., 2012).

2.1.5. Altered vs. baseline states of consciousness

Compared to the standard waking state of consciousness, which could be regarded as a *baseline state of consciousness*, human consciousness may undergo various neurocognitive alterations, such as those observed in dreaming, hypnosis or meditation. Such alterations are often referred to as *altered states of consciousness* (ASCs) (Tart, 1975), and they are typically individuated by specific changes in their contents (Chalmers, 2000) and/or their induction techniques (Vaitl et al., 2005) (see Section 3.1). Complementing the classical neurological approach of studying abnormal in order to understand normal, ASCs can be used as important contrast conditions for studying human consciousness and the breakdown of its fundamental properties (Hobson, 2001; Revonsuo, 2006).

2.1.6. Conscious, unconscious, and nonconscious processes and states

Phenomenal contents do change rapidly, and a stimulus of which we were aware a couple of seconds ago may not be a part of the current contents of consciousness. Nevertheless, some of the dissipated contents may re-enter phenomenal consciousness after, for instance, episodic memory retrieval (Düzel et al., 1997). By contrast, some other neural processes are never associated with the phenomenal contents of consciousness, for example, the early latency (55 ms) of visual evoked potential C1 (Di Russo et al., 2002) indicates that we are not directly aware of neural processes underlying its generation, as visual awareness-related electroencephalography (EEG) potentials occur in the time latencies above 100 ms (Koivisto & Revonsuo, 2010). Neural activity patterns that are not associated with conscious experiences but could, under some circumstances, produce phenomenal consciousness can be termed as unconscious contents, whereas nonconscious processes are those deprived of capacity to reach phenomenal consciousness (Searle, 1992). A similar distinction can be drawn between conscious, unconscious and nonconscious states of mind. A conscious state would contain at least one subjective experience, whereas unconscious state, such as deep NREM sleep, may have no qualia, but would maintain a potential to become conscious if some dream experiences would emerge (see Section 3.6). Nonconscious states of the brain/mind are those that have no capacity to support phenomenal consciousness, such as irreversible coma (Silverman et al., 1970). By definition, there are no unconscious contents in the nonconscious state of the brain/mind.

2.2. Fundamental properties of consciousness

Given that subjective experiences accompany a very wide range of sensory, cognitive and behavioural functions, most of which have a large number of sub-functions, phenomenal consciousness turns out to be a very diverse and generic brain process. For instance, not only has each modality of sensory awareness exclusive experiential qualities, but each sensory system codes a large number of unique stimuli features that are associated with unique qualia. Thus, for example, visual awareness is comprised of experiences of colour (Vul & MacLeod, 2006), motion (Eagleman & Sejnowski, 2000), depth (Brouwer et al., 2005), and location (Huang et al., 2007). While simple sensory qualities may accompany the realm of sensory awareness, more complicated experiences of, for instance, understanding and remembering may be associated with complex cognitive functions, such as semantic processing (Kiefer & Spitzer, 2000) or autobiographic memory (Piolino et al., 2003). In the latter cases, subjective experiences do not reflect physical properties of stimuli but instead they point to the higher order phenomenal qualities such as subjective certainty (Tulving, 1985) or an awareness of semantic ambiguities (Zipke et al., 2009).

When focusing on a specific type of subjective experiences in a restricted experimental setup, phenomenal consciousness seems to be a constellation of a vast number of subjective experiences that emerge, develop and vanish in isolation from other experiences. Thus, for example, research on participant's awareness of changing ambiguous Necker cubes (Intaite et al., 2013), while being an interesting visual phenomenon in itself, seems to have no connection to other subjective experiences that may occur during the same experiment, including feelings of hunger, boredom, or guilt after missing a stimulus, a tingling sensation in a shoulder, hearing of the background noise, or seeing objects in the peripheral visual field. However, introspection of our own experiences reveals that phenomenal consciousness, despite its richness and dynamics, is normally unified and coherent, forming a consistent complex model of external and internal environment (Revonsuo, 2006).

James (1890) argued that there is continuity in the diversity of phenomenal experiences, which can be conceptualized as the *stream of consciousness*, pointing to the constantly changing, yet coherent and continuous, contents of phenomenal consciousness. Continuity of the stream of consciousness depends on the temporal and spatial unity or binding of individual subjective experiences (Dainton, 2006; Revonsuo, 2006). Even though each distinct experience may occur at different points in the phenomenal space and time, they are joined together into a coherent flow of a single unified stream of consciousness. That is, we do not have an experience of unrelated sensations that jump in time and space, although this may take place in certain psychiatric or neurological conditions such as schizophrenia (Vogeley & Kupke, 2007) or akinetopsia (Zeki, 1991), but an experience of spatially and temporally bound qualities that form a single stream of consciousness. The concept of *consciousness-related binding* refers to the experienced phenomenal unity of the contents of consciousness, and it should be demarcated from the *stimulus-related binding*, which refers to the coherence of neural representations of external physical objects that may or may not become the phenomenal contents of consciousness (Revonsuo, 1999, 2006).

In addition to the temporal and spatial binding of phenomenal contents, they are also bound together across different mental domains, including sensory, executive, semantic and motor functions. Such *multifunctional integration* of subjective experiences brings together different types of qualia into a coherent perception of the world, for example, while talking with a friend, we may experience visual awareness of her face, auditory awareness of verbal sounds, and semantic awareness of the meaning of words, which may further surprise us or trigger some autobiographical memories.

2.2.1. Temporal binding

Temporal binding of the phenomenal contents of consciousness integrates subjective experiences over time. In phenomenal consciousness, temporal binding can be studied in terms of perception of duration and simultaneity (Kiverstein, 2010). Perception of duration is typically associated with integration of the same set of qualia over time, for example, we may feel that the experience of a bee whose flying trajectory we just followed lasted for a certain interval of time, e.g., 3 seconds. Possibly, the same neuronal populations representing the bee continued oscillating and synchronizing their activity during this period of time (Engel & Singer, 2001), and subjectively we experienced it as a single temporally extended percept of the bee. Interestingly, it has been shown that perceptual or attentional sampling of sensory information is not completely continuous, but instead it oscillates in the theta (4–8 Hz) or alpha (8–12 Hz) frequency range (Busch et al., 2009; Mathewson et al., 2009; Ng et al., 2012). Phenomenally we are not aware of such temporal snapshots underlying our perception, and their mechanisms operate at the nonconscious level of processing (see Section 2.1.6).

While perception of duration is often associated with awareness of a single object, perception of simultaneity points to the integration of several different sets of qualia over time, i.e., we may experience the flying bee and the still honeybush at the same time. Contrary to simultaneity, perception of asynchrony or temporal order reveals a temporal succession of different phenomenal contents of consciousness. In these cases, the temporal binding unifies the change from one subjective experience to another over a wider interval of time. This aspect of temporal binding likely involves neurocognitive mechanisms of phenomenal consciousness: the just-experienced contents of consciousness are kept in the iconic memory, while we experience the present contents, anticipating the subsequent ones. The tripartite temporal structure of consciousness is sometimes referred to as the *specious present* – the temporal binding of phenomenal contents of consciousness depends on the autobiographic memory, which integrates phenomenal experiences over many days and years, constituting the basis for *self-consciousness*, also called *autonoetic consciousness* (Piolino et al., 2003; Tulving, 1985; Wheeler et al., 1997).

2.2.2. Spatial binding

Spatial binding of the phenomenal contents of consciousness refers to the integration of subjective experiences in space. Each stimulus-driven experience has a specific location in relation to other experiences, which is studied under the concept of *location binding* (Treisman, 1996). Furthermore, experiences typically have a defined location with respect to our own body that forms the centre of phenomenal space (Revonsuo, 2006). The egocentric reference point is typically located in the phenomenal head or chest and it can be used to

identify the location of other phenomenal contents of consciousness (Revonsuo, 2006). Phenomenal spatiality, while typically studied in the visual domain (Clark, 2000), is inherent not only for every sensory modality but also for emotions and feelings (Nummenmaa et al., 2014). While visual and auditory experiences are typically localized in the phenomenal space outside our bodily space, tactile or emotional stimuli are primarily located within the phenomenal body. Notably, various experimental manipulations can reallocate experiences from the external phenomenal space to the internal bodily space or vice versa. For example, identical auditory stimuli presented binaurally through the headphones might be subjectively felt as localized inside the head (Hartmann & Wittenberg, 1996), whereas presentation of conflicting visual-somatosensory input in virtual reality may shift the self-centre outside of the physical body (Lenggenhager et al., 2007). Nevertheless, spatial unity of experiences is maintained even in such unexpected and artificial distortions, indicating the robustness of consciousness-related spatial binding.

2.2.3. Multifunctional integration

In addition to the temporal and spatial binding, phenomenal consciousness is integrated across different perceptual, cognitive and behavioural functions, leading to the multifunctional unity of the experience of the self in the world. Such multifunctional integration takes place along several different axes, including perception-perception, perception-language, and perception-action unity. For instance, integration of phenomenal contents across different sensory systems (e.g., seeing a face and hearing its speech), brings together very different perceptual qualia under one single stream of consciousness instead of forming two separate auditory and visual streams of phenomenal contents. Furthermore, conscious awareness itself may have a crucial role in such *multisensory interactions*, as in some cases only consciously perceived information from the dominating sensory modality is transferred to another modality (Palmer & Ramsey, 2012). Multisensory interactions between different sensory modalities may also underlie generation of spatial awareness (Clavagnier et al., 2004) and bodily self-consciousness (Blanke, 2012). Another type of multifunctional integration takes place in the form of *perceptual-semantic binding*, which enables the experience of the immediate meaningfulness of phenomenal contents of consciousness (Revonsuo, 2006). Typically, perceptual experiences are automatically bound together with semantic contents, leading to the spontaneously occurring recognition and linguistic conceptualization of these experiences.

The first-person perspective suggests that 'we see the very world we act in and we act in the world we see' (Clark, 2009, p. 1460). If so, the integration of perception and action along a single unified flow of body-world interaction might be another fundamental aspect of consciousness, which has even been suggested to be the ontological basis of consciousness (O'Regan & Noë, 2001). Indeed, embodied vision seems to be dominating over subjective experiences during wakefulness, when we perceive and interact with the world around us. However, bodily and motor processes do not seem to be necessary for visual experiences to emerge, as, for instance, people may have rather intense visual dreams during REM sleep, when their muscular system is atonic. Furthermore, the neuropsychological and neuroimaging evidence suggests that conscious 'vision for perception' in the ventral visual processing stream and non-conscious 'vision for action' in the dorsal stream can be dissociated from each other

(Goodale & Milner, 1992; Milner & Goodale, 2008). Thus, perception-action integration seems to be a typical, but not a fundamental property of consciousness.

2.3. The hard problem of subjectivity

Given that phenomenal contents of consciousness are subjective and directly available only to the experiencing self, explaining the ontology of consciousness poses a unique problem for science: it violates a universal assumption that the phenomena studied by science should be equally available to all observers, and no observer should have a privileged access to a phenomenon. Even though representational contents consciousness can be successfully mapped using neuroimaging techniques, specific blood-flow or electrophysiological activation patterns do not provide access to the subjective experiential qualities (see Section 4.4). Consequently, there seems to be an unbridgeable explanatory gap between the objective neurophysiological data and the subjective contents of consciousness: even if we are certain that the existence of consciousness is dependent on the brain, we cannot understand how brain activities cause or constitute consciousness (Levine, 1983; McGinn, 1989). Reductive analyses of consciousness, i.e., suggestions that consciousness is identical with some types of brain processes or functional descriptions, always seem to be logically consistent with the absence of phenomenal subjective experience despite the presence of these processes (Chalmers, 1996; Nagel, 1974). These intuitions have been illustrated by several famous thought-experiments (Jackson, 1986; Nagel, 1974). Nagel (1974) argued that even if objective science would acquire all physical facts about humans or other mammals, we would still miss information about their subjective point of view. For instance, even if we would know all neurophysiological and behavioural facts about a bat, including its sensory echolocation system, we would still have no idea of 'what is it like to be a bat', for example, what are its subjective experiences of flying in the complete darkness between the trees. Likewise, the same argument can extended to other conscious organisms, including humans: their subjective point of view is always left out in an objective scientific description of behaviour and neurophysiology (Nagel, 1974). Jackson (1986) argued that a neuroscientist who is congenitally colour blind but knows all objective facts about the psychophysics and physiology of colour perception would still learn something new if one day her colour vision would be cured, which seems to indicate that there might be more to consciousness than physical facts about its mechanisms.

Many of the consciousness-related research problems, such as the ability of a system to access its own internal states or the difference between wakefulness and sleep, seem to be relatively *easy*, i.e., they seem to be directly susceptible to the standard methods of cognitive neuroscience (Chalmers, 1995). Even though some of the easy problems of consciousness are not yet solved, we have a clear idea of what would be their scientific explanation: discovery and description of computational and/or neural mechanisms would solve each of these problems. Contrary to this, the problem of the material basis of conscious experience is a problem of a different type: it is a *hard problem*, which seems to resist any of the empirical methods available to the contemporary science (Chalmers, 1995). We do not have any idea of how and why subjective sensations or emotions arise from a physical basis, and yet they do.

Even worse, we cannot even imagine how this problem could be solved in terms of neurophysiological or neurochemical processes.

The *hard problem* can be tackled by a number of different philosophical theories of the mind-body relationship, and some of these theories, such as dualism or panpsychism, do not seem to have the hard problem of consciousness at all. Arguably, it applies only to materialistic theories that aim to explain subjectivity in terms of objective physical or biological processes. In Section 2.4 following, some of the most influential classical and contemporary metaphysical theories of the mind-body relationship will be presented and evaluated, with a special focus on the biological approaches to the consciousness-brain interaction.

2.4. Philosophical theories of the mind-body problem

2.4.1. Classical solutions to the mind-body problem

Throughout centuries of theoretical research, controversies between dualism versus monism, idealism versus realism, and rationalism versus empiricism, placed the mind and consciousness at the top of philosophical investigations (Crane & Patterson, 2000; Heinämaa et al., 2007). Dualistic approaches to the mind-body problem assume that the mind (or certain aspects of it) and the physical world are ontologically separate and cannot be reduced to each other. Proponents of substance dualism, most notably Descartes (1641), suggested that consciousness and the brain are different substances, which can exist independently of each other. Property dualists supposed the existence of one substance with two fundamentally different aspects, i.e., mentality and materiality, where mental properties do not exist in the absence of physical properties (e.g., Mill, 1843). Contrary to dualistic positions, monism assumed the existence of only one independent substance. For instance, La Mettrie (1748) defended a materialistic position (also called physical monism), arguing that consciousness has purely physical causes and there is no necessity to assume the existence of a separate 'soulsubstance'. On the opposite extreme of monistic positions lied *idealism* (also called *mental* monism), which states that consciousness is the only ontologically real substance, whereas all the other material substrates are derivable from it (Berkeley, 1710), i.e., brain and other physical phenomena are real, but only as the subjective contents of consciousness. Further developments of classical solutions to the mind-body problem showed that there is no 'pure dualism' or 'pure idealism', but instead there are many different versions of them (Morton, 2010), some of which continue being considered and developed. For instance, findings in modern physics and neuroscience may inspire brain and physics-oriented refinement of property dualism (Popper & Eccles, 1977) and even idealism (Bolender, 2001).

2.4.2. Contemporary theories: from eliminativism to panpsychism

Contemporary ontological theories of consciousness provide a wide range of solutions of how consciousness and the brain might be related, which indicates the lack of a single widely convincing ontological theory of consciousness. For instance, *reductionism*, one of the materialistic theories of the mind-body relationship, suggests that each type of mental state or

process is identical to some type of neural state or process (Lewis, 1966; Smart, 1959). According to the *type-type identity* theory, which is a version of reductionism, there is no difference between a particular neural level in the brain and consciousness. Yet, the identity relationship implies that certain neural processes and consciousness are sharing all properties, whereas conscious experiences have subjective qualitative aspects that the neural mechanisms do not seem to have. If so, consciousness cannot be identical or reducible to any of the objectively observable neural states or processes. A possible answer to this argument is provided by *eliminativism*, which states that subjective properties do not physically exist. The most fundamental is the existence of the brain and the whole body, while the subjective aspect of perception is nothing more than an epiphenomenal illusion; a misrepresentation about functioning of our brains without any causal power (Churchland, 1999; Dennett, 1978, 1991). Yet, even if one agrees that consciousness is illusion, there is something left to be explained: how the brain can produce this subjective illusion.

Not all ontologies of mind that take the physical world seriously seek to reduce or eliminate phenomenal consciousness. For instance, embodiment theories assume that subjective experiences depend on the physical world, but processes and entities which are necessary for consciousness are not limited to the brain: the contents of consciousness are ontologically dependant on causal relations with peripheral body as well as with external physical objects (Noë & Thompson, 2004; Thompson & Varela, 2001). The embodiment approach seems to be plausible in the case of visual perception, as one of the prerequisites of vision is external physical stimulation of retina, yet embodiment theories do not provide a satisfactory explanation of how internal subjective experiences are generated in the absence of behavioural embodied interactions with the world, such as during mental imagery or dreaming. While the embodiment approach extends physical foundations of consciousness from the purely brain-based mechanisms to the peripheral body and immediate physical stimuli, panpsychism goes further and proposes that consciousness may be just about everywhere. For instance, Chalmers (1996) argues that information implemented in any physical system, such as the brain, flowers or microwaves, is conscious, i.e., it is a natural law that information realized physically is phenomenally conscious (see Tononi 2001, 2012; Section 2.5.5). Several related proposals have been put forward by researchers arguing that consciousness is a property of a matter at the level of fundamental physical forces, such as those described by quantum mechanics (Bohm, 1990; Jibu & Yasue, 1995). However, physicsand information-oriented theories have difficulties in explaining why consciousness is so tightly related to certain processes in the brain, as demonstrated by numerous cognitive neuroscience studies (but see Hameroff, 2007).

Functionalism suggests that the ontology of consciousness is determined by abstract causal interactions between perceptual, cognitive and behavioural processes. In fact, functionalistic theories propose that any mental state could be defined by sensory inputs, inner informational processes and behavioural outputs (Fodor, 1975; Wright, 1973). Descriptions of such functional interactions have been successfully used in cognitive sciences to explain human rationality and decision making (Clark, 2001), which fuelled expectations that consciousness could be explained likewise (Dennett, 1978). Importantly, the abstract level of functional explanations implies that biological brains are not necessary for the generation of consciousness, as the same abstract functions and algorithms can be implemented in different

physical substrates. The possibility of multiple realizations of functional interactions seems to provide theoretical grounds for building artificial intelligence and consciousness (Aleksander, 2005). Yet, while being a promising theory for identifying functional roles of consciousness, functionalism does not escape from difficulties in explaining qualitative aspects of consciousness (Block, 1980), as it remains unclear why some, but not all, functional networks are conscious. Difficulties in conceptualizing a convincing ontological theory of consciousness have even led to a position called *mysterianism*, also known as *strong emergent materialism*: even if consciousness is a material phenomenon, the human brain may simply be too limited and incapable of explaining the mind-body relationship, just as, for instance, dog brains are incapable of understanding modern physics (McGinn, 1989, 1991).

2.4.3. Biological approaches to consciousness

The majority of the empirical researchers of consciousness tend to follow assumptions of biological naturalism, which is a scientific common sense position that consciousness is a qualitative, subjective, unified and (usually) representational high-level brain function (Searle, 1992, 2007). Importantly, even though consciousness is thought to be caused by the lower level brain processes, it also seems to be irreducible to them and having its own causal efficacy to influence cognition and behaviour (Searle, 2007). Consistently with these intuitions, biological realism assumes that phenomenal consciousness is a real biological phenomenon that resides within the confines of the brain and cannot be reduced to the fundamental laws of physics (Revonsuo, 2006). Given that being real implies having its own causal powers (Kim, 1992), consciousness is assumed to be subordinated to causal interactions with other neural processes in the brain and, through them, with the rest of the body and the physical world. Methodologically, biological realism proposes that subjective experiences as biologically real phenomena should be studied by the sciences of biology. In particular, material basis of consciousness should be studied within the spatial and temporal scales that are common for neurosciences. Further, biological realism assumes that we are not cognitively closed to the understanding of consciousness, but instead biological sciences, especially cognitive neuroscience, are (or will be) able to explain it (Revonsuo, 2006). Thus, metaphysically, biological realism represents the weak emergent materialism, which predicts that the complete understanding of the brain will explain how consciousness emerges from the brain processes (Revonsuo, 2010). Contrary to this, the strong emergent materialism assumes that the neural basis of consciousness will remain unresolved even when all facts about the human brain will become known to the scientific community (see Section 2.4.2).

While explanation in physics typically involves description of universal natural laws through which an explanandum, i.e., the target of explanation, can be reduced to physical processes at the smaller spatial and temporal scales, explanation in biological sciences involves description of multilevel mechanisms whose causal interactions are too complex to be described by universal laws (Bechtel & Richardson, 1993; Craver, 2007; Craver & Darden, 2001). Instead, biologists, including cognitive neuroscientists, develop constitutive, contextual and etiological explanations of their explanatory target (Revonsuo, 2006). Constitutive explanations reveal the lower levels of a phenomenon under investigation, for example, neuronal long-term potentiation can be described as an emergent outcome of the lower-level NMDA receptor activation (Craver & Darden, 2001). Contextual explanations point to the functional roles of

explanandum in interaction with the higher levels of biological organization. Etiological explanations describe biological processes that are capable of modulating the explanandum, but cannot constitute it, such as abnormal developmental pathways. Consciousness is assumed to reside in the largely unknown phenomenal level of organization in the brain (Revonsuo, 2006), yet it is expected that descriptions of interactions between the cognitive, phenomenal and neural level of organization will eventually solve the mind-body problem (Bechtel & Mundale, 1999).

However, despite continuous development of a biological research programme during the previous decades and its relative success in identifying some neural correlates of consciousness (see Section 2.5), the hard problem of phenomenal consciousness (see Section 2.3) remains unsolved. Thus, even though cognitive neuroscience and neuropsychology already yielded a number of fascinating discoveries and neurocognitive theories (see Section 2.5), it remains possible that some other approaches to consciousness than the one defended by biological realism might prove to be more accurate. For instance, it might turn out that phenomenal consciousness emerges as a product of multi-level functional interactions between biological entities and processes, and that these interactions can be replicated in artificially designed systems. This would contradict the assumptions of biological naturalism and realism, which are regarded as an alternative to functionalism (Revonsuo, 2006; Searle, 1984). Likewise, information integration and differentiation theory of consciousness (Tononi, 2012; see Section 2.5.5) may loosen its ties with neuroscience, if consciousness-generating informational complexity would be detected or designed in non-biological systems. As long as the uncertainty over the neural basis of consciousness continues, it is suggested in this doctoral thesis to commit to a less ambitious but more open and pragmatic approach to consciousness than the one defended by the biological realism, at the same time following the main methodological suggestions of biological realism. This methodological approach will be called as neurobiological pragmatism.

In the philosophy of pragmatism, scientific theories are regarded as instruments that help to facilitate empirical research and increase the understanding of explanandum (James, 1907). Theories are evaluated depending on the success of their application; when scientific progress diminishes, the existing theories and methodologies are replaced by new instruments that are more likely to facilitate further development of science. Given that currently neuroscience provides the most efficient tools for the advancement of the science of consciousness, neurobiological pragmatism would call for directing financial and human resources to the cognitive neuroscience studies of consciousness. Following methodologies outlined by biological realism (Revonsuo, 2006, see also Craver, 2007), neurobiological pragmatism would seek for a multilevel explanation of neural mechanisms of consciousness. However, biological realism assumes that a complete understanding of the neurobiology of brain would naturally lead to the explanation of consciousness (Revonsuo 2006, 2010), which excludes a possibility that consciousness might be realized as a non-biological phenomenon. Contrary to this, neurobiological pragmatism would propose that consciousness is most likely to be explained by neurosciences, however, biological explanations may not be sufficient for solving the hard problem of consciousness, and neurobiological pragmatism may eventually be replaced by 'informational pragmatism', 'physical pragmatism', or some other set of methodologies. Nevertheless, steps towards exclusively non-biological methodologies should be carried out only when the potential of neurobiological paradigms is sufficiently exploited. More likely, instead of abandoning biological approach, it may merge in future with other approaches, leading to a 'multidisciplinary pragmatism'. Notably, neurosciences are already receiving substantial contributions from other sciences. Even contemporary analyses of fMRI, EEG or magnetoencephalography (MEG) correlates of subjective experiences, as well as computational neuroscience of consciousness, are hardly limited to classical biological methods, but instead they require collaboration between neuroscience, psychology, physics, signal processing and mathematics.

Regarding metaphysical commitment, neurobiological pragmatism would remain open to different possibilities of the mind-body relationship, known here as a weak ontological pluralism. In a strong reading of ontological pluralism, it suggests that there are different kinds or modes of being, which allows co-existence between different ontological assumptions, as proposed in logic (Turner, 2010, 2012), but also discussed in the context of epistemology and metaphysics (Northoff, 2004). In the weak reading, ontological pluralism would suggest that different ontological assumptions may co-exist, but also that ontological monism may be more accurate. For instance, a biological approach may eventually require its extension towards physical accounts of consciousness, panpsychism, property dualism or dual aspect monism, but also a weak emergent materialism may turn out to be true. Until science and philosophy provide conclusive evidence and arguments, it is too early and premature to commit to a single ontology of consciousness, as this may hinder other more accurate, yet currently neglected possibilities. Importantly, none of the discussed philosophical theories of the mindbody relationship neglects the importance of biological research, as even in substance dualism non-material entities are thought to be able to cause changes in the brain (Descartes, 1641), which subsequently can be studied empirically. Thus, successful implementation of the methodology of biological realism may lead to, but currently does not require, the acceptance of emergent materialism and is compatible with metaphysical indeterminacy. While it is feasible to focus on the neurobiology of consciousness, we do not know yet where we will end up in trying to explain the hard problem of consciousness. Some of the currently most influential neurocognitive theories of consciousness are detailed in the following Section 2.5.

2.5. Neurocognitive theories of consciousness

2.5.1. Historical sources of empirical consciousness research

Beginnings of the modern empirical research of consciousness can be traced back to the behavioural psychophysics experiments conducted by Weber (1834) and Fechner (1860), who studied the relationship between the physical intensity of various stimuli and the threshold of 'just noticeable difference' in subjective sensation of these stimuli (for review, see Frith & Rees, 2007). Donders (1868) developed an experimental paradigm to compare simple and two-choice reaction time tasks, which allowed differentiating mental and motor processes preceding motor responses. Duration of mental events was further investigated, although more introspectively, by Wundt (1862), who established the first psychology laboratory in Leipzig in 1879. The first empirical studies of dream content were also carried out in the second half of the nineteenth century (for review, see Noreika & Windt, 2008). For instance,

Maury (1861) investigated the influence of external stimuli on dreams, whereas Calkins (1893) proposed a statistical analysis of dream reports. Wundt's student Titchener (1902) formalized introspection techniques and analysed consciousness by systematically breaking it down into elementary feelings and sensations, which led to the structuralistic theory of mind. James (1890) took a different approach and instead of searching for the atoms of mind emphasized the continuity of conscious experiences that flow as a single stream of mental states (see Section 2.2). Despite important theoretical advances and development of new experimental paradigms, most of the nineteenth century consciousness researchers relied on introspection, which was critically rejected by Watson (1913) and other behaviourists who argued that the only scientifically rigorous method to study the human mind is a search for systematic relationships between external stimuli and behavioural responses.

The detailed critique of the methodology of introspectionism was soon accepted as the general denial of the existence of consciousness itself. During domination of behaviourism in the first half of the twentieth century, the topic of consciousness was removed from most of the scientifically oriented textbooks or university programmes of psychology. Even though alternative approaches to the study of mind did not completely cease, for example, research of principles of visual grouping led to the description of Gestalt laws of vision (Wertheimer, 1923), they were marginalized for several decades until the development of information theory (Shannon & Weaver, 1949) enabled the mathematical account of mental processes and triggered the boost of cognitive psychology experiments (for review, see Frith & Rees, 2007). Nevertheless, it took another thirty to forty years until quantitatively oriented psychology and neuroscience returned to the topic of consciousness (Baars, 1988; Crick, 1994). Notable exceptions in the neglect of subjective experiences were split-brain studies in the 1960s and 1970s (for review, see Gazzaniga, 2005) and empirical research of psychophysical mechanisms and subjective contents of dreaming, which flourished between 1953 and 1993 (for review, see Foulkes, 1996).

2.5.2. Global workspace theory of consciousness

The return of empirical sciences to the problem of consciousness was largely inspired by Baars (1988, 1997, 2002), who developed a global workspace theory of consciousness that seeks to explain consciousness through its functional interactions with other cognitive processes and functions, including attention, working memory, action control, problem-solving, and language. In particular, Baars (2002) proposed the 'conscious access hypothesis', according to which the main function of consciousness is an integration, coordination and distribution of different brain processes that otherwise would remain isolated, limited, and unconscious. Consciousness can be metaphorically described as a spotlight (i.e., the focus of attention) on a theatre stage, called global workspace (i.e., working memory), which is controlled by the theatre director (i.e., executive Self) (Baars, 2002). From a vast number of the parallel and unconscious information processes, consciousness identifies the most relevant process or representation, and distributes it widely across other cognitive functions. In the global access networks, unconscious resources, such as memories or automatisms, may become the contents of the conscious spotlight in the working memory, and subsequently may modulate cognitive contexts, such as expectation and intentions of Self (Baars, 2007). The spotlight of attention may switch from modality to modality at a rate of about 100 ms, which is considered

sufficient to produce the sense of phenomenal unity across different contents of consciousness (Baars, 1998). Depending on the contextual demands, the duration of particular contents of consciousness may also be extended by re-initializing loops between unconscious resources, working memory and contexts, which, for instance, would be needed for episodic and explicit learning or voluntary control of motor activity.

Global workspace theory and models were further developed by Franklin (2003), who integrated it within a computation model of cognition, and Dehaene and colleagues who investigated neurophysiological mechanisms of the global distribution of conscious information in the brain (Dehaene et al., 2001; Dehaene & Changeux, 2011). For instance, it was demonstrated that unconsciously presented backward-masked visual words evoke localised activation of the extrastriate, fusiform and precentral areas, whereas conscious presentation of words in addition activate broad frontal and parietal networks (Dehaene et al., 2001), arguably reflecting wide access and broadcasting of conscious information (Baars, 2007). At the electrophysiological level of description, conscious access and global broadcasting seem to take place through the long-distance synchronization of the beta and gamma frequencies (Dehaene & Changeux, 2011). Yet, despite immense progress in the understanding of functions and neural mechanisms of conscious access, the global workspace theory of consciousness does not seem to offer a satisfactory account of the phenomenal consciousness. In fact, Baars (2007) argues that the typical consciousness-related questions, such as the status of qualia and the hard problem of subjectivity (see Section 2.3) do not apply to the global workspace theory of consciousness. This is simply because the global workspace theory is a theory of access consciousness, and perceptual processes outside of the cognitive machinery of global broadcasting, i.e., the contents of pure phenomenal consciousness, are regarded as pre-conscious at best (Dehaene et al., 2006). Such an approach may be successful in the identification of the cognitive interactions and functions of consciousness. However, it does not provide a satisfactory explanation of the neural mechanisms of subjective experience, as it assumes that there is no phenomenal consciousness independent of or prior to access consciousness. Phenomenal consciousness as such becomes largely eliminated from the theory.

2.5.3. Micro-consciousness vs. macro-consciousness

While Baars (1997) and Dehaene et al. (2006) argued for the global workspace theory of consciousness, Zeki (2003, 2007) rejected the classical notion of the unity of consciousness (see Section 2.2.) and proposed that consciousness exists in the form of many autonomous units called 'micro-consciousnesses'. Neurological observations and measurements of asynchrony of visual perception suggest that micro-consciousness units are distributed in both space and time (Zeki, 2007; Zeki & Bartels, 1999; Zeki & ffytche, 1998). For instance, lesions in the visual color area V4 produce achromatopsia, but do not affect motion awareness, whereas lesions in the visual motion area V5 produce akinetopsia, but do not deteriorate colour awareness. Neurological cases seem to suggest that an intact V5 is necessary and sufficient for the awareness of motion, even when primary visual cortex V1 is lesioned (Zeki & ffytche, 1998). Further, conscious perception, as opposed to unconscious information processing, may not involve any additional cortical region but instead it may differ by the increased activation level of the same cortical areas, and in principle any anatomical region of cortex could produce

its own micro-consciousness (Moutoussis & Zeki, 2002). Regarding temporal asynchrony of visual micro-consciousness units, there is evidence showing that the earliest to reach the required level of activation is micro-consciousness of location, followed by micro-consciousnesses of colour, orientation and then of motion (Bartels & Zeki, 2004; Moutoussis, 2012).

Zeki (2003) distinguished three levels of consciousness: *micro-consciousness* as subjective perception of basic phenomenal properties; *macro-consciousness*, i.e., the totality of micro-consciousness units, as subjective perception of meaningful objects; and *unified consciousness* as subjective awareness of Self. In this framework, the unity and binding are not treated as fundamental properties of phenomenal consciousness. Instead, temporary jumping activation of different areas in cortex is assumed to be sufficient for the awareness of objects and scenes at the macro-consciousness level of perceptual organization. Yet, even though Zeki's research programme shows that isolated but consciousness is not consistent with our phenomenology, as single objects as well as more complex sceneries appear to us in a unified form rather than as a stream of temporary and spatially disintegrated features. Instead, the findings of Zeki et al. seem to point to the lower representational or information processing levels of consciousness.

2.5.4. Recurrent processing as the neural basis of consciousness

Lamme (2006) proposed that the generation of phenomenal consciousness depends on the recurrent information processing between different modules across hierarchical neuronal networks. For instance, the feedforward information sweep across regions V1, V2, V4, TE of the visual ventral stream is not sufficient for the generation of conscious perception, although it may contribute to visually guided actions. Contrary to this, recurrent processing between V1, V2, V4, TE and between the ventral stream and fronto-parietal networks will lead to reportable conscious perception. Moreover, Lamme (2006) argued that recurrent processing within the ventral stream is capable of generating conscious perception even when the ventral stream does not interact with the fronto-parietal accessibility networks. If so, the contents of visual awareness cannot be equated with the attentional focus and working memory (as opposed to Baars, 2002, 2007, see Section 2.5.2), but instead visual awareness and attention are doubly dissociable (Lamme, 2003, 2004).

Hypothesized independence of phenomenal consciousness from the access and reportability became one of the main controversies in consciousness research (e.g., Block, 2011; Kouider et al., 2012). In particular, given that the presence of isolated micro-consciousness-type of processes (Zeki, 2003, 2007, see Section 2.5.3) cannot be proved directly as they are unreportable (see Section 4.1), there can be only indirect and thus questionable evidence of subjective experiences that are not available for access consciousness. Furthermore, it remains unclear why recurrent processing is necessary for the emergence of phenomenal consciousness. Possibly, the complexity of neuronal processing required for the generation of conscious experience cannot not be achieved instantly and requires temporally extended information processing, which may also point to the neural mechanisms of specious present (see Section 2.2.1).

2.5.5. Information integration theory of consciousness

Instead of selecting specific brain regions or electric frequency bands, information integration theory of consciousness, developed by Tononi (2001, 2004, 2010, 2012), equates human conscious experience with integrated information in the brain. Each conscious experience rules out an extraordinarily large number of alternative perceptual possibilities, making it highly informative (Tononi, 2007). Further, information in conscious systems is exceedingly well integrated between the elements constituting such systems, and the state of each element causally depends on the state of other elements. Consequently, phenomenal contents of a conscious state cannot be divided into phenomenal contents of its elements, as such a disintegration of information would lead to the loss of consciousness. Given that the presence of consciousness can be equated with the presence of integrated information, empirically feasible measures of such information can be developed to detect consciousness. Tononi and Sporns (2003) detailed some preliminary analytic tools to assess Φ – the capacity of a system to integrate information. Effective information between any two given subsets of a system can be assessed by injecting maximum possible entropy to one of the subsets and measuring complexity of responses of another subset. By comparing effective connectivity between all possible subsets of a system, one can identify if complexes of elements exist that maintain integrated information Φ that is lost if some of the elements become isolated from the complex (Tononi, 2008).

Tononi (2007) argued that information integration theory is consistent with phenomenological observations of informativeness and unity of consciousness, as well as psychophysical studies showing that at least 100-200 ms are required for the emergence of consciousness (Bachmann, 2000), which, given the neuronal processing speed, is roughly the time required for information to be integrated in the highly connected neuronal networks in the human brain. Generation of consciousness may depend on the high-level organization of thalamocortical networks, which seem to support both functional specialization (Zeki, 1993) and functional integration (Engel et al., 2001). By contrast, as predicted by the information integration theory, cerebellar networks consisting of an even higher number of neurons than cerebral cortex but lacking associative connections (Bower, 2002), do not seem to be directly involved in the generation of conscious experience. Likewise, consciousness is absent when cortical networks lose effective connectively and presumably the capacity to integrate information during slow wave sleep or anaesthesia (Massimini et al., 2012). However, even though information integration measures are already used in empirical studies of consciousness (e.g., Casali et al., 2013; Lee et al., 2009), it remains unclear how to assess the minimal degree of information integration required for consciousness to emerge. It is also unclear whether any integrated information, including artificially designed systems, would be conscious. Tononi (2007) seems to support the later conclusion, somewhat turning away from biological realism towards informational panpsychism (Chalmers, 1995) (see Sections 2.4.2 and 2.4.3).

After describing some of the most important concepts used in consciousness research, the fundamental properties of consciousness, the hard problem of consciousness-brain interaction, and how this problem is approached in philosophy and cognitive neuroscience, Section 3 following will focus on definitions and different types of ASCs.

3. ALTERED STATES OF CONSCIOUSNESS

The stream of consciousness does not always flow continuously, and every now and then we may experience a significant alteration of consciousness, for example, when we meditate, or – in a less fortunate case – when we get sedated or anaesthetized due to a medical condition. Some of these alterations are subtle and hardly noticeable, whereas others are very obvious, such as a change of subjective experiences in the transition from waking to sleep, yielding hypnagogic hallucinations (Ohayon et al., 1996). What happens to the phenomenal contents of consciousness in these states? Do they provide unique experiences that are not available in the normal waking state of consciousness? How can we assess the contents of consciousness of another person undergoing an alteration of consciousness, for example, a hallucinating patient? Studying such alterations is important not only for the social or fundamental understanding of human consciousness, but also for their clinical relevance, as unusual subjective experiences might be psychologically disturbing and distressing (Delespaul et al., 2002).

3.1. Definitions and classifications

During the last decades of research, a large number of diverse phenomena have been linked to or called altered states of consciousness (ASCs), including coma and vegetative state, daydreaming, drowsiness, epilepsy, extreme environmental conditions, hypnagogic states, hypnosis, meditation, near-death experiences, psychotic disorders, relaxation, respiratory manoeuvres, rhythm-induced trance, sensory deprivation, homogenization or overload, sexual activity and orgasm, sleep and dreaming, starvation and diet (Vaitl et al., 2005). This and similar lists of ASCs illustrate the complexity of the theoretical problem of trying to find unifying principles of all ASCs. Most of the definitions of ASCs focus on changes in subjective experiences and a subsequent reflection over these changes. For instance, Farthing (1992) defines an ASC as 'a temporary change in the overall pattern of subjective experience, such that the individual believes that his or her mental functioning is distinctly different from certain general norms for his or her normal waking state of consciousness' (p. 205) (see also Ludwig, 1966; Tart, 1972). The main problem with this group of definitions is that 'altered' is circularly defined as 'a change' and there are no formal criteria given to operationalize what degree of a change is needed for an ASC to be identified (Revonsuo et al., 2009). The requirement for self-awareness of a significant change of phenomenal consciousness does not hold for some of the classical ASCs, for example, a dreaming or a psychotic individual does not always realize that he or she is in an ASC.

Instead of focusing on changes in phenomenal contents of consciousness, an alternative representational definition of ASC was recently developed by Revonsuo et al. (2009) who suggested that an ASC is 'a state of the mechanisms of representation in the brain resulting in a global misrepresentational organization of the overall contents of consciousness at some moment in relation to the surrounding ("world") context (in which they occur)' (p. 195–196). This definition of an ASC focuses on changes in the preconscious or non-conscious background representational mechanisms of consciousness, which are more easily available for an

objective verification than internal subjective experiences. However, similarly to the phenomenal definition of an ASC, the representational definition does not seem to hold for all ASCs. For instance, neutral hypnosis, which has been suggested to be a powerful contrast condition for studying neural mechanisms of hypnosis (Kallio & Revonsuo, 2003), does not seem to involve any observable representational changes. Furthermore, conceptually, misrepresentation cannot be attributed to an object or a process that does not aim to represent. For example, a drawing of Barack Obama will probably misrepresent the current President of the United States of America, however it is not meaningful to attribute a property of misrepresentation of George Bush to the same drawing as it was not intended to represent him. Likewise, the contents of dreaming during sleep may not misrepresent external physical environment simply because they are not aimed at representing it.

Notably, it seems to be extremely difficult to grasp a common core of all ASCs in a single definition, as they might be both very similar and fundamentally different from each other depending on the comparison axis (Móró, 2010). Historically, naming different phenomena as ASCs was not theory-driven, but instead led by a general increased interest in the research of dreaming, meditation and LSD-induced alterations of consciousness in 1950s and 1960s (Metzner, 2010). The first operational definitions of an ASC were proposed when the concept of ASCs was already widely used (Krippner, 1972; Ludwig, 1966; Tart, 1972). A decade ago, instead of suggesting a single unifying definition of an ASC, a consortium of fifteen researchers working in the field took an alternative approach and suggested a unifying four-dimensional descriptive system under which all classical ASCs can be mapped (Vaitl et al., 2005). According to this proposal, consciousness can be altered across four dimensions: (1) activation or readiness to respond (aroused vs. relaxed); (2) awareness span or availability of experiences for attentional processing (wide vs. narrow); (3) self-awareness (present vs. absent); and (4) sensory dynamics (increased vs. decreased). For instance, typical dreaming can be characterized by relaxed activation, wide awareness span, present self-awareness and increased sensory dynamics (Vaitl et al., 2005).

Furthermore, Vaitl et al. (2005) classified ASCs according to the methods of their induction, namely: (1) spontaneously occurring; (2) physically and physiologically induced; (3) psychologically induced; and (4) disease induced. This classification proved useful as it helps to find commonalities within a particular class of ASCs, for example, not only dreaming, but most of the spontaneously occurring ASCs can be characterized by relaxed activation, narrow awareness span, present self-awareness and increased sensory dynamics (Vaitl et al., 2005). However, the consortium members excluded pharmacologically induced alterations of consciousness, which form one of the largest classification groups of ASCs, including those induced by recreational drugs, psychoactive medical drugs, anaesthetic agents, and legal stimulants, such as caffeine and alcohol (Meyer & Quenzer, 2005). Furthermore, physically, physiologically and psychologically induced ASCs may not be that different, as they all share a cognitive decision and effort to reach an ASC. In this doctoral thesis, four categories of ASCs will be distinguished, refining the earlier classification by Vaitl et al. (2005): (1) naturally occurring; (2) cognitively induced; (3) pharmacologically induced; and (4) pathological alterations of consciousness. The concepts of an ASC and an alteration of consciousness will be used interchangeably as synonyms to refer to the historically named ASCs (Farthing, 1992). No priority will be given to either phenomenal or representational definitions of an ASC, as none of them seem to include all relevant ASCs. Likewise, no single state will be regarded as the main baseline state for a contrast with an ASC. Even though a state of relaxed wakefulness may seem to be the most suitable baseline for an ASC (see Section 2.1.5), varying experimental paradigms of different alterations of consciousness may require choosing different states for a contrast with a particular ASC (see Section 4.1).

3.2. Dreaming: a naturally occurring alteration of consciousness

Natural alterations of consciousness take place in healthy individuals when unexpected or unusual intense perceptual, cognitive or emotional experiences occur spontaneously without any effort and, typically, without external physical stimulation. The most paradigmatic natural ASCs are nocturnal dreaming and a large group of other sleep-related alterations, including hypnogogic and hypnopompic hallucinations, sleep mentation, nightmares, lucid dreams, sleep paralysis and night terrors (Farthing, 1992; Hobson, 2001; Vaitl et al., 2005). Dreaming can be defined as the presence of 'complex, multimodal, dynamic and progressive conscious experiences during sleep that are organized in the form of sensory-perceptual world or a world simulation' (Revonsuo, 2009, p. 295). The definition emphasizes the spatial integration of multimodal perceptual experiences into a coherent world model and the temporal development of these experiences. However, not all perceptual experiences are dynamic during sleep, and minimal forms of dreaming exist with spatially complex but temporally static contents of consciousness (Study II, see Section 6.3), demonstrating an alteration of fundamental temporal properties of consciousness (Noreika et al., 2010c). Other perceptually relatively simple sleep-related ASCs are hypnagogic and hypnopompic hallucinations: sensory experiences occurring at the transition from wake to sleep, or from sleep to wake (Ohayon et al., 1996). While due to the suppression of reflective consciousness we are not typically aware of the fact we are dreaming, reflective consciousness is present during lucid dreaming in which the dreamer knows that he or she is dreaming (Farthing, 1992). Different types of dream lucidity will be investigated in Study VI (see Section 6.7).

Despite a close phenomenological resemblance of dreaming and waking experiences (Revonsuo, 2006), they are extremely different when it comes to their immediate sources: dream experiences are internally simulated, whereas waking experiences are largely driven by the prediction of external stimuli (Hobson & Friston, 2012). In addition, the dreaming mind has a number of cognitive deficiencies, such as the lack of metacognition (Windt & Metzinger, 2007), as well as unusually high incidence of negative emotions when compared to the waking life (Nielsen et al., 1991). Perhaps the most striking aspect of dreaming is the frequent bizarreness of its phenomenal contents (Hobson, 1988). Bizarreness can be defined as 'places, objects, persons, and other dream contents that deviate in peculiar ways from their counterparts in waking life, or show unusual combinations of phenomenal features' (Revonsuo, 2006, p. 240). Three different types of dream bizarreness are often distinguished: incongruity, uncertainty, and discontinuity (Hobson, 1988; Revonsuo & Salmivalli, 1995; see also Study III, Section 6.4). Incongruity refers to the mismatching features of dream contents, uncertainty points to the explicit vagueness of dream contents, whereas discontinuity refers to the abrupt change of identity, time, and place of other phenomenal contents of dreaming (Hobson, 1988). Different types of bizarreness are not uniformly distributed across different contents of dreaming, but instead they show a pattern of bizarreness-clustering, for example, the most incongruous are language and cognition-related dream contents, whereas the least incongruous is the dream Self (Revonsuo & Salmivalli, 1995). Revonsuo and Tarkko (2002) argued that dream bizarreness is a result of different types of binding errors occurring during sleep, indicating a breakdown of the fundamental properties of consciousness (see Section 2.2) and providing an important database for studying the unity of consciousness (Revonsuo, 1995b).

Given the broad complexity of dream experiences, a very large number of dream content analysis scales have been developed (Winget & Kramer, 1979), measuring selected aspects of phenomenal contents of consciousness, the most notable of which is the Hall and Van de Castle Scale for the Content Analysis of Dreams (Hall & Van de Castle, 1966). In addition to the content analyses of dream reports (for more details, see Section 4.2), an increasing number of studies aim to bridge from the dream phenomenology to the underlying sleep neurophysiology (Nir & Tononi, 2010). While the early attempt to establish a necessary and sufficient link between dreaming and physiology of REM sleep (Aserinsky & Kleitman, 1953) have largely failed, as up to 40% of NREM sleep awakenings yield dream reports (Nielsen, 2000), some of the recent studies have aimed to identify the neural mechanisms of particular contents of dreaming. For instance, analysis of diffusion tensor imaging has recently linked the trait of high dream bizarreness with the relatively low volume of left amygdala, whose decreased microstructural integrity was also associated with the lower emotional load of dream reports (De Gennaro et al., 2011). In another recent study, dream lucidity was found to be associated with the specific high-frequency EEG patterns in the frontal sensors (Voss et al., 2009). For a proposal of a novel brain stimulation paradigm to study neural mechanisms of lucid dreaming, see Study VI (Section 6.7).

3.3. Hypnosis: a cognitively induced alteration of consciousness

ASCs may occur in response to deliberate cognitive and/or behavioural efforts, when a repetitive action, thought or attention concentration leads to alterations of subjective experiences and the underlying representational and neural processes. The most known cognitively induced alterations of consciousness are meditation and hypnosis, the first of which can be achieved by meditators themselves, while the second typically requires an interaction between a hypnotist and a hypnosis participant. Typically, a hypnosis session consists of the induction and suggestion phases (Farthing, 1992). During induction, a hypnotist guides an individual participant or a group of participants through a set of relaxation commands. Afterwards, when a participant becomes deeply relaxed, a hypnotist may give ideomotor, response-inhibition and/or cognitive suggestions (Hilgard, 1965), the choice of which will depend on a particular clinical or experimental purpose of the hypnosis session. Receptivity of hypnotic suggestions is highly variable across individuals, and various hypnotizability scales, such as the Stanford Hypnotic Susceptibility Scale: Form C (Weitzenhoffer & Hilgard, 1962), are used to identify individuals with low, medium or high hypnotic susceptibility. Distinctions between hypnotizability categories are based on the count of individual's positive responses to the standardized suggestions, such as 'you are now beginning to have a sour taste in your mouth' (Weitzenhoffer & Hilgard, 1962).

Interestingly, hypnosis is a rather passive state of the mind and if left unguided by a hypnotist, individuals under hypnosis would not necessarily detect any obvious alterations of consciousness. Yet, when suggested, highly susceptible individuals may experience intense perceptual changes, such as positive and negative visual hallucinations (Spiegel, 2003), or even more complex hypnotic dreams (Raz et al., 2009). Regarding cognitive changes under hypnosis, a well-replicated finding is a facilitation of selective attention (Raz, 2005), which may even lead to the cessation of the Stroop effect (Raz et al., 2002). In a state of neutral hypnosis, i.e., when no specific post-induction suggestions are given, a common finding is an underestimation of time (Naish, 2007), suggesting that hypnosis distorts temporal processing of the brain/mind.

Hypnosis itself can be defined procedurally, i.e., by referring to the hypnotic techniques and situational conditions controlled by the hypnotist, or phenomenologically, i.e., by pointing out cognitive and perceptual alterations experienced under hypnosis. Following the procedural approach, the Division of Psychological Hypnosis of the American Psychological Association defines hypnosis as a procedure during which 'one person (the subject) is guided by another (the hypnotist) to respond to suggestions for changes in subjective experience, alterations in perception, sensation, emotion, thought or behavior' (Green et al., 2005, p. 262). Notably, such procedural definitions do not require an alteration of consciousness to occur, but instead a very specific suggestive social interaction itself is regarded as sufficient evidence for the presence of hypnosis. As an alternative to the procedural definitions that largely ignore the subjective experiences of a hypnotized participant, phenomenological definitions emphasize 'the hypnotic state or changes in the subject's consciousness and scope of attention caused by suggestions given by another person' (Revonsuo, 2009, p. 297). According to phenomenological definitions, application of hypnotic induction and suggestions without concurrent changes in consciousness would indicate the failure rather than the presence of hypnosis. Different emphases of procedural and phenomenological definitions are not minor conceptual disagreements between hypnosis researchers, but instead they stem from the major 'state vs. non-state' controversy regarding the ontological status of hypnosis. So called non-state socio-cognitive theories of hypnosis argue that hypnosis is a special type of social interaction between two people, one of whom obeys and follows suggestions and commands given by another person: despite a seemingly odd behaviour, hypnotized individuals do not undergo any substantial alterations of consciousness, but instead their behaviour and cognition can be explained by standard psychological principles of compliance to the social and contextual demands, expectation fulfillment, cognitive dissociation and mental imagery (Kirsch & Lynn, 1998; Spanos & Coe, 1992; Wagstaff, 1981). By contrast, proponents of the state theories of hypnosis argue that hypnotized individuals undergo alterations of the mind and brain, which cannot be reduced to typical flexibility of the mental and neural functions (Gruzelier, 1998, 2005). For instance, hypnotic suggestion to perceive colours in a gray-scale stimulus is followed by subjective colour reports as well as increase of the neural activation in the extrastriate visual cortices (Kosslyn et al., 2000; Mazzoni et al., 2009; but see Mazzoni et al., 2013).

To reconcile solid evidence in support of both state and non-state theories of hypnosis, Kallio and Revonsuo (2003, 2005) proposed that two different phenomena are conflated under the same term of hypnosis. While most of the hypnotic occurrences, especially those observed in low and medium hypnotic susceptibility groups, can be explained by reference to the typical socio-cognitive processes, a genuine hypnotic ASC may take place in highly susceptible individuals. Arguably, hypnotic state occurs only in so called hypnotic virtuosos, who show all classical features of hypnotic trance and do not require prolonged hypnotic induction to reach it. If so, case studies of such individuals may be the most sensitive research program aiming to reveal the neurocognitive mechanisms of hypnotic ASC (Kallio & Revonsuo, 2003). Following this programmatic suggestion, an extensive research of eye movements of a hypnotic virtuoso T.S.-H. recently showed that 'trance stare', a classical feature of deep hypnosis, is associated with large changes in the optokinetic and pupillary reflexes and saccade programming, which cannot be imitated voluntarily (Kallio et al., 2011), supporting the state theory of hypnosis. Temporal processing in T.S.-H. will be investigated in *Study VII* (see Section 6.8).

3.4. Sedation: a pharmacologically induced alteration of consciousness

Pharmacologically induced alterations of consciousness occur in response to the intake of psychoactive substances, the most common of which are caffeine, nicotine, and alcohol. Caffeine is known to trigger energetic arousal (Smit & Rogers, 2000), whereas nicotine tends to have a tranquilizing effect (Gilbert, 1979). High doses of alcohol may induce cognitive and perceptual alterations, such as a decline of logical reasoning (Williamson et al., 2001) or reduced sensitivity in recognizing social cues of sexual interest (Farris et al., 2010). Much more complex and intense perceptual alterations occur in response to hallucinogens, such as LSD, psilocybin, mescaline or salvia divinorum, when they are used with open expectations ('mental set') in the appropriate environment ('setting') (Faillace & Szara, 1968). For instance, unusual visual experiences, such as visual trails – a series of afterimages following a moving object (Dubois & VanRullen, 2011) – may occur after the LSD intake, whereas salvia divinorum often induces the experience of derealisation, i.e., sensation of entering another reality (González et al., 2006). LSD-induced modulation of time perception, which is another common pharmacological alteration of the contents of consciousness, will be reviewed in *Study VIII* (see Section 6.9).

One of the most extreme pharmacological disruptions of the stream of consciousness occurs under the effect of general anaesthetic when an anaesthetized patient loses awareness (Alkire et al., 2008; Mashour, 2010). *General anaesthesia* can be defined either by its clinical effects or by the clinical procedures, in particular by its induction and maintenance techniques (Urban & Bleckwenn, 2002). Interestingly, a parallel can be drawn between different types of definitions of general anaesthesia and hypnosis. Both of these ASCs can be defined either by reference to their procedures or by reference to the cognitive and behavioural state of an individual under effect. Typically, definition by the clinical effects regards immobility, analgesia, unconsciousness, and amnesia as the key features of general anaesthesia (Antognini & Carstens, 2002; Heinke & Schwarzbauer, 2002; Urban & Bleckwenn, 2002). Antognini and Carstens (2002) described a gradual model of general anaesthesia in which different clinical effects are achieved at different time points during administration of the increasing anaesthetic dosage. A relatively low anaesthetic dosage supresses recall of intraoperative experiences, a higher concentration induces unconsciousness, and further increase in the dosage leads to the complete immobility.

Rather shockingly, administration of anaesthetic agents does not always result in the complete loss of subjective experiences, leading to various anaesthetic-induced alterations of consciousness. Most notably, intraoperative anaesthesia awareness occurs when a patient under general anaesthesia maintains or regains awareness of being operated on (Bergman et al., 2002; Mashour et al., 2012; Sebel et al., 2004). Somewhat relieving is a finding that many of the anaesthesia awareness occurrences do not involve full consciousness. Instead, only some aspects of perception or cognition are typically regained, such as hearing talking throughout the procedure or feeling intubation (Bergman et al., 2002). Likewise, even though most of awareness patients experience anxiety, sensation of pain does not always take place (Lennmarken et al., 2002). While anaesthesia awareness is a relatively rare phenomenon estimated to occur in about 0.1% of general anaesthesia cases (Mashour et al., 2012; Sebel et al., 2004), a much more common anaesthetic-induced alteration of consciousness is anaesthesia dreaming, with incidence ranging from 4% (Leslie et al. 2005) to 53% (Errando et al., 2008) of patients undergoing general anaesthesia. Anaesthesia dreaming is often defined as 'any experience that a patient is able to recall postoperatively that he or she thinks is dreaming and which he or she thinks occurred between the induction of anesthesia and the first moment of consciousness after anesthesia' (Leslie, 2010, p. 75). While being sufficient in the clinical context, the reliance of this definition on the explicit recall of a dream excludes a scientifically interesting possibility that unrecalled subjective experiences may exist during general anaesthesia. Arguably, most of the anaesthesia-related alterations of consciousness occur at the relatively low dosage of anaesthetic agents in a state of *sedation*, in which the use of medical agents leads to the reduced level of consciousness (Morita et al., 2002), but the individual still maintains a capacity of residual consciousness. Complexity of subjective experiences in a state of anaesthetic-induced sedation will be analysed in Study I (see Section 6.2).

3.5. Psychosis: a pathological alteration of consciousness

While most of the naturally occurring, as well as cognitively- or pharmacologically- induced, ASCs can be experienced by any interested healthy individual, psychiatric or neurological patients may undergo disorder-related alterations of subjective experiences, for example, waking visual hallucinations in bipolar disorder (Baethge et al., 2005), or perceptual distortions during partial epileptic seizures (Johanson et al., 2008). Most of the pathological states of consciousness persist for relatively long periods of time, yet some of the alterations are relatively short-lasting, for example, visual migraine aura typically continues from 5 to 60 minutes (Viana et al., 2013). Notably, even though the extreme cases of pathological alterations of consciousness seem very odd, for example, bizarre delusions (Cermolacce et al., 2010), there is sometimes no clear-cut line separating pathological and normal experiences. For instance, there seems to be a continuum of paranoid thinking from occasional ungrounded suspicions in healthy individuals to fixed delusions in psychotic patients (Freeman et al., 2010, 2011).

Particularly intense and complex alterations of consciousness occur in psychosis, which can be broadly defined as the pathological presence of hallucinations and delusions (Bentall, 2007). Given that psychosis itself is not a categorized disorder but rather a set of symptoms observed in different psychiatric disorders, the more precise definitions of psychosis are disorder-specific. Thus, for example, in schizophrenia 'the term *psychotic* refers to delusions, any prominent hallucinations, disorganized speech, or disorganized or catatonic behavior' (American Psychiatric Association, 2000, p. 297), whereas in substance-induced psychotic disorder 'psychotic refers to delusions or only those hallucinations that are not accompanied by insight' (American Psychiatric Association, 2000, p. 298). In schizophrenia, psychotic episodes are associated with intense changes in perception, cognition and/or emotions. Perhaps the most distinct perceptual disturbances in schizophrenia are auditory and verbal hallucinations ('voices') (Johns & McGuire, 1999), which are likely caused by abnormal activation levels of the auditory cortex and Broca's area (Kühn & Gallinat, 2012). However, visual (David et al., 2011) and olfactory (Arguedas et al., 2012) hallucinations may also be experienced by psychotic schizophrenic patients. Regarding reflective consciousness, psychosis in schizophrenia is often marked by reasoning abnormalities and complex delusions, i.e., false beliefs (Langdon et al., 2010). Emotional changes in schizophrenia can range from emotional withdrawal (Kay et al., 1987) to high levels of agitation (Marco & Vaughan, 2005). While schizophrenia can affect very broad aspects of phenomenal and reflective consciousness, some other psychiatric disorders primarily affect only a particular dimension of consciousness, for example, emotional alterations dominate in affective disorders (American Psychiatric Association, 2000).

Importantly, psychiatric or neurological disorders may be associated with changes not only in the baseline waking state, but also in the 'normal' ASCs, for example, psychotic patients with daytime delusions have an increased frequency of grandiosity topics in their dreams (D'Agostino et al., 2013). In this case, a paradoxical 'alteration of alteration' is encountered: patients' dreams may be different from dreams of healthy individuals, whose dreams in turn are different from their waking mentation. Possible alterations in the bizarreness of dreaming in psychotic schizophrenia patients will be investigated in *Study III* (see Section 6.4).

3.6. Unconscious states of mind

As discussed earlier, an *unconscious state* of mind contains no qualia, but maintains a potential to become conscious (see Section 2.1.6), which may provide important contrast conditions for studying the neural mechanisms of consciousness. A reversible unconsciousness may be a result of various neurological disorders, for example, a complete loss of consciousness is very typical but not always present after complex partial seizures in epilepsy patients (Johanson et al., 2003). Yet, perhaps the most common unconscious states are *dreamless sleep*, which will be investigated in *Study II* (see Section 6.3), and general anaesthesia or sedation, which will be explored in *Study II* (see Section 6.2). The loss of conscious experiences during dreamless sleep (Gillespie, 2002; Tononi & Massimini, 2008) is the only naturally occurring reversible fading of the stream of consciousness. The depth of sleep is typically assessed by visual scoring of polysomnographic (PSG) measurements, in particular EEG, electrooculography (EOG) and electromyography (EMG) activity patterns (Iber et al., 2007). Consciousness is often regarded to be lost when sleep EEG is dominated by the low frequency EEG delta activity (0.5–3 Hz). During deep NREM sleep, only a limited degree of integrated information is maintained in the cortex, which may prevent conscious experiences to emerge (Tononi, 2009). Up to 20% of

sleep laboratory participants never report subjective experiences after awakenings from slow wave NREM sleep, and a further 30% of participants are very unlikely to report dreaming or mentation after awakenings from these stages of sleep (Cavallero et al., 1992).

Similarly to NREM sleep, deep anaesthesia is also associated with decreased high frequency activity (Tonner & Bein, 2006); however, this pattern is not universal and, for example, S-ketamine anaesthesia has increased spectral power in high rather than low frequencies (Maksimow et al., 2006). To monitor the depth of anaesthesia and sedation, various complex EEG-based indexes have been developed, the most commonly used of which is the bispectral (BIS) index. The BIS index is a non-linear combination of four different measures: the relative spectral power of the beta and gamma frequency range; the bispectral component of the broad frequency range; the 'QUAZI' suppression component; and the 'burst suppression' measure (Schneider, 2010). The BIS values from 100 to 80 indicate an awake state, 80 to 60 - sedation, 60 to 40 - anaesthesia, 40 to 20 - deep anaesthesia, and 20 to 0 -EEG burst suppression (Schneider, 2010). Generally, the lower the BIS index values are, the less likely it is that the patient is conscious (Kearse et al., 1998); however, intraoperative awareness may occur even when this BIS index is below the recommended value of 60 (Avidan et al., 2008). A further theoretical complication of the status of unconscious states arises from a possibility that the stream of consciousness may never be interrupted during deep sleep or general anaesthesia, for example, we may always dream in unresponsive states such as NREM sleep, but may not be able to recall it (Schredl et al., 2013). If so, the absence of reported subjective experiences may indicate a failure of memory rather than the absence of conscious experiences. Currently, there are no available empirical methods to confirm or exclude such possibilities.

3.7. Interactions between different alterations of consciousness

Even though each of the reviewed ASCs could be defined as a distinctive combination of the induction techniques and consequent alterations in perception, cognition, emotions and the underlying representational and neurobiological processes, some of the alterations seem to be phenomenologically analogous and perhaps may even share some of the neurocognitive mechanisms. For instance, Schwartz and Maquet (2002) pointed out that bizarre dreaming reminds many of the neuropsychological syndromes, such as misidentification of faces (Frégoli syndrome), places (reduplicative paramnesia), or loss of colour saturation (achromatopsia), suggesting that natural alterations of consciousness may temporarily replicate more persistent pathological ASCs. Lynn et al. (2012) argued that the depth of two major cognitively induced ASCs – hypnosis and meditation – could be modified by similar imaginative suggestions, and that both of these states could be successfully applied to train attention processing. Individual hypnotizability was also linked to habitual daytime sleepiness (Móró et al., 2011), showing that susceptibility to a cognitively induced ASC can be related to a trait of transition to a natural alteration of consciousness within the wake/sleep cycle.

Despite a number of somewhat sporadic comparisons (e.g., Lynn et al., 2012; Móró et al., 2011; Schwartz & Maquet, 2002) and several more systematic attempts to relate various alterations of consciousness (Hobson et al., 2000; Vaitl et al., 2005), contrastive analyses of

different ASCs are still in a very premature state. Likewise, methods developed to study one of the alterations, for example, content analysis of subjective dream reports (Winget & Kramer, 1979), are seldom applied to study other states of consciousness. As encouraging exceptions, the content analysis methods are sometimes applied to study subjective reports of partial epileptic seizures (Johanson et al., 2008) and auditory hallucinations (Kent & Wahass, 1996) (for the content analysis of anaesthetic-induced experiences, see *Study I*, Section 6.2). Arguably, the major reason for this somewhat surprising neglect lies in methodological difficulties of finding suitable tools and experimental manipulations for systematic comparisons, either between a selected alteration of consciousness and a baseline state of consciousness, or between different ASCs. Most of the theoretical advancements in this thesis will be directed to the search for systematic associations between different states of consciousness (*Studies IV*, *V*, *VIII*, see Sections 6.5, 6.6, 6.9). Methodological limitations and possible solutions for studying ASCs are discussed in the following section.

4. EMPIRICAL METHODS OF STUDYING ALTERED STATES OF CONSCIOUSNESS

Experimental research of consciousness is limited by the *asymmetry of access* to subjective experiences (Velmans, 2007): while the empirical sciences of consciousness seek explanation of the mechanisms of consciousness from the objective *third-person perspective*, the experiencing subjects as participants of experimental studies can report the contents of their consciousness only from the subjective *first-person perspective* (Nagel, 1974). Given that the first-person data are not directly accessible by the third-person scientific methods, the asymmetry of access leads to a number of methodological obstacles that are specific to consciousness studies.

4.1. Facing the obstacles: introspection, reportability, baseline, and sampling

As long as the necessary and sufficient neural correlate of the phenomenal consciousness is unknown, the presence and the contents of subjective experiences in another person can only be reliably confirmed through a subjective report of a conscious agent, for example, by a button press (Salminen-Vaparanta et al., 2012), an eye blink (Marcel, 1993), a verbal answer (Schredl et al., 2009), or an act of mental imagery (Cruse et al., 2011). The most novel fMRI analysis methods can decode hemodynamic representations of visual experiences not only during waking but also during sleep onset, yet such experimental designs still rely on participants' confirmatory verbal reports (Horikawa et al., 2013). Naturally, progress of consciousness research largely depends on the accuracy of participants' *introspection*, i.e., 'thinking about one's primary conscious experiences [...] for the purpose of describing and interpreting them' (Farthing, 1992, p. 58). Given that introspection depends on memory (James, 1890), it can be called a *retrospection*, i.e., thinking about remembered past experiences (Farthing, 1992), a paradigmatic example of which is a post-awakening reporting of dream experiences.

Introspecting participants are expected to be cooperative and trustworthy, i.e., they should not censor or confabulate their reports (Farthing, 1992), which is very complicated to control. These methodological constraints, i.e., reportability of the contents of consciousness and cooperativeness of the conscious agent, are particularly challenging when studying ASCs. For instance, reportability might be severely impaired due to limited meta-consciousness, i.e., explicit re-representation of the contents of consciousness (Schooler, 2002). In particular, hallucinating individuals might be so absorbed by unusual experiences that little resources are left to reflect on and verbalize their perceptual or cognitive experiences. Likewise, individuals experiencing alterations of consciousness may not be as cooperative and willing to report as they would be in a normal baseline state of consciousness, a paradigmatic example of which is the lack of cooperativeness of a depressed patient (Goekoop et al., 2009).

Furthermore, reportability might be completely lacking in unresponsive alterations of consciousness, such as sleep. In cases when an agent is unable or unwilling to report their experiences, a conscious person might be misjudged as unconscious (Mashour & LaRock, 2008),

as unreportable or unreported mental processes might still be subjective, although unavailable for further cognitive and behavioural processing. In addition to sleep, other paradigmatic alterations with a possible dissociation between (un)responsiveness and (un)consciousness are general anaesthesia and vegetative or minimally conscious state (Boly et al., 2013). Unfortunately, given the methodological limitations of reportability and cooperativeness, there are very few studies that have systematically assessed the presence of subjective experiences in a well-controlled unresponsive state of the brain (but see *Study I*, Section 6.2).

Uncertain status of a baseline state of consciousness is yet another major obstacle that complicates research of the perceptual, cognitive and neural mechanisms of ASCs. Metaphorically, an alteration of consciousness could be defined as a significant diversion from the mainstream of consciousness, i.e., from a normal state of a healthy waking mind (see Sections 2.1.5 and 3.1). Yet, the choice of the normal baseline state against which an altered state should be identified and studied is uncertain, as there are no widely established settings (such as time of the day, body position, experimental task, etc.) under which the baseline state should be in order to be identified as an alteration of consciousness, for example, it is debatable whether diurnal fluctuations of perception (Alfuth & Rosenbaum, 2011), cognition (Marek et al., 2010), or emotions (Clark et al., 1989) would count as naturally occurring alterations of consciousness.

A further methodological complication arises when a contrast between a baseline waking state and an ASC is not feasible due to unresponsiveness of research participant. For instance, while the same time perception task could be carried out in hypnosis and in baseline waking state (Naish, 2001), it could not be performed by dreaming participants, unless they would become lucid (Erlacher & Schredl, 2004). In such cases, a different type of alteration could be used as a baseline contrast condition, for example, EEG studies aiming to find electrophysiological correlates of dreaming have tried to compare different sleep stages (Corsi-Cabrera et al., 2003), dreaming vs. dreamless sleep (Esposito et al., 2004), or to contrast different types of dreams (Nielsen & Chénier, 1999). Comparison of dreaming and dreamless sleep enables a sharp contrast between the presence and the absence of consciousness itself or its recall, as dreamless sleep might contain forgotten dreaming, as argued by Schredl et al. (2013). A contrast of dreams with varying frequency of selected experiences highlights a possibility of studying gradual transitions between different alterations of consciousness.

The major practical limitation of studying ASCs is inefficient data collection. Some of the ASC occur so seldom, for example, lucid dreaming, that it is almost impossible to study them experimentally, in which case studying a single well-trained participant might be an optimal choice (for a hypnotic virtuoso's case study, see *Study VII*, Section 6.8). While other natural ASCs, such as REM sleep dreaming, may occur much more frequently, perhaps each night, only four or five measurements are usually possible during one night. *Study II* aimed to improve the efficiency of laboratory research of dreaming by introducing a serial awakenings paradigm (see Section 6.3). Finally, in addition to practical limitations, there are certain ethical restrictions to study some of the pharmacologically induced or pathological ASCs, as, for instance, research participants may not always have a cognitive and/or behavioural capacity to consent their participation or to withdraw from the study if they would wish so.

4.2. Content analysis of subjective reports

In laboratory research of dreaming, the reportability constraint is addressed by collecting subjective reports after participants are awakened (e.g., Schredl et al., 2009). Typically, verbal dream reports are audio-recorded and later transcribed for the content analysis of written texts, during which a rater identifies and classifies specific contents of dreaming, such as a number of live characters, or rates more complex features of dreaming, such bizarreness or lucidity. This way, subjective experiences within an ASC are evaluated outside of the alteration itself. Obviously, this approach has a number of introspection-related limitations, including forgetting, reconstruction errors, verbal description difficulties, censorship of intimate experiences, and lack of independent verification (Farthing, 1992), as well as errors made by the experimenter performing content analysis. While all of these limitations are methodologically challenging to deal with, the memory confound is particularly problematic in dream research, as sleepy participants may forget or censor certain dream experiences, or they may not even recall the presence of dreaming itself. Thus, a participant could report a state of dreamless sleep when in fact there was an intense, but forgotten, dream. Feasibility of such cases is supported by frequent occurrence of white dreaming, a state when an awakened person has a strong feeling that he or she was dreaming but does not recall any specific experience from the dream, which may happen in about 40% of awakenings from the early night stages 2 and 3 of NREM sleep (Study II, see Section 6.3). Hurlburt and Heavy (2004) described a number of methods that may improve the accuracy of reports of beep-interrupted waking mentation, for example, the beep should come from the participant's own natural environment. Similarly, awakening process should be well-controlled in dream research, as gradual awakenings from sleep result in higher rate of thought-like reports than do abrupt awakenings (Shapiro et al., 1963). Also, participants should be instructed to restrain their body movements after awakening, as motor activity can interfere with dream recall (Calkins, 1893).

In order to minimize content analysis errors, quantitative analyses are often carried out by several blind judges aiming at a strong inter-judge agreement (>75%), this way reducing possible bias in identification and classification of subjective experiences (e.g., Valli et al., 2005). Reliability assessment is particularly important when new content analysis scales or systems are devised (such as in Studies I, II, see Sections 6.2, 6.3). In addition to the reliability of dream content analysis, the validity of measurement should be assessed when possible (Kramer, 2007). Ideally, a dream scale should be content-valid, i.e., appropriate to the concept being measured. Further, a scale should show construct validity, i.e., it should be measuring the right construct and its different items should be related around the same construct or constructs. Finally, a valid dream scale should show predictive validity, i.e., it should be sensitive to detect the predicted research outcomes or hypothesized group differences. Unfortunately, formal psychometric analyses of dream content scales are seldom carried out (for a recent attempt, see Stumbrys et al., 2013). Given that neither experimenters nor awakened research participants have direct access to dream experiences, assessment of the content and construct validity of dream scales must rely on the poor memory and limited verbal reports of unique dreams. Unfortunately, well-controlled, repeated measurements are impossible in dream research, for instance, while doubtful measurements of hypnotic susceptibility could be repeated with the same individual on another day, sleep researchers do not have a privilege to ask their participants to undergo the same dream one more time. Thus,

there is no direct way to check if the scale showing only negative emotions may have missed the positive ones that were experienced but forgotten.

Stability of dream content analysis is also inflated by a large intra-individual variability of dream characteristics, such as bizarreness and emotional tone (Schredl et al., 2001). It is recommended that at least 20 dream reports per individual should be collected in a short period of time in order to obtain reliable measures of dream content (Schredl et al., 2001), yet it is not always feasible in time-consuming laboratory studies of dreaming (efficiency of laboratory collection of dream reports will be addressed in *Study II*, see Section 6.3). If possible, highly motivated and introspective participants should be recruited for the laboratory studies of dreaming, as higher recall is associated with an individual's positive attitude towards dreams (Beaulieu-Prévost & Zadra, 2007). In addition, dream recall can be increased through training to pay attention to dreams, especially for the low and medium dream recallers (SchredI, 2002; SchredI & Doll, 1997).

Despite the suggested methodological improvements, it is necessary to acknowledge that dream research remains dependent on the limitations of subjective introspective reports. Two alternatives exist here: a simple sceptical solution would be to regard dream research as a non-scientific enterprise (e.g., Malcolm, 1959), yet this position ignores certain empirical facts about the world, i.e., the subjective reality of our dream experiences. Further, given that behaviouristic scepticism regards dreams as unsuitable objects for the science of empirical facts, a natural deduction from this position would be that dream experiences are not, or do not, depend on empirical, i.e., neurobiological, processes, which is not compatible with biological approaches that treat consciousness as an empirically and causally real phenomenon. Instead, a psychobiologically more plausible approach would be to continue developing introspection methods, aiming at as controlled and robust experimental designs as possible. Conscious states derived from introspective reports are often good predictors of human behaviours (Wilson, 2003), and as such they do generate descriptive categories that can be used in scientific hypotheses and theories. In fact, research using content analysis of dream reports has already yielded a number of well-replicated interesting findings (Barrett & McNamara, 2007), and other sciences that have to deal with subjective experiences may benefit from the adaptation of dream research methods. For instance, studies of anaesthesia dreaming typically evaluate only the presence or absence of subjective experiences without analysing further the contents of phenomenal consciousness (e.g., Leslie et al., 2009). Arguably, research of consciousness alteration during sedation and anaesthesia would benefit a lot from the application of dream analysis methods (as was done in *Study I*, see Section 6.2).

4.3. Behavioural methods to study alterations of consciousness

Instead of collecting and analysing verbal introspective reports, consciousness can be investigated by behavioural methods, for example, by analysing participants' responses in stimuli detection or discrimination experiments (Braun & Julesz, 1998; Fisk & Haase, 2005). In a typical behavioural experiment of the contents of consciousness, physical stimuli are presented at the threshold level of awareness, and participants are asked to report if they perceived the stimulus or not (detection tasks), or alternatively, which of the several possible

stimuli did they experience (discrimination tasks). By systematically manipulating physical properties of stimuli, for example, the duration of targets (Pessoa et al., 2005), and/or cognitive functions, such as expectancy and attention (Koivisto & Revonsuo, 2007), and analysing how these manipulations are reflected in behavioural measures of awareness, a large range of questions can be investigated, such as: what is the temporal order of awareness of different visual features (Zeki & Bartels, 1999); is awareness necessary for top-down attention and working memory (Hsieh & Colas, 2012); how does preceding unconscious or non-conscious processing modulate subsequent conscious awareness (Peremen et al., 2013); and is conscious perception gradual or dichotomous (Overgaard et al., 2006; Sergent & Dehaene, 2004). The presence of specific contents of consciousness is typically assessed by the dichotomous 'yes' or 'no' responses, for example, by measuring the hit rate in a certain condition or calculating more complex signal detection measures, such as discriminability and response criterion (Ricci & Chatterjee, 2004). In addition to dichotomous responses, other behavioural measures can also be used to assess the consciousness, including reaction times (Kentridge et al., 2008) or, for instance, a temporal rate of perceptual reversals in response to the presentation of ambiguous figures (Intaite et al., 2013). Further, several awarenessspecific behavioural measures may be used to assess phenomenal consciousness more accurately, such as confidence ratings of target visibility, which enables more differentiated analysis of subjective experiences, for example, visual awareness of stimulus presence can be compared with more complex awareness of stimulus type (Salminen-Vaparanta et al., 2012).

Similarly to the experiments of the contents of consciousness in a typical waking state, behavioural measures can be applied to study ASCs. For instance, individuals can perform the same cognitive task in a baseline state and an ASC or in several different ASCs, which would allow a contrast of cognitive processes in different states of consciousness with phenomenal contents being relatively well-controlled and largely driven by experimental stimuli (for such a contrast between time perception in hypnosis and baseline waking state, see Study VII, see Section 6.8). However, even though paradigms using behavioural methods have much higher control of the phenomenal contents of consciousness than experiments relying on the content analysis methods (see Section 4.2), they still largely rely on the accuracy of participants' introspection. In dream research, participants introspect their subjective experiences and then give a verbal report, whereas in behavioural studies they have to report their introspection by a motor response, unless it is a speeded reaction times task. Notably, analysis of verbal reports enables multidimensional measurement of relatively complex subjective experiences, whereas behavioural methods typically focus on a very narrow aspect of phenomenal or access consciousness, ignoring the broader spatial and temporal unity of consciousness (see Section 2.2). However, behavioural methods allow rapid sampling of the same type of subjective experiences, and provide tools for studying interaction between consciousness and other mental functions. Thus, more powerful cognitive and statistical analyses are possible in behavioural studies than those available for the content analysis studies.

4.4. Neuroimaging of altered states of consciousness

The two major approaches to study the neural mechanisms of consciousness include experimental designs focusing on the selected contents of consciousness, and paradigms

designed to study the overall states of consciousness (Dehaene & Changeux, 2011; Rees & Frith, 2007). The first class of studies may present stimuli close to threshold (Kim & Blake, 2005; Macknik, 2006) or, for instance, show ambiguous stimuli with alternating perceptual interpretations (Kleinschmidt et al., 2012; Kornmeier & Bach, 2012), and then contrast brain responses between trials that differ in awareness of these stimuli. The second class of experiments typically contrasts evoked responses or spontaneous brain activity between the baseline state of consciousness and different ASCs, such as sleep (Massimini et al., 2010), sedation (Långsjö et al., 2012), and minimally conscious states (Boly et al., 2004). The most widely used neuroimaging methods to study ASCs are EEG, which records electrical brain activity along the scalp (e.g., Voss et al., 2009), fMRI, which measures cerebral blood flow coupled with neuronal activity (e.g., Owen et al., 2006), positron emission tomography (PET), which detects blood flow or glucose metabolism (e.g., Boly et al., 2008), and combined transcranial magnetic stimulation (TMS)-EEG, which can assess cortical information flow in response to transcranial perturbation of spontaneous electric activity (e.g., Massimini et al., 2012). It has been argued, however, that the aforementioned neuroimaging methods are very unlikely to solve the hard problem of consciousness, as they are spatially and/or temporally too coarse to detect the electrophysiologically complex phenomenal level of organization in the brain (Revonsuo, 2001). For instance, the speed of neurobiological processes constituting consciousness is much faster than the speed of measurement of fMRI or PET scanning, whereas the sources of EEG signals registered on the scalp are largely unknown.

Nevertheless, the neuroimaging studies have helped to detect a number of neural correlates (even though not constituents) of altered consciousness, such as a functional recoupling of subcortical and fronto-parietal regions upon regaining consciousness after anaesthetic-induced unresponsiveness (Långsjö et al., 2012). A recently developed perturbational complexity index (PCI), a TMS-EEG measure of cortical information integration and differentiation (see Section 2.5.5), can be used to distinguish different levels of consciousness in sedation, sleep, and traumatic brain injury (Casali et al., 2013). Yet, many such types of studies conflate consciousness with responsiveness and do not systematically assess if unresponsive participants still have internally generated subjective experiences (Mashour & LaRock, 2008, see also Study I, Section 6.2). Arguably, instead of focusing on the raw comparison between the baseline and ASCs, a direct contrast between brain activity related to different types of spontaneous subjective experiences when the brain state is controlled, such as different types of dreams or dreaming vs. dreamless sleep when the sleep stage is kept the same (see Study II, Section 6.3), is more eligible to reveal the neural mechanisms of altered phenomenal consciousness. The earlier attempts to start the neurophenomenology programme of research, which would systematically match the firstperson data with the large-scale neurodynamics (Varela, 1996), did not flourish, largely because of the methodological limitations and biases related to the first-person reports (Lutz & Thompson, 2003). This program was also heavily anchored to the philosophical tradition of phenomenology (Petitot et al., 1999), the concepts and metaphysical assumptions of which were not intuitive for empirical scientists to accept. It is expected that Studies I-VI (see Sections 6.2–6.7) will contribute to the development of more controlled and efficient analysis of the first-person reports, eventually leading to the more rigorous neuroimaging studies of ASCs.

5. AIMS OF THE PRESENT THESIS

The main purpose of this thesis was to investigate subjective experiences and cognitive processes in four different types of alterations of consciousness: naturally occurring dreaming, cognitively induced hypnosis, pharmacologically induced sedation, and pathological changes of consciousness in schizophrenia. More specific goals of the thesis were:

- 1. To empirically verify the dissociation between consciousness as responsiveness and consciousness as the presence of subjective experiences (*Study I*, see Section 6.2).
- 2. To develop and test new content analysis methods for mapping phenomenology in unresponsive states, such as pharmacologically induced sedation (*Study I*, see Section 6.2) and natural sleep (*Study II*, see Section 6.3).
- 3. To develop an efficient laboratory paradigm for conducting awakenings and collecting data for a contrast between dreaming and dreamless NREM sleep (*Study II*, see Section 6.3), and to propose novel approaches for a contrast between lucid and non-lucid dreams (*Study VI*, see Section 6.7).
- 4. To contrast dreaming and waking mentation reports collected from psychotic schizophrenia patients and healthy controls (*Study III*, see Section 6.4), and to develop a conceptual framework of how these natural and pathological alterations might be related (*Study IV*, see Section 6.5).
- 5. To analyse whether dreaming can be treated as a research model of psychotic and/or intact waking state of consciousness (*Study V*, see Section 6.6).
- 6. To investigate time perception distortions under cognitively induced hypnosis (*Study VII*, see Section 6.8), and to develop a theoretical framework of how such distortions might be associated with other ASCs that undergo similar changes in time perception (*Study VIII*, see Section 6.9).

6. ORIGINAL STUDIES

6.1. General overview

Deriving from the formulated goals, both empirical and theoretical research were carried out in the present thesis (for an overview of all studies, see Table 1). Theoretical studies (*IV–VI*, *VIII*, see Sections 6.5–6.7, 6.9) reviewed scientific literature and suggested new ways of how research of different ASCs could be advanced and integrated. Empirical studies (*I–III*, *VII*, see Sections 6.2–6.4, 6.8) focused on the specific contrasts between two or more states of consciousness (for an overview of contrasted states of mind, see Table 2). Experimental paradigms and methods used in these studies are described in the following sections.

Studies	Participants	Rationale	Methods	Publication type
I	40	To analyse subjective experiences occurring under unresponsive sedation and to demonstrate discrepancy between different definitions of conscious states.	 Experimentally induced sedation. EEG-derived analysis of sedation depth. Content analysis of introspective reports. 	Peer-reviewed empirical article.
II	5	To develop a new laboratory paradigm to efficiently collect data for a contrast between minimal NREM dreaming and dreamless sleep.	 PSG. EEG analysis. Content analysis of dream reports. 	Peer-reviewed empirical article.
III	10	To analyse bizarreness in dreams and to compare overall similarity of dreaming and spontaneous waking mentation in healthy individuals and psychotic patients.	 Assessment of psychotic symptoms, memory functioning and sleep structure. Content analysis of dream reports. Blind categorization of dream and waking mentation reports. 	Peer-reviewed empirical article.
IV	-	To draw a formal scheme of how dreaming and schizophrenia might be related.	A theoretical paper.	An extended invited commentary on Zanasi, M., Calisti, F., Di Lorenzo, G., Valerio, G., & Siracusano, A. (2011a). Oneiric activity in schizophrenia: Textual analysis of dream reports. <i>Consciousness and</i> <i>Cognition</i> , 20, 337–348.
V	-	To evaluate theoretical and empirical feasibility of using dreaming as a research model of waking consciousness and/or psychosis.	A theoretical paper.	Peer-reviewed review article.

Table 1: General overview of empirical and theoretical studies.

Studies	Participants	Rationale	Methods	Publication type
VI	-	To develop experimental hypotheses for studying dream lucidity and its neural correlates by focusing on non-lucid or pre-lucid dreams.	A theoretical paper.	An extended invited commentary on Hobson, A. (2009). The neurobiology of consciousness: Lucid dreaming wakes up. International Journal of Dream Research, 2, 41–44.
VII	1	To compare perception of duration under hypnosis and in the baseline waking state.	 Hypnotic induction. Psychophysical assessment of duration perception. Cognitive modelling of behavioural data. 	Peer-reviewed empirical article.
VIII	-	To propose a formal model of how pathological, cognitively and pharmacologically induced, and natural distortions of time perception might interact.	A theoretical paper.	Invited book chapter.

Table 2: Experimental contrasts between alterations of consciousness, baseline waking states and unconscious states of mind.

Studies	Naturally occurring ASCs	Cognitively induced ASCs	Pharmacologically induced ASCs	Pathological ASCs	Unconscious states	Baseline waking consciousness
I			Dexmedetomidine sedation dreaming Propofol sedation dreaming. Sevoflurane sedation dreaming Xenon sedation dreaming.		Dreamless dexmedetomidine sedation.	
II	NREM sleep dreaming.				NREM dreamless sleep.	
	Natural sleep dreaming.			Psychotic sleep dreaming. Psychotic waking mentation.		Natural waking mentation.
VII		Cognition in neutral hypnosis.				Baseline waking cognition.

6.2. Study I: Consciousness lost and found: Subjective experiences in an unresponsive state (Noreika et al., 2011)

6.2.1. Methods

Anaesthesiological definition of consciousness equates it with responsiveness, yet it is conceivable to assume that an unresponsive individual may be subjectively conscious (Mashour & LaRock, 2008). To test a hypothesized dissociation between the presence of subjective experiences and responsiveness in pharmacologically induced alterations of

consciousness, 40 right-handed male volunteers (age: M=23.3, SD=2.7; American Society of Anesthesiologists physical status I) were sedated in *Study I*. Four participant groups were formed with the following sedative/anaesthetic drugs used to induce sedation: dexmedetomidine group – 10 participants (2 sessions each); propofol group – 10 participants (2 sessions each); sevoflurane group – 10 participants (1 session each); and xenon group – 10 participants (1 session each); and xenon group – 10 participants (1 session each); and xenon group – 10 participants (1 session each). During all sessions, the depth of sedation was monitored with an EEG-derived BIS index (BIS XP, algorithm version 4.0, smoothing rate 15 s, Aspect Medical Systems). BIS index is calculated from a signal recorded from an electrode strip attached to the forehead, and it ranges from 0, which reflects isoelectric EEG silence, to 100, which is observed when an adult participant is fully awake and alert (see Section 3.6).

Concentration of a dexmedetomidine, propofol or sevoflurane was stepwise increased until the loss of responsiveness (LOR). LOR was defined by an absence of behavioural response to a verbal command 'Open your eyes!', which was requested at 5 min intervals. Due to pharmacokinetic properties of xenon, its sedation was induced more abruptly. After discontinuation of drug delivery, the regaining of responsiveness (ROR), i.e., the first meaningful response to the verbal command 'Open your eyes!', was assessed for each drug group at 1 min intervals. ROR confirmation was followed by resting EEG recording that lasted for 5 min. Afterwards, a structured interview was taken from each participant regarding subjective experiences during sedation. The interview was repeated after 10–30 min, when participants had recovered from sedation.

For the analysis of frequency and quality of subjective experiences in post-sedation reports, the *Subjective Experiences During Anaesthesia Coding System* (SEDA-Coding) was developed. SEDA-Coding distinguishes 3 categories of subjective experiences: (1) micro-level experiences (visual, auditory, sensorimotor, olfactory, gustatory, positive affective, negative affective, thought-like); (2) macro-level experiences (dream-like, laboratory-related, out-of-body experiences, changed experience of time); and (3) white reports when participant remembers having had some experiences but cannot recall any specific details.

6.2.2. Results

Content analysis of interviews showed that subjective experiences were reported in 58.6% of sedation sessions. When reports of the first session were compared across sedative/anaesthetic agents, the difference in the frequency of reports with subjective contents approached significance (Fisher's exact test: p=.057, N=40). Individual drug comparisons showed that dexmedetomidine sedation was accompanied by higher incidence of reports with subjective experiences than propofol sedation (Fisher's exact test: p<.05, N=20). Regarding content categories of SEDA-Coding, the four sedative/anaesthetic agents differed in the frequency of reports with laboratory-related experiences (Fisher's exact test: p<.05, N=40). Individual drug comparisons showed that sevoflurane sedation induced more experiences related to laboratory, medical team and hospital than dexmedetomidine sedation (Fisher's exact test: p<.01, N=20).

Analysis of averaged BIS values showed significant differences between propofol, dexmedetomidine and sevoflurane sedation during the loss of responsiveness period

(repeated measures ANOVA: F(2,27)=5.79, p<.01). In particular, BIS values were lower, suggesting a deeper sedation, during propofol (M=60.54) than during sevoflurane (M=70.00) sessions (independent t test: t(18)=2.99, p<.01, d=1.41). Similarly, BIS values were lower during dexmedetomidine (M=55.10) than during sevoflurane (M=70.00) sessions (independent t test: t(18)=2.94, p<.01, d=1.39). Yet, even though dexmedetomidine sedation was accompanied by higher frequency of reports with subjective experiences than propofol sedation, the BIS values did not differ between these two agents during the loss of responsiveness period (independent t test: t(18)=1.07, p=.298). This finding indicates independence of the 'depth' of sedation required to maintain unresponsiveness as measured by the BIS index and the occurrence of subjective experiences during sedation in between-agent comparisons.

Within-agent comparison between sessions with the presence of subjective experiences and sessions with their absence during a period of dexmedetomidine-induced unresponsiveness showed higher BIS values and shallower sedation when experiences were present (M=58.60, SD=15.63) compared to sessions when they were absent (M=47.01, SD=12.80) (paired samples t test: t(4)=4.58, p=.01, d=2.05). Given that the depth of sedation did not differ between the presence and absence of subjective experiences during the induction phase (paired samples t test: t(4)=-.87, p>.05) or the recovery phase (paired samples t test: t(4)=1.14, p>.05) of dexmedetomidine sessions, these findings provide indirect evidence that subjective experiences occurred during the actual period of unresponsiveness. Similar inferential statistical analyses were not carried out with other anaesthetic agents due to a low number of observations in one of the comparison conditions.

6.3. Study II: Early-night serial awakenings as a new paradigm for studies on NREM dreaming (Noreika et al., 2009)

6.3.1. Methods

Aiming to develop an efficient paradigm for studying alterations of consciousness during sleep and to describe perceptual complexity and dynamics of NREM sleep dreaming, five volunteer students were recruited to take part in *Study II* (age: M=23.7, 4 females). Each of them spent one adaptation night and four non-consecutive experimental nights in the sleep laboratory. A novel paradigm, called the early-night serial awakenings (ENSA), was applied during experimental nights, with up to 12 awakenings each night from NREM sleep Stages 2 and 3 during the first 2–4 hours of sleep. Scoring of sleep stages was based on the recordings from two EOG and four EEG electrodes (C3, C4, O1, O2). To evaluate the homogeneity of data across the serial awakenings produced by the ENSA paradigm, a quantitative EEG analysis of possible changes in the spectral power of different frequency bands (delta: 0.5–4.0 Hz, theta: 4.0–8.0 Hz, alpha: 8.0–12.0 Hz, beta1: 12.0–16.0 Hz, beta2: 16.0–32.0 Hz) was carried out with the Brain Vision Analyzer (v1.05.0002) software.

After each awakening, participants were asked to report 'everything that was going through their mind before awakening'. If participants reported at least one perceptual experience, a set of further 18 questions was given, including questions about objects and the

self in a dream. Two blinded judges carried out content analysis of post-awakening reports and separated three recall categories: (1) reports of dreamless sleep, if participant did not report any perceptual experience; (2) reports of white dreaming, if participant remembered having a dream, but could not recall any concrete perceptual experiences; and (3) dream reports, if participant reported at least one perceptual experience. Complexity and temporal dynamics of perceptual experiences in dream reports were further analysed according to the *Orlinsky's Modified Scale for Perceptual Complexity of Dreams* (O-PC-Dreams, following Orlinsky, 1962). The O-PC-Dreams consists of 7 categories that describe an overall perceptual complexity of subjective reports, ranging from category 1, when participant recalls only a single isolated experience, to category 7, when he or she remembers a complex dynamic dream with several changes of scenery. Categories 1–4 refer to temporally static dreams, whereas categories 5–7 refer to temporally dynamic dream experiences (see Table 3).

Categories of perceptual	Categories of perceptual complexity	Examples
dynamics		
Static dream: person experiences only static percepts and/or never-	 Person remembers a specific topic but in isolation: a fragmentary percept, unrelated to anything else. 	'There was a boy.'
ending repetitive actions that do not accumulate	2. Person remembers several unrelated perceptual experiences.	'There was a boy as well as a sound of water flow.'
into major perceptual changes.	 Person remembers several interconnected perceptual experiences. 	'There were several boys sitting on the boat.'
	 Person remembers a short but coherent dream, the parts of which seem to be immersed within a unified scene. 	'There were several boys sitting on the boat in a river. It was a very sunny day.'
Dynamic dream: person experiences dynamic changes, during which something happens, followed by some	 Person remembers a coherent dream, in which something happens within a unified scene, i.e., one perceptual experience is replaced by another. 	'There were several boys sitting on the boat in a river. It was a very sunny day. Then one of them invited me to join them and so I did – I was very happy about it.'
consequence or in which one perceptual experience – object, scene, or main interacting character – is replaced by another.	 Person remembers a long, detailed dream in which the whole scene is replaced by another scenery. 	'There were several boys sitting on the boat in a river. It was a very sunny day. Then one of them invited me to join them and so I did – I was very happy about it. We navigated downstream for a while, and stopped near the riverside restaurant. The restaurant had Eastern décor with bright colours.'
	 Person remembers a long, detailed dream in which the whole scene is replaced by another scenery more than once. 	There were several boys sitting on the boat in a river. It was a very sunny day. Then one of them invited me to join them and so I did – I was very happy about it. We navigated downstream for a while, and stopped near the riverside restaurant. The restaurant had Eastern décor with bright colors. Suddenly, one stranger in the restaurant became very large and scary. I was afraid of him and ran outside, which appeared to be a city with shining lights and busy streets [etc.].'

Table 3: Orlinsky's Modified Scale for Perceptual Complexity of Dreams (O-PC-Dreams).

6.3.2. Results

The ENSA paradigm appeared to be practically feasible and data-wise very efficient. On average, 8.2 (SD=2.6) awakenings were performed during 20 experimental nights (164 awakenings as a total), with a mean total recording time of just 193 mins (SD=57.4) per single night. There were only 23.5 mins (SD=14.7) on average between two adjacent awakenings. During pilot stages of the project a concern was raised of whether participants will be able to fall asleep after serial awakenings due to repetitive disruption of normal circadian patterns. However, the opposite effect was observed in the main experiment: there was a strong negative correlation between the sleep latency and the count of previous awakenings (Spearman's rank order correlation: rho=-.98, n=8, p<.001), i.e., the more awakenings the participants underwent, the shorter time it took for them to fall asleep again. To investigate possible changes in spectral power between the early ENSA awakenings (1-4 awakenings) and the late ENSA awakenings (5–12 awakenings), separate 2×4×5 ANOVA tests were carried out for Stages 2 and 3 of NREM sleep with the following factors: Awakenings (2: early ENSA, late ENSA), Electrodes (C3, C4, O1, O2) and EEG bands (delta, theta, alpha, beta1, beta2). The main effect of Awakenings as well as the interactions between the Awakenings and Electrodes or EEG Bands were non-significant, indicating a high stability of spectral power patterns across the ENSA sessions.

27 awakenings and subjective reports were excluded from further analysis due to various technical problems, and a total of 137 NREM sleep reports were processed further for dream content analysis. Analysis of inter-rater judgments indicated strong agreement in scoring of reports for the recall categories (89.8% of reports; Cohen's kappa coefficient test: κ =.85) and for the categories of perceptual complexity of dreams (78%; κ=.88). Proportions of different types of reports between sleep stages were as follows: 23.8% of dreamless, 52.4% of white dreaming and 23.8% of dream reports after Stage 2 sleep awakenings; 37.3% of dreamless, 35.2% of white dreaming, and 27.5% of dream reports after awakenings from transitional stage of sleep (which was defined by the first appearance of delta EEG waves); and 29.6% of dreamless, 31.8% of white dreaming, and 38.6% of dream reports after Stage 3 sleep awakenings. Notably, percentage of white dream reports significantly decreased from Stage 2 to transition to Stage 3 sleep (one-way ANOVA: F(1.88,7.53)=7.15, p<.05, η^2 =.64; all pairwise comparisons: p<.05). These findings suggest that NREM Stage 3 might be the most suitable period of sleep for awakenings that aim to produce comparably high proportions of dreaming and dreamless sleep reports. Content analysis using O-PC-Dreams indicated that majority of 41 NREM dream reports were perceptually simple and static: five dreams fell under category 1, two dreams under category 2, eighteen dreams under category 3, fourteen dreams under category 4, and only two dreams fell under dynamic category 5.

To contrast EEG activity between dreaming and dreamless sleep, a follow up study was carried out using a higher density EEG setup (25 electrodes) and the same ENSA paradigm. A similar efficiency of the data collection was achieved with 10 new participants providing a total of 322 reports. Results of this study have not been included in this dissertation as they are still being analysed.

6.4. Study III: Dream bizarreness and waking thought in schizophrenia (Noreika et al., 2010a)

6.4.1. Methods

Bizarreness of dream experiences seems to remind psychotic perception and cognition observed in schizophrenia patients (Kelly, 1998), pointing to phenomenological similarities between naturally occurring and pathological ASCs. To test the bizarreness level in dreams of psychotic patients, and the resemblance of their waking and sleep mentation, five male schizophrenia inpatients (age: M=34.8, SD=6.3) and five male members of the nursing staff working at the same hospital (age: M=33.2, SD=6.5) took part in Study III. Inclusion criteria for the patients' group were unambiguous diagnosis of schizophrenia, current psychosis with positive symptoms, and well-preserved cognitive functioning, which was assessed by experienced psychiatrist. Cognitive functioning of all participants was tested with a detailed neuropsychological battery: WAIS tests Similarities, Digit span forward, Digit span backward, Digit symbol, Block design, 20 objects naming time, 20 objects immediate recall, 30 paired word associates; Benton Visual Memory test; clinical semantic and phonological fluency tests; and the Mini-Mental State Examination. For further assessment of memory functioning, in particular the recall of perceptual experiences, an ecologically valid experimental test was developed. Participants were asked to view a 5-min video recording of people behaving in ordinary situations, and afterwards to write down everything that they could recall from the video. Furthermore, to exclude possible sleep disorders, patients were monitored for one night with the static charge sensitive bed (SCSB) and actometers combined with video observation. Even though the SCSB-assessed quality of sleep varied from patient to patient, all patients had both quiet and active sleep, and there were no indications of sleep disorders.

Following clinical assessment, all participants were instructed to collect five dream reports by keeping a written dream diary after natural morning awakenings. Waking thought reports were collected on five different days by bringing participants to a silent dim room, letting them relax on a bed, and after 10 mins asking them to write down everything that was going through their mind. Four blind judges classified all dream and waking reports across two axes: origin of report (dream/waking thought) and reporter (patient/control). This classification was unguided, i.e., the judges were not provided with formal scoring guidelines and had to make judgments based on their own subjective criteria. In addition, two judges analysed dream reports more formally with the *Content Analysis of Bizarreness scale* (Revonsuo & Salmivalli, 1995). This scale evaluates ratios between the bizarre and the non-bizarre occurrences of basic dream elements for three different types of bizarreness: incongruity, which refers to a dream element inconsistent with waking reality; vagueness, which refers to an obscure or indeterminate element; and discontinuity, which refers to an element suddenly (dis)appearing or transforming. The agreement in bizarreness classification was 78% for the patients' and 86% for the controls' dream reports.

6.4.2. Results

Each of the control participants wrote down five dream reports, whereas the patients reported on average 6.8 dreams. The length of dream reports was significantly shorter in the

patient group (M=71 words, SD=64.5) than in the control group (M=201 words, SD=114.4) (Wilcoxon rank-sum test: W_s =727, p<.001). Arguably, the shorter length of dream reports in the patient group did not depend on the more mundane dream contents, but rather on the poorly functioning memory of semantic and perceptual experiences, which was evidenced by several memory tests. In particular, the patient group performed worse than the control group in the WAIS test for immediate recall of paired word associates (independent t test: t(8)=3.2, p<.05) and the Benton Visual Memory test (t(8)=2.9, p<.05). Furthermore, patients gave significantly shorter reports of the 5-min video recording (word count: M=54, SD=32) than did control participants (M=122.2, SD=41.5) (independent t test: t(8)=2.9, p<.05).

The blind judges made only one mistake in distinguishing dream reports from waking thought reports for the control group. However, the same judges misjudged that 52.9% of dream reports represent waking thoughts for the patients group compared to no mistakes for the controls (Pearson chi-square test: χ^2 =16.6, df=1, p<.001). Similarly, the judges decided that 48% of patients' waking thought reports are dreams, compared to 4% of such mistakes for the control group (χ^2 =10.4, df=1, p<.01). These findings indicate that overall content of reported subjective experiences did not differ between nocturnal dreaming and waking consciousness in psychotic schizophrenia patients. Regarding the reporter of subjective reports, the judges could not distinguish patients and controls, as many of the patients' waking reports were treated as standard dreams of control participants. Content analysis of dream reports using the Bizarreness scale showed that 33.2% of dream elements were bizarre in the patient group, compared to 20.7% of such elements in the control group (χ^2 =38.1, df=1, p<.001). The largest differences between the study groups were in the Incongruity (χ^2 =21.3, df=1, p<.001) and Vagueness (χ^2 =43.0, df=1, p<.001) categories of subjective content bizarreness.

6.5. Study IV: Dreaming and waking experiences in schizophrenia: How should the (dis)continuity hypotheses be approached empirically? (Noreika, 2011)

6.5.1. Literature review

Empirical findings regarding subjective contents of dreaming in schizophrenia patients are often interpreted in the framework of the *continuity hypothesis* (e.g., Kramer & Roth, 1973; Stompe et al., 2003; Zanasi et al., 2011a). Most generally, the continuity hypothesis suggests that subjective experiences in dreaming are modulated by or are a reflection of subjective experiences during wakefulness (Schredl, 2003). For instance, Schredl and Hofmann (2003) showed that the amount of time spent driving a car and reading during wakefulness correlates positively with frequency of these activities in dreams. Following a similar theoretical background, Zanasi et al. (2011a) compared dream reports of 123 schizophrenia patients and 123 healthy individuals using linguistic textual analysis methods. Some of their findings, such as a decreased frequency of hearing-related lemmas in schizophrenia dreams, were interpreted as reflecting schizophrenia-related impairments during wakefulness. In particular, the reduced frequency of hearing words in dream reports of schizophrenia patients was hypothetically related to the auditory memory deficits, which are frequently observed in schizophrenia patients during wakefulness (Zanasi et al., 2011a). In *Study IV*, the continuity interpretation of schizophrenic dreams was critically examined, arguing that it is not fully

substantiated in Zanasi et al. (2011a) as well as in several other studies on schizophrenia dreaming.

6.5.2. Original proposal

Conceptually, interaction between dream and waking experiences could have a form of 'continuity', 'discontinuity' or 'independence'. Continuity hypothesis would be supported by a positive correlation of rates of certain experiences, for example, meeting strangers, in dream and waking thought reports. Causally, continuity hypothesis could be further divided into the 'waking-to-dreaming' hypothesis, which states that a frequent occurrence of certain waking experiences increase the frequency of their occurrence in dreams (e.g., Schredl & Hofmann, 2003), and the 'dreaming-to-waking' hypothesis. In the context of psychosis studies, the latter hypothesis would state that certain dream experiences might intrude into waking consciousness (e.g., Hempel et al., 2003). Contrary to the continuity hypothesis, discontinuity hypothesis would predict a negative correlation between the frequencies of certain type of experiences in waking and dream reports (e.g., Cartwright, 1972), possibly referring to a hypothetical compensatory mechanism that stabilizes the total rate of certain experiences. For example, increase in the frequency of certain dream experiences might be related to their decrease during wakefulness in a case of the 'compensated waking' discontinuity, and the opposite would be true in a case of the 'compensated dreaming' discontinuity. Finally, the independence hypothesis would predict no association between waking and dreaming experiences.

Importantly, these possibilities are not mutually exclusive and while some of the experiences may show mismatch between different states of consciousness, other experiences may be continuous or vary independently between the states of consciousness under investigation. Thus, even though there is an increasing evidence for the continuity between waking and dreaming in psychosis (Arnulf et al., 2000; Hempel et al., 2003; Vita et al., 2008), some of the experiences may be discontinuous or independent. For example, the reduced frequency of hearing-related words in dream reports of schizophrenia patients (Zanasi et al., 2011a), which was originally interpreted by the authors as reflecting continuity with the auditory sensory memory deficits found in schizophrenia during wakefulness (e.g., Kasai et al., 2003), may have an alternative explanation. In particular, it is known that up to 74% of schizophrenia patients report elevated levels of auditory hallucinations, or 'voices', during wakefulness (Sartorius et al., 1974). Therefore, reduced hearing experiences in dreams might balance out increased levels of auditory hallucinations in waking life, as predicted by the hypothetical compensatory discontinuity of mental contents. Zanasi et al. (2011b) replied that such a suggestion does not take into consideration that schizophrenia patients tend to have not only auditory but also visual hallucinations, while no reduction of sight-related words was observed in their dream reports, which is not compatible with the 'compensatory hypothesis'. A further response to their reply would be that perceptual and cognitive interactions between different states of consciousness are multidimensional and that the discontinuity of auditory experiences does not exclude a possible independence or continuity of visual experiences.

6.6. Study V: How to integrate dreaming into a general theory of consciousness – A critical review of existing positions and suggestions for future research (Windt & Noreika, 2011)

6.6.1. Literature review

As the most frequently occurring natural ASC, dreaming could be regarded as an important phenomenon for testing predictions derived from the dominating theories of consciousness. However, recent attempts to integrate research of dreaming with the mainstream cognitive neuroscience of consciousness and cognition (Hobson et al., 2000; Nir & Tononi, 2010; Revonsuo, 2006; Schwartz & Maquet, 2002) did not yield yet straightforward results. The most programmatic yet rather contradictory suggestions treat dreaming as a model system continuous or similar either with the standard waking consciousness (Revonsuo, 2006) or with the pathological wake states (Hobson et al., 2000). Formally, dreaming can function as a research model of the explanatory target in at least three different ways: as a global model (when it replicates all or most of the central features of the explanatory target); as a restricted model (when it replicates some of the features of the explanatory target).

Perhaps the most detailed proposal of how to treat dreaming as a research model of waking consciousness was developed by Revonsuo (1995b, 2000, 2006), who argued that dreaming is a pure form of consciousness, isolated from ontologically confounding sensory inputs and motor outputs, yet containing the fundamental properties of consciousness, such as the presence of multimodal sensory experiences and an immersion in the world simulation. Given that subjective qualities of dreaming are continuous with the subjective qualities of waking consciousness (Revonsuo, 2006), the former could be regarded as a global model of the latter. From this perspective, analyses of the functional organization and neural mechanisms of dreaming are expected to identify the fundamental properties of waking consciousness. While agreeing on the usefulness of dreaming as a simplified domain for studying waking consciousness, Churchland (1988) took an antagonistic approach to that of Revonsuo (2006) and argued that there are characteristic differences between dreaming and wakefulness, such as the level of cognitive bizarreness. Consequently, dreaming can be treated as a model of wakefulness via negative analogy, which may reveal systematic differences between these states of consciousness. In this case modelling relationship may show that there is no single homogeneous underlying organizing principle of consciousness.

Hobson (1996, 1999, 2001) argued that hallucinations and cognitive deficiencies demarcate dreaming from the standard waking consciousness. However, dreaming can still be regarded as a global model of conscious mind, yet not of its intact form, but of pathological psychoses, for example, delirium tremens. In addition to similar phenomenology, both dreaming (or rather REM sleep) and clinical delirium are marked by a shift towards cholinergic neuromodulation of the brain when compared to a standard waking state (Hobson, 1999). Neurobiological resemblance between REM sleep and another psychopathology, schizophrenia, was further detailed by Gottesmann (2006), who argued that natural REM sleep is a remarkable collection of schizophrenia endophenotypes. Several other authors took a somewhat more modest position and proposed that dreaming could serve as a restricted

model of some aspects of psychopathologies rather than claiming an overall continuity between them. For instance, dreaming could serve to study the lack of involvement and emotion expression in schizophrenia (Hadjez et al., 2003), or several types of psychotic or neurological delusions that are common in dreams, such as age disorientation, paranoid experience of threats (Kelly, 1998), and delusional misidentifications, for example, mis- or hyper-identification of faces (Schwartz & Maquet, 2002).

Overall, there is very little consensus among researchers working on dreaming as to whether it can be treated as a model of wakefulness, and of what type of waking state: an intact consciousness or a psychotic mind. Furthermore, diverging views on the (dis)similarities between the waking and dreaming states result in the choice of different types of modelling relationships, for example, a global or a restricted modelling of waking consciousness if dreaming and wakefulness are regarded as similar (Revonsuo, 2006), and a modelling via negative analogy if they are regarded as dissimilar (Churchland, 1988).

6.6.2. Original proposal

Regarding proposals of using dreaming as a model of waking consciousness, which are often contradictory and even antagonistic (e.g., Hobson, 2001; Revonsuo, 2006), three types of problems were raised in Study V: (1) theoretical; (2) adequacy; and (3) practical feasibility concerns. Theoretical concerns deal with the concept of modelling and its representational relationship with the explanatory target. The strongest possible relationship is seen in the proposals of the global model of dreaming, some of which argue for the isomorphism, i.e., an identity between the model and its explanatory target, leaving little explanatory value in the modelling relationship. A more adequate relationship between a model and its explanatory target may be defined in terms of similarity, leaving dissimilar properties aside. Yet, the relevant degree of similarity is very difficult to define, as there is an infinite number of ways in which any two objects (or states of consciousness) can be similar or dissimilar (Wartofsky, 1979). Consequently, the relevant (dis)similarities have to be determined by researchers who may have different views regarding the key aspects of the mental states that can serve as a model, making the modelling relationship theory- and interest-dependent. For instance, the proposal for investigating dreaming as a model of natural waking consciousness presumes that dreaming is continuous with standard consciousness (Revonsuo, 2006). By contrast, the proposal of using dreaming as a model of pathological psychoses emphasizes neurophysiological and phenomenological dissimilarities between dreaming and standard waking states, and presumes continuity between dreaming and delirium tremens (Hobson, 1999, 2001).

Given that a widely accepted theory of dreaming is not yet in place, all uses of dreaming as a model of waking consciousness are naturally controversial. This raises a more general concern of whether it is adequate to use dreaming as a model of another conscious state when a scientific theory of dreaming itself is still limited, including its neural basis, function, and even definition (Windt, 2010). Arguably, scientific modelling can be successful only if a selected model is already well-understood, allowing the transfer of this knowledge to the understanding of the modelling target. Given that psychoses and waking consciousness are much more widely studied than dreaming, it is rather unlikely that a less researched dreaming can serve as a model of more explored states of consciousness. One may argue that this situation would rapidly change if more resources would be directed towards research of dreaming, however, a number of practical feasibility concerns raise doubt if a fast progress is likely to happen. A successful research model, such as the drosophila melanogaster model in genetics and biology (Arias, 2008), should provide large datasets in a relatively short time, observations and experimental manipulations with it should be relatively easy, and its running costs should be relatively low. Experimental research of dreaming seems to be an example of the opposite. Most notably, efficiency of data collection is very low, allowing for instance only four or five observations of REM sleep dreams during the whole night of recording, and even the more efficient ENSA paradigm enables to collect only about 10 NREM sleep reports in three hours (see Study II, Section 6.3), which is far from the efficiency of typical cognitive psychology paradigms that allow hundreds of observations (or trials) in a period of one hour or so (e.g., Study VII, Section 6.8). Furthermore, given that a neural marker of dreaming, such as P300 as a marker of conscious attention (Chennu et al., 2013), is unknown, the presence of dreaming can be observed only indirectly, through subjective reports that are affected by post-awakening amnesia, especially for schizophrenia patients (Okuma et al., 1970), making the dreaming model of psychosis particularly vulnerable to the mnemonic biases (see Sections 4.1 and 4.2). Manipulation with dream content, such as through a sensory stimulation at the sub-awakening threshold (e.g., Schredl et al., 2009) or through a transcranial direct current stimulation (tDCS) of cortex (as suggested in Study VI, see Section 6.7), is not always successful and often leads to somewhat unpredictable changes of dream content or even worse, to unintentional awakenings. Finally, laboratory research of dreaming requires long scanning hours and expensive personnel, partially explaining why there are so few neuroimaging studies of the neural mechanisms of dreaming.

Instead of a modelling relationship between dreaming and standard or abnormal waking states, which presumes a continuity between dreaming and waking consciousness, a more modest but also arguably a more promising approach is suggested in Study V, namely running a series of contrastive analyses studies (Baars, 1994) that would systematically compare dreaming and waking states. Contrary to the modelling approach, contrastive analyses do not rely on a unidirectional reasoning regarding relationship between two states under contrast, but instead, they systematically compare various possible similarities and dissimilarities between two contrasts. Instead of increasing the controversy between different proposals of dream modelling (Hobson, 2001; Revonsuo, 2006), contrastive analysis may be able to integrate diverging research lines by comparing dreaming with both standard waking state and pathological alterations of consciousness. In addition to a series of contrasts between dreaming and a passive standard or psychotic mentation (e.g., regarding self, cognitive bizarreness, metacognition, beliefs), a promising new endeavour may be analysis of (dis)similarities between dreaming (or lucid dreaming) and other natural, cognitively or pharmacologically induced ASCs, such as out-of-body experiences, hypnotic hallucinations or hallucinogenic psychomodulation. Arguably, contrastive analysis program may finally identify the most consistent similarities and dissimilarities between dreaming and waking states, which may also eventually lead to a restricted modelling program that would focus on a selected well-established contrast.

6.7. Study VI: New perspectives for the study of lucid dreaming: From brain stimulation to philosophical theories of self-consciousness (Noreika et al., 2010b)

6.7.1. Literature review

Cognitive deficiencies of REM sleep dreaming, such as a loss of rational thought, orientational instability and impaired self-reflection (Hobson et al., 2000; Kahn, 2007), are distinctive features of arguably the most intense natural ASC. It has been suggested that impaired dream cognition may be caused by hypoactivation of dorsolateral prefrontal cortex (DLPFC) during REM sleep (Hobson et al., 2000; Maquet et al., 2005; Muzur et al., 2002). However, not all of the REM sleep dreams are irrational (Meier, 1993). In particular, lucid dreaming is characterized by metacognitive insight into the hallucinatory nature of the dream state. It has been hypothesized that lucid dreaming is associated with a reactivation of DLPFC during REM sleep (Hobson et al., 2000; Kahn & Hobson, 2005), which was recently confirmed in an EEG/fMRI study by Dresler et al. (2012).

Lucid dreaming is thus dissimilar from both standard waking consciousness and typical REM sleep dreaming. In particular, perceptual experiences in lucid dreaming are hallucinatory unlike in wakefulness but similar to typical dreams, whereas cognitive abilities in lucid dreaming are much stronger than in non-lucid dreaming but somewhat similar to cognition in wakefulness. At the electrophysiological level of description, Voss et al. (2009) showed that lucid dreaming is associated with higher EEG gamma power over frontal regions when compared to non-lucid REM sleep dreams. When compared to waking state characterized by high alpha band coherence, lucid dreaming was found to have higher coherence in the delta and theta band spectrum. Following these observations, wakefulness, typical REM sleep dreaming, and lucid dreaming were suggested to be three sharply distinct states of consciousness (Hobson, 2009). However, differences between lucid and non-lucid dreaming may not be as sharp as previously suggested, but instead certain elements of dream lucidity may occur in non-lucid dreams (Windt, 2012; Windt & Metzinger, 2007), pointing to a gradual transition between different alterations of consciousness.

6.7.2. Original proposal

Even though lucid and non-lucid dreams seem to be very different with respect to their cognitive clarity, it is known that cognition in the form of speech and thought may occur in typical dreams (Kahn & Hobson, 2005; Meier, 1993), while lucid dreamers may sometimes lack full control of their actions and may show somewhat erratic reasoning (Brooks & Vogelsong, 1999; LaBerge & DeGracia, 2000). Arguably, there might be a slow gradation of different levels of cognition and insight from clearly non-lucid to the most intense lucid dreams. In particular, there seems to be a variety of the so called pre-lucid dreams, which are characterized by partial awareness of being in a dream (Brooks & Vogelsong, 1999). In the strongest sense of dream lucidity, it involves full cognitive clarity, including abilities to control the dream and to access autobiographical memory (Metzinger, 2009; Tart, 1988). This type of lucid dreams would be clearly different from the typical REM sleep dreams that have very limited and selective access to autobiographical memories (Windt & Metzinger, 2007). However, there may be many other types of dreams that are lucid in a weaker sense.

At least five different types of lucidity can be distinguished: attentional A-lucidity; behavioural B-lucidity; emotional E-lucidity; and cognitive C-lucidity; all of which together would form the full-scale F-lucidity (Windt & Metzinger, 2007). These distinctions are based on the availability of different types of information that reveal the virtual nature of a particular dream. A-lucidity takes place when a dreamer becomes capable of focusing their attention to the virtual fantasy-like aspects of dreaming. B-lucidity emerges when a dreamer gains a deliberate control over specific dream objects, characters or the whole dream plot, showing signs of lucid behaviour. E-lucidity is characterized by emotional reactions to dream events that would not be appropriate if these events would take place during wakefulness, but instead they are coherent given the virtual character of the dream. C-lucidity takes place when virtual reality of a dream becomes available to cognition and thought, and a dreamer begins to reason abstractly about the virtual nature of a dream. This type of lucidity forms the core of most of the definitions of lucid dreaming (Hobson, 2009; LaBerge & Gackenbach, 2000; Metzinger, 2009; Tart, 1988). Notably, A-, B- and E-lucidity types alone are not sufficient to form a convincing case of lucid dreaming; instead, they can be regarded as cases of pre-lucid dreams. Nevertheless, A-, B- and E-lucidity forms are required for the fully-fledged F-lucidity that would encompass all aspects of lucid dreaming.

Arguably, dream lucidity is a complex and graded phenomenon that ranges from isolated aspects of A-, B- or E- pre-lucid dreams to cognitively lucid dreams, in which a dreamer recognizes the virtual nature of a dream, to the full-scale lucidity with all aspects of mental functioning showing a coherent response to the internally generated dream events. Therefore, lucid dreaming is not sharply discontinuous from other types of dreaming, but instead its levels may fluctuate, i.e., increase or decrease, within a broader distribution of dream reports or even within a single dream. Thus, instead of dichotomous categorisation between lucid and non-lucid dreams, a more gradual measure could be applied to study lucid dreaming, such as the recently developed *Lucidity and Consciousness in Dreams scale* (LuCid; Voss et al., 2013). The use of such scales would allow studying certain aspects of dream lucidity as a gradual measure could be studied by a parametric interference with the natural brain functioning during REM sleep, such as by applying tDCS over DLPFC. Hypothetically, such stimulation would alter dream lucidity, and controlled manipulation with either the intensity or the duration of tDCS would result into a changing degree of lucidity.

6.8. Study VII: Perception of short time scale intervals in a hypnotic virtuoso (Noreika et al., 2012)

6.8.1. Methods

Among non-pathological ASCs, changes in time processing, which is a fundamental dimension of the stream of consciousness (see Section 2.2.1), are most frequently studied in hypnosis. Typically, hypnotized individuals underestimate supra-second intervals when compared to the duration estimates obtained from the baseline waking state, which has been linked to the increased absorption and limited attentional resources for monitoring time-related cues (for a review, see Naish, 2007). However, none of the previous hypnosis studies has investigated possible changes in the sub-second interval range, which provides temporal axis for feature binding, and which can be analysed with the currently available cognitive models of time perception (e.g., Wearden, 1992). To study duration perception in this interval range, a case study (*Study VII*) was carried out with an experienced hypnosis virtuoso T.S.-H. (female, 43 years old), who has previously taken part in a number of hypnosis studies (Fingelkurts An. et al., 2007; Fingelkurts Al. et al., 2007; Kallio et al., 1999, 2005, 2011). T.S.-H. can experience vivid auditory and visual hallucinations under hypnosis, and she immediately responds to a posthypnotic suggestion of being hypnotized again without any standard induction procedures, which is associated with only exceptionally susceptible individuals. In the present study hypnosis was induced and cancelled by a single word, which was suggested during previous hypnosis sessions.

Two behavioural experiments of auditory duration perception were carried out within and outside hypnosis using Inquisit 3 program (Millisecond Software, 2008). During the repeated-standards temporal generalization experiment, the participant listened to pairs of tones (500 Hz) consisting of a standard tone (either 600 ms or 1000 ms) followed by a probe tone (for 600 ms standard: 300, 450, 600, 750, or 900 ms; for 1000 ms standard: 500, 750, 1000, 1250, or 1500 ms). The participant was instructed to report by pressing keyboard buttons whether the two tones were the same or not. During the time estimation task, the participant listened to different durations around 3 seconds (2100 ms, 2400 ms, 2700 ms, 3000 ms, 3300 ms, 3600 ms, and 3900 ms; all 500 Hz). After each stimulus the participant reported by a button press whether the sound was three seconds long or not.

6.8.2. Results

In temporal generalization experiment, baseline and hypnosis conditions differed when probes were compared to the 600 ms standard, but not when probes were compared to the 1000 ms standard. In particular, fewer 'same' responses were given under hypnosis than in the baseline condition when the 600 ms standard was compared to a 450 ms probe (χ^2 =4.5, df=1, p<.05, Cramer's V=.24), whereas more 'same' responses were given under hypnosis than in the baseline condition when 600 ms standard was compared to a 750 ms probe (χ^2 =6.3, df=1, p<.05, Cramer's V=.28). Temporal generalization was inspected further by plotting the counts of 'same' responses against the probe duration divided by the standard duration in the 600 ms and 1000 ms conditions, separately for hypnosis and baseline states of consciousness. Temporal generalization gradients, i.e., functions representing 600 ms and 1000 ms conditions, did not superimpose well in the baseline state, suggesting violation of the scalar property of timing in a sub-second-to-second scale. On the contrary, temporal generalization gradients superimposed well under hypnosis, indicating more orderly interval timing in hypnosis than in the baseline waking state. Modelling of behavioural responses suggested that T.S.-H. had higher sensitivity to the sub-second intervals in hypnosis than in the baseline waking state. In addition, the distortion parameter showed the opposite memory bias for the 600 ms standard in the two states of consciousness: underestimation of standard in the baseline state, and overestimation of standard in hypnosis.

In the time estimation experiment, responses indicating '3-second' judgement peaked at 2100 ms in the baseline condition and at 2700 ms under hypnosis. These findings replicate

earlier reports showing that supra-second intervals are underestimated in hypnosis when compared to the baseline waking state (e.g., Naish, 2001). It seems that hypnosis facilitated sub-second and inhibited supra-second perception, providing further support to a theory of two distinct timing systems in the brain (Lewis & Miall, 2003).

6.9. Study VIII: Variability of duration perception: From natural and induced alterations to psychiatric disorders (Noreika et al., 2014)

6.9.1. Literature review

One of the most distinctive cognitive distortions, which may open up new possibilities for studying ASCs, is the change of perceived duration, as for instance reported by a hospitalized psychiatric patient: 'I can't estimate time. [...] I noticed my watch was accelerated... [...] The time you keep here isn't Greenwich time' (Lewis, 1932, p. 617–618). Distorted perception of duration has been reported in many pathological ASCs, including stimulant dependence (Wittmann et al., 2007), mania (Bschor et al., 2004), borderline personality disorder (Berlin et al., 2005), and most notably, schizophrenia (Carroll et al., 2008) and depression (Gil & Droit-Volet, 2009). Patients with schizophrenia tend to overestimate intervals as if the subjective time is going faster for them (Densen, 1977; Johnson & Petzel, 1971; Tysk, 1983a, 1983b; Wahl & Sieg, 1980), while patients with depression tend to underestimate durations (Gil & Droit-Volet, 2009), although, these findings are not always replicated (for schizophrenia, see Carroll et al., 2008; for depression, see Bschor et al., 2004). Time perception may also change in nonpathological ASCs, such as dreaming (Moiseeva, 1975) or LSD psychomodulation (Aronson et al., 1959). Likewise, research of duration perception under hypnosis reveals very consistent findings of underestimation of temporal intervals (Bowers, 1979; Naish, 2001; Noreika et al., 2012 (Study VII, Section 6.8); St. Jean & MacLeod, 1983; for a review, see Naish, 2007).

Notably, changes in perceived duration are not limited to alterations of consciousness. Contrary to this, they seem to be rather frequent, albeit less intense and somewhat unnoticed, in normal waking life. In has been shown in numerous studies that perception of duration changes depending on such diverse natural factors as: emotions (Angrilli et al., 1997; Gil et al., 2007), physical properties of stimuli (Wearden et al., 1998; Xuan et al., 2007), environmental effects of light (Aschoff & Daan, 1997; Morita et al., 2007), colour (Humphrey & Keeble, 1977), space (Collett, 1974), and menstrual (Montgomery, 1979; Morofushi et al., 2001) or circadian (Aritake-Okada et al., 2009; Kuriyama et al. 2005) cycles. For instance, Nakajima et al. (1998) instructed participants to sit on a chair for a period of 36 hours (this is not a typo!), and each second hour participants were asked to produce 10 s and 60 s intervals. The produced durations were the longest early in the morning, revealing a circadian modulation of time perception.

Numerous reports of cognitive instability of duration perception demonstrate the enormous flexibility of time processing mechanisms. Interestingly, behaviourally very similar if not identical changes of duration perception can be observed in various waking life situations and naturally occurring, cognitively or pharmacologically induced and pathological ASC, such as a comparable underestimation of duration in depression (Gil & Droit-Volet, 2009) and

hypnosis (Naish, 2001). What are the underlying causes of such distortions of duration perception, and would understanding them partially explain the relationship between different states of consciousness? Do behavioural similarities in duration perception imply that healthy individuals may occasionally undergo pathological abnormalities of duration perception, which can be otherwise observed in patients with neuropsychiatric disorders? Would it be possible to regard artificially induced changes of duration perception as a model of pathological alterations of time? To address these questions, a formal model of intra- and inter-individual distortions of duration perception was developed in *Study VIII*.

6.9.2. Original proposal

It was proposed in *Study VIII* that duration perception is a multifactorial function unifying a large number of externally (e.g., spatial properties of the environment) and internally (e.g., phase of menstrual cycle) driven factors. Thus, a perceived duration $D_{perceived}$ depends on the physical duration $D_{physical}$ and a number *n* of modulating factors x_i whose effects can be mathematically described by functions f_i . Thus, $D_{perceived}$ can be regarded as a product of $D_{physical}$ and all relevant functions f_n :

 $D_{perceived} = D_{physical} \cdot f_1(x_1) \cdot \ldots \cdot f_n(x_n)$

Each function f_i has its own range of possible values, e.g., f_1 may have a range of [0.6-0.9], while f_2 may have a range of [1.1-1.5]. Function values < 1 shorten $D_{perceived}$ and function values > 1 lengthen $D_{perceived}$, e.g., f_1 with a range of [0.6-0.9] will result in the shorter $D_{perceived}$ than $D_{physical}$. Distinct factors may also cancel each other's effects, e.g., if a value of $f_1(x_1)$ is 1.25 and the value of $f_2(x_2)$ is 0.8, and all other functions have a value of 1, then $D_{physical}$ of 10 sec yields a precise $D_{perceived}$:

$$D_{perceived} = D_{physical} \cdot f_1(x_1) \cdot f_2(x_2) \cdot \ldots \cdot f_n(x_n) = 10 \cdot 1.25 \cdot 0.8 \cdot 1 \cdot \ldots \cdot 1 = 10$$
 (sec)

Furthermore, factors capable of modulating duration perception can either be independent or form clusters of mutually dependent factors. For instance, Kuriyama et al. (2005) demonstrated that a fluctuation of produced intervals over a 30-hour period correlates with the core body temperature, suggesting that the circadian and temperature factors are associated. Contrary to this, factor of colour (Humphrey & Keeble, 1977) might be independent of the circadian and temperature factors:

$$D_{perceived} = D_{physical} \cdot f_{color}(x_{red}) \cdot \ldots \cdot f_n(x_n) \cdot f_{cycle}(x_{circadian}, x_{temperature}) \cdot \ldots \cdot f_m(x_m, y_m, \ldots)$$

Regarding the occurrence of distortions of duration perception, there are several routes of how functions can modulate $D_{perceived}$. First, $D_{perceived}$ differences between the standard and pathological states of consciousness might be related to the ordinary variability in duration processing, defined by the width of a possible range of function values. For instance, f_1 with a range of [0.5–1.5] in a healthy person might correspond to f_1 with a range of [0.5–1.1] in a depressed patient and a range of [0.9–1.5] in a psychotic patient. In line with these formal possibilities, Morofushi et al. (2001) reported that healthy women produced longer durations at 20:00 in the early luteal phase and at 08:00, 14:00, and 20:00 in the follicular phase of the menstrual cycle, when compared to other phases of the circadian and menstrual cycles. On the contrary, no within-participant variation of $D_{perceived}$ was observed in women with premenstrual syndrome, suggesting that a range of function values may have been narrowed in the premenstrual syndrome group. The range of function values may also broaden, e.g., from [0.5–1.5] to [0.2–1.6], or shift, e.g., from [0.5–1.5] to [0.4–1.4], between different states of consciousness or under different naturalistic conditions.

Another possible route for the occurrence of $D_{perceived}$ distortions is the involvement of different modulating factors. Factor reduction would take place if some of the functions modulating duration perception in one ASC would show a value of 1 in another state of brain/mind. Likewise, factor increase would take place if a factor with a value of 1 in one state would change its value in another state of consciousness. Obviously, a difference between factor reduction and factor increase is relative and depends on which state of consciousness is regarded as a baseline. Arguably, when a standard waking state is compared to a non-pathological or pathological ASC, the first one should be regarded as the baseline for a contrast between the states. More conceptual complications emerge when several ASCs are compared, in which case the baseline choice may depend on the research pragmatics, for example, the more investigated state could be regarded as a baseline state for another ASC.

Assuming that behavioural associations of D_{perceived} between pathological and nonpathological ASCs are functionally grounded, an intriguing possibility is to model D_{perceived} distortions in psychiatric disorders (e.g., depression) by means of inducing alterations of consciousness (e.g., hypnosis) in healthy individuals. However, conceptual analysis of the $D_{perceived}$ model shows that cognitive similarities between the naturally occurring, induced and pathological ASCs do not necessarily imply the same underlying functions and/or their values. Thus, convincing modeling of pathological distortions of D_{perceived} by studying healthy participants in non-pathological states of consciousness requires more direct evidence that the underlying variability sources are the same in both states of consciousness. Future studies should focus on the development of tools to identify different causes of $D_{perceived}$ variability. For instance, a broadening or a narrowing of the range of function values could be inferred from the respective increases or decreases in variation coefficients, e.g., intra-individual standard deviation of D_{perceived} scores. A shift of the range could be suspected from changes in the central tendency, e.g., mean of D_{perceived}. Factor reduction or increase could be inferred from significant group x factor interactions, when certain factors show an impact on $D_{perceived}$ in one group only. Until such studies are carried out, despite some intriguing similarities between duration perception in different alterations of consciousness, possibilities for their modeling relationship remain largely speculative.

7. DISCUSSION

The main aims of this dissertation were to advance theory of how different alterations of consciousness are related, and to propose new efficient experimental paradigms for studying subjective experiences and their changes in ASCs. The results were published in eight studies, four empirical (*Studies I–III, VII*, see Sections 6.2–6.4, 6.8) and four theoretical (*Studies IV–VI, VIII*, see Sections 6.5–6.7, 6.9). While theoretical studies focused largely on dreaming (*Studies IV–VI*, see Sections 6.5–6.7), which is the most common paradigmatic ASC, each of the empirical studies investigated subjective experiences and/or cognitive processes in different types of ASCs: naturally occurring dreaming (*Study II*, see Section 6.3), cognitively induced hypnosis (*Study VII*, see Section 6.8), pharmacologically induced sedation (*Study II*, see Section 6.4).

It was found that phenomenal consciousness does not necessarily diminish in unresponsive states of consciousness. In particular, research participants reported having had subjective experiences in more than 50% of unresponsive sedation sessions (Study I, see Section 6.2). Likewise, minimal static dreams were reported in about 30% of the early night NREM sleep awakenings (Study II, see Section 6.3). In Studies I and II, new methods were successfully introduced for the content analysis of subjective reports, namely the Orlinsky's Modified Scale for Perceptual Complexity of Dreams (O-PC-Dreams) and the Subjective Experiences during Anaesthesia Coding System (SEDA-Coding). A further methodological advancement was the application of a new early-night serial awakenings (ENSA) paradigm, which enabled the collection of a large number of dream reports in a relatively short time (Study II, see Section 6.3). It was also suggested that lucid dreaming, which is one of the most complicated ASCs to investigate due to its infrequency, could be studied by analysing different aspects of lucidity in the so-called pre-lucid dreams (Study VI, see Section 6.7). Content analysis of dreams of psychotic patients indicated higher bizarreness level when compared to dreams of healthy controls (Study III, see Section 6.4). Blind judges were not able to distinguish above chance level patients' dream and waking mentation reports, indicating a close phenomenological resemblance between psychotic dreaming and wakefulness (Study III, see Section 6.4). Yet, causal relationship between these states of consciousness remains uncertain. It was argued that dreaming may modulate subjective contents of waking consciousness, or likewise that waking consciousness may modulate dream experiences, or that these states of consciousness may be independent (Study IV, see Section 6.5). Regarding more ambitious suggestions to treat dreaming as a model of waking consciousness, a number of theoretical, adequacy and practical feasibility issues were raised, indicating that such suggestions are not likely to develop into influential research programs (Study V, see Section 6.6). Instead, more modest contrastive analyses were suggested between different states of consciousness, comparing their phenomenology, cognitive processes and neural mechanisms. Behavioural analysis and modelling of time perception in a hypnotic virtuoso found a more consistent perceptual timing under hypnosis than in the waking state (Study VII, see Section 6.8). Theoretical analyses of time perception studies led to the proposal of a new framework of how time distortions in different ASCs and natural fluctuations might be cognitively related, but have diverse underlying causes and mechanisms (Study VIII, see Section 6.9). In the following sections, the most important findings are discussed in more detail, placing them in a context of contemporary research of ASCs.

7.1. Methodological advancements

To map subjective experiences in unresponsive states of consciousness, two scoring systems were developed for the content analysis of subjective post-alteration reports: the SEDA-Coding (Study I, see Section 6.2) and the O-PC-Dreams (Study II, see Section 6.3). The SEDA-Coding was designed with the aim to assess the presence of different types of subjective qualities (e.g., the presence of visual consciousness) rather than the exact contents of consciousness (e.g., experience of individual visual images). The SEDA-Coding identifies different types of micro-level (e.g., auditory, sensorimotor sensations) and macro-level experiences (e.g., dream-like, out of body experiences). Micro-level experiences are atomistic and cannot be subdivided into further modalities, whereas macro-level experiences consist of micro-level experiences that may or may not form more complex macro-forms of consciousness. Compared to other available instruments for classification of intraoperative experiences (e.g., Mashour et al., 2010), the SEDA-Coding includes additional categories that clinically might be less relevant, for example, visual experiences or time perception, but may have fundamental importance for the experimental research of phenomenal consciousness during sedation and anaesthesia. Instead of focusing on different modalities of experiences, the O-PC-Dreams of Study II was designed with an aim to scale subjective dream reports according to their overall perceptual complexity, with score 1 referring to a single isolated experience and score 7 referring to the most complex sets of experiences within changing hallucinatory sceneries. Compared to the original version of Orlinsky's Sleep Mentation Scale (Farthing, 1992; Orlinsky, 1962), the O-PC-Dreams omitted the category of white dreaming because of its uncertain status, i.e. given that the contents of white dreaming are unknown, they cannot be correctly placed on the scale. Both of the developed content analysis methods proved to be reliable and yielded strong inter-judge agreement (Studies I, II, see Sections 6.2, 6.3). They also proved to be empirically meaningful and helped to identify important aspects of subjective experiences in unresponsive states, for example, the relatively high frequency of subjective experiences under dexmedetomidine sedation (Study I, see Section 6.2), and the predominantly static nature of the early night NREM sleep dreaming (Study II, see Section 6.3).

As an alternative to the use of formalized content analyses systems, which may be sensitive in identifying specific experiences under interest but may also miss other important aspects of conscious experiences, an unguided analysis was carried out when comparing sleep and waking mentation reports in *Study III* (see Section 6.3). Four independent blind judges were instructed to read the subjective reports and rate their origin (dreaming/waking) and reporter (schizophrenia patient/healthy control). Thus, instead of focusing on specific aspects of reports, judges rated the overall content of subjective experiences. This approach appeared to be successful and yielded significant results: while judges had no difficulties in differentiating between waking and dream reports of a control group, they could not distinguish above chance level reports of psychotic schizophrenia patients, suggesting a strong phenomenological similarity between dreaming and waking psychosis (see also Hadjez, 2003; Hobson, 2001; Skrzypińska & Szmigielska, 2013). Arguably, unguided classification of

subjective experiences into pre-determined dichotomous categories, such as waking/dreaming or patient/control (see Study *III*, see Section 6.4), might be a simple but efficient method to treat the first-person data in neurophenomenological research (Lutz & Thompson, 2003; Varela, 1996).

Research of ASCs requires establishment of a baseline state of consciousness, yet it is not always self-evident which state should be taken as a baseline condition for a particular alteration of consciousness. In the present dissertation, experimental feasibility of different contrast conditions was tested for three classical ASCs, hypnosis (Study VII, see Section 6.8), sedation (Study I, see Section 6.2) and dreaming (Study II, see Section 6.3). In hypnosis studies, perceptual or cognitive changes are often tested by giving suggestions that are expected to alter the brain function of interest (e.g., Mazzoni et al., 2009). It is usually expected that a contrast between cognitive or behavioural responses to the suggestions given in and out of hypnosis will reveal the underlying neurocognitive mechanisms of hypnotic suggestibility. Yet, even though such comparisons may reveal mechanisms of specific experiences under hypnosis, they are limited in revealing the neural mechanisms of a hypnotic state itself (Kallio & Revonsuo, 2003). Instead, a more promising contrast might be between a waking state of consciousness and a pure hypnotic state when no suggestions are given, which is sometimes called neutral hypnosis (Kallio & Revonsuo, 2003; Naish, 2007). In Study VII (see Section 6.8), such contrast was carried out in a single participant, hypnosis virtuoso T.S.-H., who has previously taken part in a series of EEG studies (Fingelkurts Al. et al., 2007; Fingelkurts An. et al., 2007; Kallio et al., 1999), serving as an efficient case model to study neural mechanisms of neutral hypnosis. However, it was not tested in previous studies if T.S.-H. is capable of carrying out long lasting, cognitively demanding behavioural tasks in a deep hypnotic state. In Study VII, the neutral hypnosis-induced changes in perception of sub-second intervals were reliably detected in a single participant when waking and hypnotic states were relatively rapidly switched, avoiding prolonged periods of hypnotic induction and recovery. Interestingly, T.S.-H. improved her time perception in hypnosis, pointing out that behavioural tasks for studying highly susceptible individuals should be programmed to avoid ceiling effects, as performance accuracy might significantly increase during hypnosis.

The choice of contrast conditions is particularly complicated when studying unresponsive states, as the confound of forgetting cannot be avoided when subjective reports are collected after regaining of responsiveness. In this dissertation, subjective experiences under sedation were successfully analysed by contrasting the effects of different anaesthetic/sedative agents. For instance, the frequency of laboratory-related experiences was found to be higher during sevoflurane compared to dexmedetomidine sedation (*Study I*, see Section 6.2). These findings suggest that there is no single route of transition from relaxed waking to unconscious anaesthesia, and that administration of different anaesthetic agents brings out different rates or types of subjective experiences until consciousness is lost. Another possible comparison condition for an unresponsive alteration of consciousness may be an unconscious state; arguably, contrasting them may reveal the neural correlates of consciousness (Revonsuo, 2006). In *Study II* (see Section 6.3), NREM sleep stage 3 was identified as the most suitable awakening time for experiments aiming to contrast similar proportions of dream and dreamless periods of sleep when the sleep stage is kept constant. Notably, it is particularly important to control the sleep stage in studies aiming to identify the neural correlates of

dreaming, because the frequently performed contrast between REM sleep dreaming and NREM dreamless sleep (e.g., Massimini et al., 2010) may turn out to be a hardly interpretable comparison of different sleep stages which vary in many different cognitive or neurophysiological functions.

While it is feasible to carry out case studies aiming to identify perceptual or cognitive changes during alterations of consciousness (*Study VII*, see Section 6.8), larger participant samples are necessary in order to draw more general conclusions. Unfortunately, practical constraints often limit data collection from large groups of participants undergoing natural or induced ASCs, which is particularly problematic in dream research (*Study V*, see Section 6.6). To increase efficiency of collection of dream reports in the sleep laboratory, a novel ENSA paradigm was tested in *Study II* (see Section 6.3). Continuous interruption of the natural development of sleep cycles proved to be an efficient way of collecting NREM sleep reports, yielding on average eight reports in three hours while maintaining the stable profile of pre-awakening EEG spectral power. Given that NREM sleep dreams appeared to be perceptually simple and static, the ENSA paradigm is suitable for studying minimal forms of consciousness. A recent extension of serial awakenings to the whole night recording can be applied to study more complex REM sleep dreams (Siclari et al., 2013).

7.2. Theoretical advancements

While the field of consciousness studies, especially visual awareness research, is dominated by experiments on specific isolated contents of consciousness (e.g., Koivisto & Revonsuo, 2010), an alternative approach to focus on the totality of subjective experiences in different states of consciousness may provide new perspectives for studying subjectivity. In particular, research of the overall stream of subjective experiences and their changes may reveal the global properties and internal structure of consciousness, which are not necessarily observable when focusing on a single isolated experience in standard behavioural experiments. For instance, Study II (see Section 6.3) showed that minimal forms of consciousness exist during early night NREM sleep: awakened participants reported seeing static visual images, suggesting that a broad perceptual or semantic context and temporal dynamics are not necessary for a conscious experience to arise. It seems that the stream of consciousness can stop developing; yet, conscious experiences do not necessarily cease. Furthermore, a follow-up study showed that the experience of time can arise irrespective of the absence of temporal changes of experiences, as research participants often rated their static dreams to last for 30 s, 1 min and even longer (Noreika et al., 2010c). These observations indicate that subjective experiences maintain the temporal structure even in the most minimal static forms of dreams, confirming that temporality, i.e., extension in time, is a fundamental property of phenomenal consciousness (see Section 2.2.1).

It has been suggested that dreaming, the most common and paradigmatic ASC, could be treated as a research model of other states consciousness (Hobson, 2001; Revonsuo, 2006). However, *Study V* (see Section 6.6) suggests that even though many perceptual and cognitive processes are similar between dreaming and psychosis (*Studies III, IV*, see Sections 6.4, 6.5) or between dreaming and standard waking consciousness (Revonsuo, 2006), proposals to use

dreaming as a research model of consciousness are not currently achievable due to theoretical, adequacy and practical feasibility limitations. Instead, the more modest phenomenological and neurophysiological contrasts between different alterations of consciousness are more likely to reveal the neurocognitive mechanisms shared by different ASCs. For instance, comparison of the overall stream of experiences in schizophrenia patients and healthy individuals showed that in some cases the psychotic and the dreaming minds might be indistinguishable (*Study III*, see Section 6.4). This finding points to the clinical importance of systematic contrasts between different states of consciousness. In particular, schizophrenia patients may experience similar delusions in their waking life and dreaming, although this does not seem to hold for each type of delusion (D'Agostino et al., 2013), showing simultaneous continuity and discontinuity between psychotic waking and dreaming states of consciousness (*Study IV*, see Section 6.5).

Two different types of contrasts between ASCs were considered in *Studies I–VIII*: a classical sharp contrast and a gradual or transitional contrast. A sharp contrast compares an overall ASC and a baseline waking state of consciousness (*Studies III, VII*, see Sections 6.4, 6.8), or several different alterations of consciousness (*Studies I–III*, see Sections 6.2–6.4), which is a standard approach in most of the contemporary studies of ASCs. A gradual contrast, suggested in *Study VI* (see Section 6.7), however, systematically maps different degrees of alteration, for example, the increasing levels of lucidity in dreams, which may be a more sensitive approach for detecting mild distortions of consciousness than a sharp contrast. In particular, a gradual contrast might be able to identify temporal dynamics of changing perceptual and cognitive functions in the transition from one to another state of consciousness.

As an alternative to the content analysis of subjective reports, different states of consciousness can be tested using the same behavioural paradigms and cognitive modelling (*Studies VII, VIII*, see Sections 6.8, 6.9). In this type of contrasting, a particular promising domain is perception of time. Given that temporal binding is the core of the stream of consciousness (Dainton, 2006), temporal distortions are very common in natural, induced or pathological alterations of consciousness (*Study VIII*, see Section 6.9). Arguably, a systematic comparison of such temporal distortions may reveal shared cognitive processes of different states of consciousness.

Unfortunately, despite the methodological advancements of the content analysis methods, reportability and cooperativeness constraints remain major obstacles for drawing decisive inferences about the presence or absence of consciousness. Furthermore, even if a conscious agent is cooperative and willing to report all experiences, some of them may not be available for conscious access due to phenomenological overflow of cognitive accessibility (Block, 1995, 2007, 2011, see also Sections 2.5.4 and 4.1). In cases when the contents of phenomenal consciousness emerge but are not available for access consciousness (Block, 1995), the trap of reportability leaves no room for a certainty in the assessment of subjective states. Arguably, it may remain impossible to distinguish a genuine dreamless sleep from sleep with forgotten dreaming until a necessary and sufficient neural correlate of phenomenal consciousness is determined, in which case the stream of consciousness would be assessed by the neural activity measures.

7.3. The future perspectives for studying alterations of consciousness

It is expected that future studies of ASCs, while continuing the development of the first-person data analyses, will focus on the neural mechanisms of spontaneously generated subjective experiences and their distortions. However, instead of correlative EEG or fMRI designs that dominate in the contemporary neuroimaging of ASCs, experimental paradigms enabling causal interference with the spontaneous neural functioning, such as by non-invasive brain stimulation methods TMS or tDCS (as suggested in *Study VI*, see Section 6.7), will become more popular. The first attempts to apply tDCS for studying the neural mechanisms of dreaming during REM sleep confirmed homology of brain regions supporting the same type of subjective experiences and brain functions in waking and dreaming: tDCS over prefrontal regions was associated with increased cognitive insight and lucidity in dreams (Stumbrys et al., 2013), stimulation of visual cortex induced higher rates of dreamed visual experiences (Jakobson et al., 2012), whereas inhibitory stimulation of sensorimotor cortex reduced the frequency of dreams with motor activity (Noreika et al., in preparation). Such studies may eventually indicate which brain regions, through their excitation or inhibition, lead to various distortions of subjective experiences in ASCs.

When combined with EEG or local field potential measurements, brain stimulation studies may also reveal what consequences the altered processing in these regions have for the consciousness-related electrodynamics of the widespread cortical networks. Recently, Casali and colleagues (2013) applied TMS pulses over different cortical regions during different states of consciousness, and measured complexity of widespread EEG responses to the TMS perturbation of spontaneous cortical activity. They found that perturbational complexity index (PCI) of information integration and differentiation in EEG signal, derived from the information integration theory of consciousness (Tononi, 2010, 2012, see Section 2.5.5) successfully distinguished different levels of consciousness in sedation, sleep and traumatic brain injury. Development of consciousness science in recent years suggests that information integration account of consciousness (Tononi, 2010, 2012) may become the most dominant neurocognitive theory of consciousness, even though it is not entirely compatible with the strong version of biological realism (see Sections 2.4.3 and 2.5.5). However, the current version of PCI does not seem to detect the essential theory-driven features of consciousness. In particular, PCI does not differ between the 'eyes closed' and 'eyes open' conditions in standard waking state (Casali et al., 2013), ignoring differentiation or phenomenal diversity of consciousness. Further developments of PCI and other similar EEG indexes should aim to be more sensitive to alterations in both states and contents of consciousness.

Neuroimaging studies aiming to contrast dreaming and dreamless sleep as conscious and unconscious states of mind, as suggested in *Study II* (see Section 6.3), will need to develop empirical arguments for rejection (or acceptance) of a possibility that dreamless sleep is forgotten dreaming and thus, a conscious state (Schredl et al., 2013). An ultimate test of it would be to check if neural correlates of dreaming, but not of active memory formation or working memory updating in response to the stream of internally generated experiences, are present during dreamless sleep, which would indicate the presence of dreaming, but absence of its recall. However, as long as the neural markers of dreaming or working memory during sleep are unknown, the status of dreamless sleep could be investigated indirectly, for

example, by studying neural mechanisms underlying white dreaming, when awakened participants have a feeling that they were dreaming, but cannot remember any further details about it. White dreaming is surprisingly frequent: participants report it when explicitly asked in about 40% of early night awakenings from NREM sleep stages 2 and 3 (Study II, see Section 6.3). White dreaming is also present, but to lesser extent, in sedation reports (Study I, see Section 6.2). Arguably, systematic contrasts between white dreaming and dreamless sleep as well as between white dreaming and recalled dreams may reveal whether white dreaming is more similar to dreams or dreamless sleep, eventually developing paradigms to map active memory updating while dreaming, such as those used to study waking memory consolidation and reactivation during sleep (Diekelmann et al., 2009; Schönauer et al., 2014). Recently, Eichenlaub et al. (2014) introduced a novel approach for studying the neural basis of dream recall and its failure: high and low dream recallers were compared using a passive auditory oddball paradigm during both wakefulness and sleep. High recallers showed higher P3a evoked EEG potentials to rare sounds, indicating a more efficient attention orientation to novel sensory information. Interestingly, the same ERP differences between the high and low dream recallers were observed during NREM sleep, suggesting that more efficient attentional processing is maintained during sleep, which may eventually lead to higher recall of dreamed perceptual experiences. Finally, carefully designed studies of behavioural training of dream recall in the sleep laboratory, such as presentation of sensory memory cues of possible dream content or enhancement of rapid orientation after awakening, may increase dream recall at the cost of decreasing white dreaming or even dreamless sleep. Such training would reduce the rate of misjudgements of recall categories and would improve the quality of a neuroimaging contrast between dreaming and dreamless sleep.

Interactions between different states of consciousness should also be studied using a gradual parametric approach, such as in the transition from non-lucid to pre-lucid to fully-lucid dreams (Study VI, see Section 6.7), or between dreams of increasing perceptual complexity (Study II, see Section 6.3). Perhaps one of the easiest transitions of conscious states to study and manipulate experimentally takes place during sleep onset, when cognitive functions and conscious experiences undergo changes from alert to relaxed to drowsy to unresponsive states of mind (Goupil & Bekinschtein, 2011). Recently, it was demonstrated that auditory perception undergoes spatial distortion in the drowsy state of consciousness, and healthy individuals begin showing neglect-like misjudgments regarding sounds coming from the right side of their perceptual space (Bareham et al., submitted). Given that space perception is closely associated with time perception and processing of other magnitudes (Fabbri et al., 2012; Walsh, 2003), it is likely that the latter would also be affected by the increasing sleepiness. Tracing down changes in these and other cognitive functions as well as the underlying neural processes may eventually reveal how an ASC emerges and develops in time. Some of the perceptual and cognitive changes might be unique for specific ASCs, such as susceptibility to external influences in hypnosis, whereas some other distortions seem to be shared by different ASCs, such as changes in time perception, observed in hypnosis, LSD psychomodulation, depression, schizophrenia (Studies VII, VIII, see Sections 6.8, 6.9), as well as several other mental disorders, including attention-deficit/hyperactivity disorder (Noreika et al., 2013) and autism spectrum disorders (Falter et al., 2012). Arguably, continuing research of ASCs will lead to better understanding, if not research models (Study V, see Section 6.6), of regular waking consciousness and cognition in some of these pathologies. Likewise, several

recent studies began to investigate interactions between ASCs and baseline waking states, which seem to share the same psychological and neural traits, such as increased attentional waking ability in frequent lucid dreamers (Blagrove et al., 2010) and high dream recallers (Eichenlaub et al., 2014, see also Ruby et al., 2013), or reduced amygdala volume in participants prone to bizarre dreaming (De Gennaro et al., 2011). Future studies may find that a relationship between an ASC and a standard waking state is bidirectional with both of them having the capability to influence each other (*Study IV*, see Section 6.5).

8. CONCLUSIONS

- 1. Phenomenal consciousness is dissociable from responsiveness, as subjective experiences do occur in unresponsive states of mind, including sedation (*Study I*, see Section 6.2) and sleep (*Study II*, see Section 6.3).
- 2. Efficiency of laboratory research of alterations of consciousness can be increased by serially interrupting NREM sleep (*Study II*, see Section 6.3), or by studying selected individuals who show extremely strong forms of an ASC, such as hypnotic virtuosos (*Study VII*, see Section 6.8).
- 3. Strong inter-judge agreement can be achieved using both guided (*Studies I, II*, see Sections 6.2, 6.3) and unguided (*Study III*, see Section 6.4) analyses of subjective reports, confirming empirical reliability of phenomenological analysis of alterations of consciousness. Relatively rare ASCs, such as lucid dreaming, might be better studied by a gradual contrast of subjective changes (*Study IV*, see Section 6.5).
- 4. Content-wise, dreaming and waking mentation reports collected from psychotic patients can be indistinguishable for blind judges (*Study III*, see Section 6.4), indicating a close resemblance of these states of consciousness.
- Despite phenomenological similarities, dreaming should not be treated as a uniform research model of psychotic or intact stream of consciousness (*Study V*, see Section 6.6). On the contrary, there seems to be a multiplicity of routes of how different states of consciousness, such as dreaming and psychosis, can be associated (*Study IV*, see Section 6.5).
- 6. Time perception distortions observed in hypnosis (*Study VII*, see Section 6.8) are also common in other alterations of consciousness (*Study VIII*, see Section 6.9). Yet, despite cognitive similarity, the mechanisms of such temporal changes might be different and cognitive modelling-driven comparative research is needed in order to identify the underlying routes of time distortions in different states of consciousness (*Study VIII*, see Section 6.9).

9. **REFERENCES**

- Aleksander, I. (2005). Machine consciousness. Progress in Brain Research, 150, 99–108.
- Alfuth, M., & Rosenbaum, D. (2011). Are diurnal changes in foot sole sensation dependent on gait activity? Neuroscience Letters, 504, 247–251.
- Alkire, M. T., Hudetz, A. G., & Tononi, G. (2008). Consciousness and anesthesia. Science, 322, 876–880.
- American Psychiatric Association (2000). *Diagnostic and Statistical Manual of Mental Disorders*, 4th Edition, Text Revision (DSM-IV-TR). Washington, DC: Author.
- Ames, D. R. (2005). Everyday solutions to the problem of other minds: Which tools are used when? In B. F.
 Malle & S. D. Hodges (Eds.), Other Minds: How Humans Bridge the Divide between Self and Others (pp. 158–173). New York, NY: Guilford Press.
- Angrilli, A., Cherubini, P., Pavese, A., & Manfredini, S. (1997). The influence of affective factors on time perception. *Perception & Psychophysics*, 59, 972–982.
- Antognini, J. F., & Carstens, E. (2002). In vivo characterization of clinical anaesthesia and its components. British Journal of Anaesthesia, 89, 156– 166.
- Arguedas, D., Langdon, R., & Stevenson, R. (2012). Neuropsychological characteristics associated with olfactory hallucinations in schizophrenia. *Journal of the International Neuropsychological Society*, 18, 799– 808.
- Arias, A. M. (2008). Drosophila melanogaster and the development of biology in the 20th century. Methods in Molecular Biology, 420, 1–25.
- Aritake-Okada, S., Uchiyama, M., Suzuki, H., Tagaya, H., Kuriyama, K., Matsuura, M., Takahashi, K., Higuchi, S., & Mishima, K. (2009). Time estimation during sleep relates to the amount of slow wave sleep in humans. *Neuroscience Research*, 63, 115–121.
- Arnulf, I., Bonnet, A.-M., Damier, P., Bejjani, B.-P., Seilhean, D., Derenne, J.-P., & Agid, Y. (2000). Hallucinations, REM sleep, and Parkinson's disease: A medical hypothesis. *Neurology*, 55, 281–288.
- Aronson, H., Silverstein, A. B., & Klee, G. D. (1959). Influence of lysergic acid diethylamide (LSD-25) on subjective time. *Archives of General Psychiatry*, 1, 469–472.
- Aschoff, J., & Daan, S. (1997). Human time perception in temporal isolation: Effects of illumination intensity. *Chronobiology International*, 14, 585–596.
- Aserinsky, E., & Kleitman, N. (1953). Regularly occurring periods of eye motility, and concomitant phenomena, during sleep. *Science*, 118, 273–274.

- Avidan, M. S., Zhang, L., Burnside, B. A., Finkel, K. J., Searleman, A. C., Selvidge, J. A., Saager, L., Turner, M. S., Rao, S., Bottros, M., Hantler, C., Jacobsohn, E., & Evers, A. S. (2008). Anesthesia awareness and the bispectral index. *New England Journal of Medicine*, 358, 1097–1108.
- Baars, B. J. (1988). A Cognitive Theory of Consciousness. New York, NY: Cambridge University Press.
- Baars, B. J. (1994). A thoroughly empirical approach to consciousness. *Psyche*, 1(6). Retrieved from http://www.theassc.org/files/assc/2278.pdf
- Baars, B. J. (1997). In the Theater of Consciousness: the Workspace of the Mind. New York, NY: Oxford University Press.
- Baars, B. J. (1998). Metaphors of consciousness and attention in the brain. *Trends in Neurosciences*, *21*, 58–62.
- Baars, B. J. (2002). The conscious access hypothesis: Origins and recent evidence. *Trends in Cognitive Sciences*, 6, 47–52.
- Baars, B. J. (2007). The global workspace theory of consciousness. In M. Velmans & S. Schneider (Eds.), *The Blackwell Companion to Consciousness* (pp. 236– 246). Malden, MA: Blackwell Publishing.
- Bachmann, T. (2000). Microgenetic Approach to the Conscious Mind. Amsterdam, Netherlands: John Benjamins.
- Baethge, C., Baldessarini, R. J., Freudenthal, K., Streeruwitz, A., Bauer, M., & Bschor, T. (2005). Hallucinations in bipolar disorder: Characteristics and comparison to unipolar depression and schizophrenia. *Bipolar Disorders*, 7, 136–145.
- Barrett, D., & McNamara, P. (Eds.) (2007). *The New Science of Dreaming* (Vol. 1-3). Westport, CT: Praeger Publishing.
- Bartels, A., & Zeki, S. (2004). The chronoarchitecture of the human brain: Functional anatomy based on natural brain dynamics and on the principle of functional independence. In R. S. J. Frackowiak, K. J. Friston, C. D. Frith, R. J. Dolan, C. J. Price, S. Zeki, J. Ashburner, & W. Penny (Eds.), *Human Brain Function* (pp. 201–229). San Diego, CA: Academic Press.
- Bartolomeo, P., Zieren, N., Vohn, R., Dubois, B., & Sturm, W. (2008). Neural correlates of primary and reflective consciousness of spatial orienting. *Neuropsychologia*, 46, 348–361.
- Beaulieu-Prévost, D., & Zadra, A. (2007). Absorption, psychological boundaries and attitude towards dreams as correlates of dream recall: Two decades of research seen through a meta-analysis. *Journal of Sleep Research*, 16, 51–59.

- Bechtel, W., & Mundale, J. (1999). Multiple realizability revisited: Linking cognitive and neural states. *Philosophy of Science*, 66, 175–207.
- Bechtel, W., & Richardson, R. C. (1993). Discovering Complexity: Decomposition and Localization as Strategies in Scientific Research. Princeton, NJ: Princeton University Press.
- Bentall, R. P. (2007). Clinical pathologies and unusual experience. In M. Velmans & S. Schneider (Eds.), *The Blackwell Companion to Consciousness* (pp. 130–140). Malden, MA: Blackwell Publishing.
- Bergman, I. J., Kluger, M. T., & Short, T. G. (2002). Awareness during general anaesthesia: A review of 81 cases from the Anaesthetic Incident Monitoring Study. *Anaesthesia*, 57, 549–556.
- Berkeley, J. (1710). A Treatise Concerning the Principles of Human Knowledge [reprinted in 1998]. Oxford, UK: Oxford University Press.
- Berlin, H. A., Rolls, E. T., & Iversen, S. D. (2005). Borderline personality disorder, impulsivity, and the orbitofrontal cortex. *The American Journal of Psychiatry*, 162, 2360–2373.
- Blagrove, M., Bell, E., & Wilkinson, A. (2010). Association of lucid dreaming frequency with Stroop task performance. *Dreaming*, 20, 280–287.
- Blanke, O. (2012). Multisensory brain mechanisms of bodily self-consciousness. *Nature Reviews Neuroscience*, 13, 556–571.
- Block, N. (1980). Troubles with functionalism. In N. Block (Ed.), *Readings in the Philosophy of Psychology* (Vol. 1, pp. 268–305). Cambridge, MA: Harvard University Press.
- Block, N. (1995). On a confusion about a function of consciousness. *Behavioral and Brain Sciences*, 18, 227–247.
- Block, N. (2007). Consciousness, accessibility and the mesh between psychology and neuroscience. *Behavioral and Brain Sciences*, 30, 481–499.
- Block, N. (2011). Perceptual consciousness overflows cognitive access. *Trends in Cognitive Sciences*, 15, 567–575.
- Bohm, D. (1990). A new theory of the relationship of mind and matter. *Philosophical Psychology*, 3, 271– 286.
- Bolender, J. (2001). An argument for idealism. Journal of Consciousness Studies, 8, 37–61.
- Boly, M., Faymonville, M.-E., Peigneux, P., Lambermont, B., Damas, P., Del Fiore, G., Degueldre, C., Franck, G., Luxen, A., Lamy, M., Moonen, G., Maquet, P., & Laureys, S. (2004). Auditory processing in severely brain injured patients: Differences between the minimally conscious state and the persistent vegetative state. Archives of Neurology, 61, 233–238.

- Boly, M., Faymonville, M.-E., Schnakers, C., Peigneux, P., Lambermont, B., Phillips, C., Lancellotti, P., Luxen, A., Lamy, M., Moonen, G., Maquet, P., & Laureys, S. (2008). Perception of pain in the minimally conscious state with PET activation: An observational study. *Lancet Neurology*, 7, 1013–1020.
- Boly, M., Sanders, R. D., Mashour, G. A., & Laureys, S. (2013). Consciousness and responsiveness: Lessons from anaesthesia and the vegetative state. *Current Opinion in Anaesthesiology*, 26, 444–449.
- Bower, J. M. (2002). The organization of cerebellar cortical circuitry revisited: Implications for function. *Annals of the New York Academy of Sciences*, 978, 135–155.
- Bowers, K. S. (1979). Time distortion and hypnotic ability: Underestimating the duration of hypnosis. *Journal of Abnormal Psychology*, *88*, 435–439.
- Braun, J., & Julesz, B. (1998). Withdrawing attention at little or no cost: Detection and discrimination tasks. *Perception & Psychophysics*, 60, 1–23.
- Brooks, J. E., & Vogelsong, J. A. (1999). The Conscious Exploration of Dreaming. Discovering How We Create and Control Our Dreams. Bloomington, IN: First Books Library.
- Brouwer, G. J., van Ee, R., & Schwarzbach, J. (2005). Activation in visual cortex correlates with the awareness of stereoscopic depth. *Journal of Neuroscience*, 9, 10403–10413.
- Bschor, T., Ising, M., Bauer, M., Lewitzka, U., Skerstupeit, M., Müller-Oerlinghausen, B., & Baethge, C. (2004). Time experience and time judgment in major depression, mania and healthy subjects. A controlled study of 93 subjects. Acta Psychiatrica Scandinavica, 109, 222–229.
- Busch, N. A., Dubois, J., & VanRullen, R. (2009). The phase of ongoing EEG oscillations predicts visual perception. *Journal of Neuroscience*, 29, 7869–7876.
- Calkins, M. W. (1893). Statistics of dreams. American Journal of Psychology, 5, 311–343.
- Carroll, C. A., Boggs, J., O'Donnell, B. F., Shekhar, A., & Hetrick, W. P. (2008). Temporal processing dysfunction in schizophrenia. *Brain and Cognition*, 67, 150–161.
- Cartwright, R. D. (1972). Sleep fantasy in normal and schizophrenic persons. *Journal of Abnormal Psychology*, 80, 275–279.
- Casali, A. G., Gosseries, O., Rosanova, M., Boly, M., Sarasso, S., Casali, K. R., Casarotto, S., Bruno, M.-A., Laureys, S., Tononi, G., & Massimini, M. (2013). A theoretically based index of consciousness independent of sensory processing and behavior. *Science Translational Medicine*, 5:198ra105. doi:10.1126/scitranslmed.3006294
- Cavallero, C., Cicogna, P., Natale, V., Occhionero, M., & Zito, A. (1992). Slow wave sleep dreaming. *Sleep, 15*, 562–566.

- Cermolacce, M., Sass, L., & Parnas, J. (2010). What is bizarre in bizarre delusions? A critical review. *Schizophrenia Bulletin, 36*, 667–679.
- Chalmers, D. J. (1995). Facing up to the problem of consciousness. *Journal of Consciousness Studies*, 2, 200–219.
- Chalmers, D. J. (1996). *The Conscious Mind*. New York, NY: Oxford University Press.
- Chalmers, D. J. (2000). What is a neural correlate of consciousness? In T. Metzinger (Ed.), Neural Correlates of Consciousness: Empirical and Conceptual Questions (pp. 31–63). Cambridge, MA: MIT Press.
- Chennu, S., Noreika, V., Gueorguiev, D., Blenkmann, A., Kochen, S., Ibáñez, A., Owen, A. M., & Bekinschtein, T. A. (2013). Expectation and attention in hierarchical auditory prediction. *Journal of Neuroscience*, 33, 11194–11205.
- Churchland, P. S. (1988). Reduction and the neurobiological basis of consciousness. In A. Marcel & E. Bisiach (Eds.), *Consciousness in Contemporary Science* (pp. 273-304). Oxford, UK: Clarendon Press.
- Churchland, M. P. (1999). *Matter and Consciousness*. Cambridge, MA: MIT Press.
- Clark, A. (2000). A Theory of Sentience. Oxford, UK: Oxford University Press.
- Clark, A. (2001). Mindware: An Introduction to the Philosophy of Cognitive Science. New York, NY: Oxford University Press.
- Clark, A. (2009). Perception, action, and experience: Unraveling the golden braid. *Neuropsychologia*, 47, 1460–1468.
- Clark, L. A., Watson, D., & Leeka, J. (1989). Diurnal variation in the positive affects. *Motivation and Emotion*, 13, 205–234.
- Clavagnier, S., Falchier, A., & Kennedy, H. (2004). Longdistance feedback projections to area V1: Implications for multisensory integration, spatial awareness, and visual consciousness. *Cognitive, Affective, & Behavioral Neuroscience, 4*, 117–126.
- Collett, B. A. (1974). Variation in body temperature, perceived duration and perceived personal space. *International Journal of Nursing Studies*, 11, 47–60.
- Corsi-Cabrera, M., Miró, E., del-Río-Portilla, Y., Pérez-Garci, E., Villanueva, Y., & Guevara, M. A. (2003). Rapid eye movement sleep dreaming is characterized by uncoupled EEG activity between frontal and perceptual cortical regions. *Brain and Cognition*, *51*, 337–345.
- Crane, T., & Patterson, S. (2000). History of the Mind-Body Problem. London Studies in the History of Philosophy (Vol 3). London, UK: Routledge.
- Craver, C. F. (2007). Explaining the Brain: Mechanisms and the Mosaic Unity of Neurosciences. New York, NY: Oxford University Press.

- Craver, C. F., & Darden, L. (2001). Discovering mechanisms in neurobiology. The case of spatial memory. In P. K. Machamer, R. Grush, & P. McLaughlin (Eds.), *Theory and Method in Neuroscience* (pp. 112–137). Pittsburgh, PA: University of Pittsburgh Press.
- Crick, F. (1994). *The Astonishing Hypothesis*. New York, NY: Scribner.
- Critchley, H. D., Wiens, S., Rotshtein, P., Öhman, A., & Dolan, R. J. (2004). Neural systems supporting interoceptive awareness. *Nature Neuroscience*, 7, 189–195.
- Cruse, D., Chennu, S., Chatelle, C., Bekinschtein, T. A., Fernández-Espejo, D., Pickard, J. D., Laureys, S., & Owen, A. M. (2011). Bedside detection of awareness in the vegetative state: A cohort study. *Lancet, 378*, 2088–2094.
- D'Agostino, A., Aletti, G., Carboni, M., Cavallotti, S., Limosani, I., Manzone, M., & Scarone, S. (2013). Are delusional contents replayed during dreams? *Consciousness and Cognition*, 22, 708–715.
- Dainton, B. (2006). *Stream of Consciousness: Unity and Continuity in Conscious Experience* (2nd ed.). Abingdon, UK: Routledge.
- David, C. N., Greenstein, D., Clasen, L., Gochman, P., Miller, R., Tossell, J. W., Mattai, A. A., Gogtay, N., & Rapoport, J. L. (2011). Childhood onset schizophrenia: High rate of visual hallucinations. *Journal of the American Academy of Child & Adolescent Psychiatry*, 50, 681–686.
- De Gennaro, L., Cipolli, C., Cherubini, A., Assogna, F., Cacciari, C., Marzano, C., Curcio, G., Ferrara, M., Caltagirone, C., & Spalletta, G. (2011). Amygdala and hippocampus volumetry and diffusivity in relation to dreaming. *Human Brain Mapping*, 32, 1458–1470.
- Dehaene, S., Naccache, L., Cohen, L., Bihan, D. L., Mangin, J. F., Poline, J. B., & Rivière, D. (2001). Cerebral mechanisms of word masking and unconscious repetition priming. *Nature Neuroscience*, 4, 752–758.
- Dehaene, S., Changeux, J.-P., Naccache, L., Sackur, J., & Sergent, C. (2006). Conscious, preconscious, and subliminal processing: A testable taxonomy. *Trends in Cognitive Sciences*, 10, 204–211.
- Dehaene, S., & Changeux, J. P. (2011). Experimental and theoretical approaches to conscious processing. *Neuron*, *70*, 200–227.
- Delespaul, P., deVries, M., & van Os, J. (2002). Determinants of occurrence and recovery from hallucinations in daily life. *Social Psychiatry and Psychiatric Epidemiology*, *37*, 97–104.
- Dennett, D. (1978). Brainstorms. Hassocks, UK: Harvester Press.
- Dennett, D. (1991). *Consciousness Explained*. Boston, MA: Little, Brown & Co.

- Densen, M. E. (1977). Time perception and schizophrenia. *Perceptual and Motor Skills, 44,* 436–438.
- Descartes, R. (1641). Meditationes de prima philosophia, in qua Dei existentia et animæ immortalitas demonstratur. In L. I. Lafleur (Trans.). (1960). Meditations on First Philosophy. New York, NY: Bobbs-Merrill Company.
- Diekelmann, S., Wilhelm, I., & Born, J. (2009). The whats and whens of sleep-dependent memory consolidation. *Sleep Medicine Reviews*, 13, 309–321.
- Di Russo, F., Martínez, A., Sereno, M. I., Pitzalis, S., & Hillyard, S. A. (2002). Cortical sources of the early components of the visual evoked potential. *Human Brain Mapping*, 15, 95–111.
- Donders, F. C. (1868). Over de snelheid van psychische processen. In W. G. Koster (Trans.). (1969). On the speed of mental processes. Acta Psychologica, 30, 412–431.
- Dresler, M., Wehrle, R., Spoormaker, V. I., Koch, S. P., Holsboer, F., Steiger, A., Obrig, H., Sämann, P. G., & Czisch, M. (2012). Neural correlates of dream lucidity obtained from contrasting lucid versus non-lucid REM sleep: A combined EEG/fMRI case study. *Sleep*, 35, 1017–1020.
- Dubois, J., & VanRullen, R. (2011). Visual trails: Do the doors of perception open periodically? *PLoS Biology*, 9(5): e1001056. doi:10.1371/journal.pbio.1001056
- Düzel, E., Yonelinas, A. P., Mangun, G. R., Heinze, H. J., & Tulving, E. (1997). Event-related brain potential correlates of two states of conscious awareness in memory. *Proceedings of the National Academy of Sciences*, 27, 5973–5978.
- Eagleman, D. M., & Sejnowski, T. J. (2000). Motion integration and postdiction in visual awareness. *Science*, 287, 2036–2038.
- Eichenlaub, J.-B., Bertrand, O., Morlet, D., & Ruby, P. (2014). Brain reactivity differentiates subjects with high and low dream recall frequencies during both sleep and wakefulness. *Cerebral Cortex, 24*, 1206-1215.
- Engel, A. K., Fries, P., & Singer, W. (2001). Dynamic predictions: Oscillations and synchrony in top-down processing. *Nature Reviews Neuroscience*, 2, 704–716.
- Engel, A. K., & Singer, W. (2001). Temporal binding and the neural correlates of sensory awareness. *Trends in Cognitive Sciences*, 5, 16–25.
- Erlacher, D., & Schredl, M. (2004). Time required for motor activity in lucid dreams. *Perceptual and Motor Skills*, 99, 1239–1242.
- Errando, C. L., Sigl, J. C., Robles, M., Calabuig, E., García, J., Arocas, F., Higueras, R., Del Rosario, E., López, D., Peiró, C. M., Soriano, J. L., Chaves, S., Gil, F., & García-Aguado, R. (2008). Awareness with recall during general anaesthesia: A prospective observational evaluation of 4001 patients. *British Journal of Anaesthesia*, 101, 178–185.

- Esposito, M. J., Nielsen, T. A., & Paquette, T. (2004). Reduced Alpha power associated with the recall of mentation from Stage 2 and Stage REM sleep. *Psychophysiology*, 41, 288–297.
- Fabbri, M., Cancellieri, J., & Natale, V. (2012). The A Theory Of Magnitude (ATOM) model in temporal perception and reproduction tasks. Acta Psychologica, 139, 111–123.
- Faillace, L. A., & Szara, S. (1968). Hallucinogenic drugs: Influence of mental set and setting. *Diseases of the Nervous System*, 29, 124–126.
- Falter, C. M., Noreika, V., Wearden, J. H., & Bailey, A. J. (2012). More consistent, yet less sensitive: Interval timing in autism spectrum disorders. *Quarterly Journal of Experimental Psychology*, 65, 2093–2107.
- Farris, C., Treat, T. A., & Viken, R. J. (2010). Alcohol alters men's perceptual and decisional processing of women's sexual interest. *Journal of Abnormal Psychology*, 119, 427–432.
- Farthing, G. W. (1992). *The Psychology of Consciousness*. Englewood Cliffs, NJ: Prentice-Hall.
- Fechner, G. T. (1860). *Elemente der Psychophysik* [Elements of Psychophysics]. Leipzig, Germany: Breitkopf und Härtel.
- Fingelkurts Al. A., Fingelkurts An. A., Kallio S., & Revonsuo A. (2007). Hypnosis induces a changed composition of brain oscillations in EEG: A case study. *Contemporary Hypnosis, 24*, 3–18.
- Fingelkurts An. A., Fingelkurts Al. A., Kallio S., & Revonsuo A. (2007). Cortex functional connectivity as a neurophysiological correlate of hypnosis: An EEG case study. *Neuropsychologia*, 45, 1452–1462.
- Fisk, G. D., & Haase, S. J. (2005). Unconscious perception or not? An evaluation of detection and discrimination as indicators of awareness. *American Journal of Psychology*, 118, 183–212.
- Fodor, J. A. (1975). *The Language of Thought*. Hassocks, UK: Harvester Press.
- Foulkes, D. (1996). Dream research: 1953-1993. *Sleep*, 19, 609–624.
- Franklin, S. (2003). IDA: A conscious artifact? Journal of Consciousness Studies, 10, 47–66.
- Freeman, D., Pugh, K., Vorontsova, N., Antley, A., & Slater, M. (2010). Testing the continuum of delusional beliefs: An experimental study using virtual reality. *Journal of Abnormal Psychology*, 119, 83–92.
- Freeman, D., McManus, S., Brugha, T., Meltzer, H., Jenkins, R., & Bebbington, P. (2011). Concomitants of paranoia in the general population. *Psychological Medicine*, 41, 923–936.
- Frith, C., & Rees, G. (2007). A brief history of the scientific approach to the study of consciousness. In M. Velmans & S. Schneider (Eds.), *The Blackwell Companion to Consciousness* (pp. 9–22). Malden, MA: Blackwell Publishing.

- Gazzaniga, M. S. (2005). Forty-five years of split-brain research and still going strong. *Nature Reviews Neuroscience*, 6, 653–659.
- Gil, S., Niedenthal, P. M., & Droit-Volet, S. (2007). Anger and time perception in children. *Emotion*, 7, 219–225.
- Gil, S., & Droit-Volet, S. (2009). Time perception, depression and sadness. *Behavioural Processes*, 80, 169–176.
- Gilbert, D. G. (1979). Paradoxical tranquilizing and emotion-reducing effects of nicotine. *Psychological Bulletin, 86,* 643–661.
- Gillespie, G. (2002). Dreams and dreamless sleep. *Dreaming*, 12, 199–207.
- Goekoop, J. G., de Winter, R. F., Wolterbeek, R., Spinhoven, P., Zitman, F. G., & Wiegant, V. M. (2009). Reduced cooperativeness and reward-dependence in depression with above-normal plasma vasopressin concentration. *Journal of Psychopharmacology*, 23, 891–897.
- González, D., Riba, J., Bouso, J. C., Gómez-Jarabo, G., & Barbanoj, M. J. (2006). Pattern of use and subjective effects of Salvia divinorum among recreational users. *Drug and Alcohol Dependence*, 85, 157–162.
- Goodale, M. A., & Milner, A. D. (1992). Separate visual pathways for perception and action. *Trends in Neurosciences*, 15, 20–25.
- Goupil, L., & Bekinschtein, T. A. (2011). Cognitive processing during the transition to sleep. Archives Italiennes de Biologie, 150, 140–154.
- Gottesmann, C. (2006). The dreaming sleep stage: A new neurobiological model of schizophrenia? *Neuroscience*, 140, 1105–1115.
- Green, J. P., Barabasz, A. F., Barrett, D., & Montgomery, G. H. (2005). Forging ahead: The 2003 APA Division 30 definition of hypnosis. *International Journal of Clinical* and Experimental Hypnosis, 53, 259–264.
- Gruzelier, J. (1998). A working model of the neurophysiology of hypnosis: A review of evidence. *Contemporary Hypnosis*, 15, 5–23.
- Gruzelier, J. (2005). Altered states of consciousness and hypnosis in the twenty-first century. *Contemporary Hypnosis*, 22, 1–7.
- Gutschalk, A., Micheyl, C., & Oxenham, A. J. (2008). Neural correlates of auditory perceptual awareness under informational masking. *PLoS Biology 6*(6): e138. doi:10.1371/journal.pbio.0060138
- Hadjez, J., Stein, D., Gabbay, U., Bruckner, J., Meged, S., Barak, Y., Elizur, A., Weizman, A., & Rotenberg, V. S. (2003). Dream content of schizophrenic, nonschizophrenic mentally ill, and community control adolescents. *Adolescence*, *38*, 331–342.
- Hall, C. S., & Van de Castle, R. L. (1966). *The Content* Analysis of Dreams. New York, NY: Appleton Century Crofts.

- Hameroff, S. R. (2007). The brain is both neurocomputer and quantum computer. *Cognitive Science*, 31, 1035– 1045.
- Hartmann, W. M., & Wittenberg, A. (1996). On the externalization of sound images. *Journal of the Acoustical Society of America*, 99, 3678–3688.
- Haynes, J. D., & Rees, G. (2005). Predicting the stream of consciousness from activity in human visual cortex. *Current Biology*, 15, 1301–1307.
- Heinämaa, S., Lähteenmäki, V. & Remes, P. (Eds.). (2007). Consciousness: From Perception to Reflection in the History of Philosophy. Studies in the History of Philosophy of Mind (Vol. 4). Dordrecht, Netherlands: Springer.
- Heinke, W., & Schwarzbauer, C. (2002). In vivo imaging of anaesthetic action in humans: Approaches with positron emission tomography (PET) and functional magnetic resonance imaging (fMRI). British Journal of Anaesthesia, 89, 112–122.
- Hempel, A. G., Felthous, A. R., & Meloy, J. R. (2003). Psychotic dream-related aggression: A critical review and proposal. Aggression and Violent Behavior, 8, 599–620.
- Hilgard, E. R. (1965). *Hypnotic Susceptibility*. New York, NY: Harcourt, Brace & World.
- Hobson, J. A. (1988). *The Dreaming Brain*. New York, NY: Basic Books.
- Hobson, J. A. (1996). How the brain goes out of its mind. Endeavour, 20, 86–89.
- Hobson, J. A. (1999). Dreaming as Delirium How the Brain Goes Out of Its Mind. Cambridge, MA: MIT Press.
- Hobson, J. A. (2001). The Dream Drugstore: Chemically Altered States of Consciousness. Cambridge, MA: MIT Press.
- Hobson, A. (2009). The neurobiology of consciousness: Lucid dreaming wakes up. International Journal of Dream Research, 2, 41–44.
- Hobson, J. A., Pace-Schott, E. F., & Stickgold, R. (2000). Dreaming and the brain: Toward a cognitive neuroscience of conscious states. *Behavioral and Brain Sciences*, 23, 793–842.
- Hobson, J. A., & Friston, K. J. (2012). Waking and dreaming consciousness: Neurobiological and functional considerations. *Progress in Neurobiology*, 98, 82–98.
- Horikawa, T., Tamaki, M., Miyawaki, Y., & Kamitani, Y. (2013). Neural decoding of visual imagery during sleep. *Science*, 340, 639–642.
- Hsieh, P.-J., & Colas, J. T. (2012). Awareness is necessary for extracting patterns in working memory but not for directing spatial attention. *Journal of Experimental Psychology: Human Perception and Performance, 38*, 1085–1090.

- Huang, L., Treisman, A., & Pashler, H. (2007). Characterizing the limits of human visual awareness. *Science*, 317, 823–825.
- Humphrey, N. K., & Keeble, G. R. (1977). Do monkeys' subjective clocks run faster in red light than in blue? *Perception*, 6, 7–14.
- Hurlburt, R. T., & Heavy, C. L. (2004). To beep or not to beep: Obtaining accurate reports about awareness. *Journal of Consciousness Studies*, 11, 113–128.
- Iber, C., Ancoli-Israel, S., Chesson, A., & Quan, S. F. (2007). The AASM Manual for the Scoring of Sleep and Associated Events. Westchester, IL: American Academy of Sleep Medicine.
- Intaitė, M., Noreika, V., Šoliūnas, A., & Falter, C. M. (2013). Interaction of bottom-up and top-down processes in the perception of ambiguous figures. *Vision Research*, 89, 24–31.
- Jackson, F. (1986). What Mary didn't know. Journal of Philosophy, 83, 291–295.
- Jakobson, A. J., Fitzgerald, P. B., & Conduit, R. (2012). Induction of visual dream reports after transcranial direct current stimulation (tDCs) during Stage 2 sleep. *Journal of Sleep Research*, 21, 369–379.
- James, W. (1890). *The Principles of Psychology*. New York, NY: Holt.
- James, W. (1907). Pragmatism: A New Name for Some Old Ways of Thinking. New York, NY: Longmans, Green, and Co.
- Jibu, M., & Yasue, K. (1995). Quantum Brain Dynamics and Consciousness: An Introduction. Amsterdam, Netherlands: John Benjamins.
- Johanson, M., Revonsuo, A., Chaplin, J., & Wedlund, J.-E. (2003). Level and contents of consciousness in connection with partial epileptic seizures. *Epilepsy & Behavior*, 4, 279–285.
- Johanson, M., Valli, K., Revonsuo, A., & Wedlund, J.-E. (2008). Content analysis of subjective experiences in partial epileptic seizures. *Epilepsy & Behavior*, 12, 170–182.
- Johns, L. C., & McGuire, P. K. (1999). Verbal selfmonitoring and auditory hallucinations in schizophrenia. *Lancet*, 353, 469–470.
- Johnson, J. E., & Petzel, T. P. (1971). Temporal orientation and time estimation in chronic schizophrenics. *Journal of Clinical Psychology*, 27, 194–196.
- Kahn, D. (2007). Metacognition, recognition and reflection while dreaming. In D. Barrett & P. McNamara (Eds.), *The New Science of Dreaming* (Vol. 1, pp. 245–268). Westport, CT: Praeger.
- Kahn, D., & Hobson, J. A. (2005). State-dependent thinking: A comparison of waking and dreaming thought. *Consciousness and Cognition*, 14, 429–438.

- Kallio, S., Revonsuo, A., Lauerma, H., Hämäläinen, H., & Lang, H. (1999). The MMN amplitude increases in hypnosis: A case study. *Neuroreport*, 10, 3579–3582.
- Kallio, S., & Revonsuo, A. (2003). Hypnotic phenomena and altered states of consciousness: A multilevel framework of description and explanation. *Contemporary Hypnosis, 20,* 111–164.
- Kallio, S., & Revonsuo, A. (2005). Altering the state of the altered state debate: Reply to commentaries. *Contemporary Hypnosis*, 22, 46–55.
- Kallio, S., Revonsuo, A. & Lang, H. (2005). Can hypnosis and hypnotic hallucination change information processing in the brain – a case report. *Hypnos*, 32, 25–35.
- Kallio, S., Hyönä, J., Revonsuo, A., Sikka, P., & Nummenmaa, L. (2011). The existence of a hypnotic state revealed by eye movements. *PLoS ONE*, 6(10): e26374. doi:10.1371/journal.pone.0026374
- Kasai, K., Yamada, H., Kamio, S., Nakagome, K., Iwanami, A., Fukuda, M., Yumoto, M., Itoh, K., Koshida, I., Abe, O., & Kato, N. (2003). Neuromagnetic correlates of impaired automatic categorical perception of speech sounds in schizophrenia. *Schizophrenia Research*, 59, 159–172.
- Kay, S. R., Fiszbein, A., & Opler, L. A. (1987). The positive and negative syndrome scale (PANSS) for schizophrenia. *Schizophrenia Bulletin*, 13, 261–276.
- Kearse, L. A. Jr., Rosow, C., Zaslavsky, A., Connors, P., Dershwitz, M., & Denman, W. (1998). Bispectral analysis of the electroencephalogram predicts conscious processing of information during propofol sedation and hypnosis. *Anesthesiology*, *88*, 25–34.
- Kelly, E. R. (1882). *The Alternative: A Study in Psychology*. London, UK: Macmillan and Co.
- Kelly, P. H. (1998). Defective inhibition of dream event memory formation: A hypothesized mechanism in the onset and progression of symptoms of schizophrenia. *Brain Research Bulletin, 46*, 189–197.
- Kent, G., & Wahass, S. (1996). The content and characteristics of auditory hallucinations in Saudi Arabia and the UK: A cross-cultural comparison. Acta Psychiatrica Scandinavica, 94, 433–437.
- Kentridge, R. W., Nijboer, T. C. W., & Heywood, C. A. (2008). Attended but unseen: Visual attention is not sufficient for visual awareness. *Neuropsychologia*, 46, 864–869.
- Kiefer, M., & Spitzer, M. (2000). Time course of conscious and unconscious semantic brain activation. *Neuroreport*, 11, 2401–2407.
- Kim, J. (1992). 'Downword causation' in emergentism and nonreductive materialism. In A. Beckermann, H. Flohr, & J. Kim (Eds.), Emergence or Reduction? Essays on the Prospects of Nonreductive Physicalism (pp. 119–138). Berlin, Germany: Walter de Gruyter.

- Kim, C. Y., & Blake, R. (2005). Psychophysical magic: Rendering the visible 'invisible'. *Trends in Cognitive Sciences*, 9, 381–388.
- Kirsch, I. & Lynn, S. J. (1998). Dissociation theories of hypnosis. Psychological Bulletin, 123, 100–115.
- Kiverstein, J. (2010). Making sense of phenomenal unity: An intentionalist account of temporal experience. *Royal Institute of Philosophy Supplement*, 67, 155– 181.
- Kleinschmidt, A., Sterzer, P., & Rees, G. (2012). Variability of perceptual multistability: From brain state to individual trait. *Philosophical Transactions of the Royal Society B*, 367, 988–1000.
- Koivisto, M., Revonsuo, A., & Lehtonen, M. (2006). Independence of visual awareness from the scope of attention: An electrophysiological study. *Cerebral Cortex*, 16, 415–424.
- Koivisto, M., & Revonsuo, A. (2007). How meaning shapes seeing. *Psychological Science*, 18, 845–849.
- Koivisto, M., & Revonsuo, A. (2010). Event-related brain potential correlates of visual awareness. *Neuroscience* and Biobehavioral Reviews, 34, 922–934.
- Kornmeier, J., & Bach, M. (2012). Ambiguous figures what happens in the brain when perception changes but not the stimulus. *Frontiers in Human Neuroscience*, 6:51. doi:10.3389/fnhum.2012.00051
- Kosslyn, S. M., Thompson, W. L., Costantini-Ferrando, M. F., Alpert, N. M., & Spiegel, D. (2000). Hypnotic visual illusion alters color processing in the brain. *American Journal of Psychiatry*, 157, 1279–1284.
- Kouider, S., Sackur, J., & de Gardelle, V. (2012). Do we still need phenomenal consciousness? Comment on Block. *Trends in Cognitive Sciences*, 16, 140–141.
- Kramer, M. (2007). The Dream Experience: A Systematic Exploration. New York, NY: Routledge.
- Kramer, M., & Roth, T. (1973). A comparison of dream content in laboratory dream reports of schizophrenic and depressive patient groups. *Comprehensive Psychiatry*, 14, 325–329.
- Krippner, S. (1972). Altered states of consciousness. In J. White (Ed.), *The Highest State of Consciousness* (pp. 1–5). Garden City, NJ: Doubleday.
- Kühn, S., & Gallinat, J. (2012). Quantitative meta-analysis on state and trait aspects of auditory verbal hallucinations in schizophrenia. *Schizophrenia Bulletin*, 38, 779–786.
- Kuriyama, K., Uchiyama, M., Suzuki, H., Tagaya, H., Ozaki, A., Aritake, S., Shibui, K., Xin, T., Lan, L., Kamei, Y., & Takahashi, K. (2005). Diurnal fluctuation of time perception under 30-h sustained wakefulness. *Neuroscience Research*, 53, 123–128.
- LaBerge, S., & DeGracia, D. J. (2000). Varieties of lucid dreaming experience. In R. G. Kunzendorf & B. Wallace (Eds.), *Individual Differences in Conscious Experience* (pp. 269–307). Amsterdam, Netherlands: John Benjamins.

- LaBerge, S., & Gackenbach, J. (2000). Lucid dreaming. In E. Cardeña, S. J. Lynn, & S. Krippner (Eds.), Varieties of Anomalous Experience: Examining the Scientific Evidence (pp. 151–182). Washington, DC: American Psychological Association.
- La Mettrie, J. O. de (1748). L'homme machine. In R. A. Watson & M. Rybalka (Trans.). (1994). Man a Machine and Man a Plan. Indianapolis, IN: Hackett Publishing Company.
- Lamme, V. A. F. (2003). Why visual attention and awareness are different. *Trends in Cognitive Sciences*, 7, 12–18.
- Lamme, V. A. F. (2004). Separate neural definitions of visual consciousness and visual attention: A case for phenomenal awareness. *Neural Networks*, 17, 861– 872.
- Lamme, V. A. F. (2006). Towards a true neural stance on consciousness. *Trends in Cognitive Sciences*, 10, 494– 501.
- Langdon, R., Ward, P. B., & Coltheart, M. (2010). Reasoning anomalies associated with delusions in schizophrenia. Schizophrenia Bulletin, 36, 321–330.
- Långsjö, J. W., Alkire, M. T., Kaskinoro, K., Hayama, H., Maksimow, A., Kaisti, K. K., Aalto, S., Aantaa, R., Jääskeläinen, S. K., Revonsuo, A., & Scheinin, H. (2012). Returning from oblivion: Imaging the neural core of consciousness. *Journal of Neuroscience*, 32, 4935–4943.
- Lee, U., Mashour, G. A., Kim, S., Noh, G.-J., & Choi, B.-M. (2009). Propofol induction reduces the capacity for neural information integration: Implications for the mechanism of consciousness and general anesthesia. *Consciousness and Cognition*, 18, 56–64.
- Lenggenhager, B., Tadi, T., Metzinger, T., & Blanke, O. (2007). Video ergo sum: Manipulating bodily selfconsciousness. *Science*, 317, 1096–1099.
- Lennmarken, C., Bildfors, K., Enlund, G., Samuelsson, P., & Sandin, R. (2002). Victims of awareness. Acta Anaesthesiologica Scandinavica, 46, 229–231.
- Leslie, K. (2010). Dreaming during anesthesia. In G. A. Mashour (Ed.), *Consciousness, Awareness, and Anesthesia* (pp. 74–89). New York, NY: Cambridge University Press.
- Leslie, K., Myles, P. S., Forbes, A., Chan, M. T. V., Swallow, S. K., & Short, T. G. (2005). Dreaming during anaesthesia in patients at high risk of awareness. *Anaesthesia*, 60, 239–244.
- Leslie, K., Sleigh, J., Paech, M. J., Voss, L., Lim, C. W., & Sleigh, C. (2009). Dreaming and electroencephalographic changes during anesthesia maintained with propofol or desflurane. *Anesthesiology*, 111, 547–555.
- Levine, J. (1983). Materialism and qualia: The explanatory gap. *Pacific Philosophical Quarterly*, 64, 354–361.

- Lewis, A. (1932). The experience of time in mental disorder. Proceedings of the Royal Society of Medicine, 25, 611–620.
- Lewis, C. I. (1929). *Mind and the World Order: Outline of a Theory of Knowledge* (Reprint, 1991). New York, NY: Dover Publications.
- Lewis, D. (1966). An argument for the identity theory. *Journal of Philosophy, 63*, 17–25.
- Lewis, P. A., & Miall, R. C. (2003). Distinct systems for automatic and cognitively controlled time measurement: Evidence from neuroimaging. *Current Opinion in Neurobiology*, 13, 250–255.
- Li, W., Lopez, L., Osher, J., Howard, J. D., Parrish, T. B., & Gottfried, J. A. (2010). Right orbitofrontal cortex mediates conscious olfactory perception. *Psychological* Science, *21*, 1454–1463.
- Ludwig, A. M. (1966). Altered states of consciousness. Archives of General Psychiatry, 15, 225–234.
- Lutz, A., & Thompson, E. (2003). Neurophenomenology: Integrating subjective experience and brain dynamics in the neuroscience of consciousness. *Journal of Consciousness Studies*, 10, 31–52.
- Lynn, S. J., Malaktaris, A., Maxwell R., Mellinger, D. I., & van der Kloet, D. (2012). Do hypnosis and mindfulness practices inhabit a common domain? Implications for research, clinical practice, and forensic science. *Journal of Mind–Body Regulation*, 2, 12–26.
- Macknik, S. L. (2006). Visual masking approaches to visual awareness. *Progress in Brain Research*, 155, 177–215.
- Maksimow, A., Särkelä, M., Långsjö, J. W., Salmi, E., Kaisti, K. K., Yli-Hankala, A., Hinkka-Yli-Salomäki, S., Scheinin, H., & Jääskeläinen, S. K. (2006). Increase in high frequency EEG activity explains the poor performance of EEG spectral entropy monitor during S-ketamine anesthesia. *Clinical Neurophysiology*, 117, 1660–1668.
- Malcolm, N. (1959). *Dreaming*. London, UK: Routledge and Kegan Paul.
- Maquet, P., Ruby, P., Maudoux, A., Albouy, G., Sterpenich, V., Dang-Vu, T., Desseilles, M., Boly M., Perrin, F., Peigneux, P., & Laureys, S. (2005). Human cognition during REM sleep and the activity profile within frontal and parietal cortices: A reappraisal of functional neuroimaging data. *Progress in Brain Research*, 150, 219– 227.
- Marcel, A. (1993). Slippage in the unity of consciousness. In G. R. Bock & J. Marsh (Eds.), *Experimental and Theoretical Studies of Consciousness* (pp. 168–179). Chichester, UK: John Wiley and Sons.
- Marco, C. A. & Vaughan, J. (2005). Emergency management of agitation in schizophrenia. The American Journal of Emergency Medicine, 23, 767– 776.

- Marek, T., Fafrowicz, M., Golonka, K., Mojsa-Kaja, J., Oginska, H., Tucholska, K., Urbanik, A., Beldzik, E., & Domagalik, A. (2010). Diurnal patterns of activity of the orienting and executive attention neuronal networks in subjects performing a Stroop-like task: A functional magnetic resonance imaging study. *Chronobiology International*, 27, 945–958.
- Mashour, G. A. (Ed.). (2010). *Consciousness, Awareness,* and Anesthesia. New York, NY: Cambridge University Press.
- Mashour, G. A., & LaRock, E. (2008). Inverse zombies, anesthesia awareness, and the hard problem of unconsciousness. *Consciousness and Cognition*, 17, 1163–1168.
- Mashour, G. A., Esaki, R. K., Tremper, K. K., Glick, D. B., O'Connor, M., & Avidan, M. S. (2010). A novel classification instrument for intraoperative awareness events. Anesthesia and Analgesia, 110, 813–815.
- Mashour, G. A., Shanks, A., Tremper, K. K., Kheterpal, S., Turner, C. R., Ramachandran, S. K., Picton, P., Schueller, C., Morris, M., Vandervest, J. C., Lin, N., & Avidan, M. S. (2012). Prevention of intraoperative awareness with explicit recall in an unselected surgical population: A randomized comparative effectiveness trial. Anesthesiology, 117, 717–725.
- Massimini, M., Ferrarelli, F., Murphy, M., Huber, R., Riedner, B., Casarotto, S., & Tononi. G. (2010). Cortical reactivity and effective connectivity during REM sleep in humans. *Cognitive Neuroscience*, 1, 176–183.
- Massimini, M., Ferrarelli, F., Sarasso, S., & Tononi, G. (2012). Cortical mechanisms of loss of consciousness: Insight from TMS/EEG studies. Archives Italiennes de Biologie, 150, 44–55.
- Mathewson, K. E., Gratton, G., Fabiani, M., Beck, D. M., & Ro, T. (2009). To see or not to see: Prestimulus alpha phase predicts visual awareness. *Journal of Neuroscience*, 29, 2725–2732.
- Maury, A. (1861). *Le sommeil et les rêves* [*Sleep and Dreams*]. Paris, France: Didier.
- Mazzoni, G., Rotriquenz, E., Carvalho, C., Vannucci, M., Roberts, K., & Kirsch, I. (2009). Suggested visual hallucinations in and out of hypnosis. *Consciousness* and Cognition, 18, 494–499.
- Mazzoni, G., Venneri, A., McGeown, W. J., & Kirsch, I. (2013). Neuroimaging resolution of the altered state hypothesis. *Cortex*, 49, 400–410.
- McGinn, C. (1989). Can we solve the mind-body problem? *Mind*, 98, 349–366.
- McGinn, C. (1991). *The Problem of Consciousness*. Oxford, UK: Blackwell.
- Meier, B. (1993). Speech and thinking in dreams. In C. Cavallero & D. Foulkes (Eds.), *Dreaming as Cognition* (pp. 58–76). New York, NY: Harvester Wheatsheaf.
- Meyer, J. S., & Quenzer, L. F. (2005). *Psychopharmacology: Drugs, the Brain, and Behavior*. Sunderland, MA: Sinauer Associates.

- Metzinger, T. (2009). *The Ego Tunnel: The Science of the Mind and the Myth of the Self.* New York, NY: Basic Books.
- Metzner, R. (2010, January). What is an altered state of consciousness? *Brainwaving*. Retrieved from http://www.brainwaving.com/2010/01/22/what-is-an-altered-state-of-consciousness/
- Mill, J. S. (1843). A System of Logic. In J. M. Robson (Ed.). (1974). The Collected Works of John Stuart Mill (Vol. 7). Toronto, Canada: University of Toronto Press.
- Millisecond Software. (2008). Inquisit 3.0 [Computer software]. Seattle, WA: Author.
- Milner, A. D., & Goodale, M. A. (2008). Two visual systems re-viewed. *Neuropsychologia*, 46, 774–785.
- Moiseeva, N. I. (1975). The characteristics of EEG activity and the subjective estimation of time during dreams of different structure. *Electroencephalography and Clinical Neurophysiology*, 38, 569–577.
- Montgomery, J. D. (1979). Variations in perception of short time intervals during menstrual cycle. *Perceptual and Motor Skills*, 49, 940–942.
- Morita, T., Tsuneto, S., & Shima, Y. (2002). Definition of sedation for symptom relief: A systematic literature review and a proposal of operational criteria. *Journal* of Pain and Symptom Management, 24, 447–453.
- Morita, T., Fukui, T., Morofushi, M., & Tokura, H. (2007). Subjective time runs faster under the influence of bright rather than dim light conditions during the forenoon. *Physiology & Behavior*, 91, 42–45.
- Møller, A. R. (2003). Sensory Systems: Anatomy and Physiology. Amsterdam, Netherlands: Academic Press.
- Móró, L. (2010). Hallucinatory altered states of consciousness. *Phenomenology and the Cognitive Sciences*, 9, 241–252.
- Móró, L., Noreika, V., Revonsuo, A., & Kallio, S. (2011). Hypnotizability, sleepiness, and subjective experience. International Journal of Clinical and Experimental Hypnosis, 59, 211–224.
- Morofushi, M., Shinohara, K., & Kimura, F. (2001). Menstrual and circadian variations in time perception in healthy women and women with premenstrual syndrome. *Neuroscience Research*, *41*, 339–344.
- Morton, P. A. (2010). A Historical Introduction to the Philosophy of Mind (2nd ed.). Peterborough, Ontario: Broadview Press.
- Moutoussis, K. (2012). Asynchrony in visual consciousness and the possible involvement of attention. *Frontiers in Psychology*, 3:314, doi:10.3389/fpsyg.2012.00314
- Moutoussis, K., & Zeki, S. (2002). The relationship between cortical activation and perception investigated with invisible stimuli. *Proceedings of the National Academy of Sciences, 99*, 9527–9532.

- Muzur, A., Pace-Schott, E. F., & Hobson, J. A. (2002). The prefrontal cortex in sleep. *Trends in Cognitive Sciences*, 6, 475–481.
- Nagel, T. (1974). What is it like to be a bat? *Philosophical Review*, 83, 435–450.
- Naish, P. L. N. (2001). Hypnotic time perception: Busy beaver or tardy timekeeper? *Contemporary Hypnosis*, 18, 87–99.
- Naish, P. L. N. (2007). Time distortion and the nature of hypnosis and consciousness. In G. Jamieson (Ed.), *Hypnosis and Conscious states: The Cognitive-Neuroscience Perspective* (pp. 270–293). Oxford, UK: Oxford University Press.
- Nakajima, T., Uchiyama, M., Enomoto, T., Shibui, K., Ishibashi, K., Kanno, O., & Okawa, M. (1998). Human time production under constant routine. *Psychiatry* and clinical neurosciences, 52, 240–241.
- Ng, B. S., Schroeder, T., & Kayser, C. (2012). A precluding but not ensuring role of entrained low-frequency oscillations for auditory perception. *Journal of Neuroscience*, 32, 12268–12276.
- Nielsen, T. A. (2000). A review of mentation in REM and NREM sleep: "Covert" REM sleep as a possible reconciliation of two opposing models. *Behavioral and Brain Sciences*, 23, 851–866.
- Nielsen, T. A., Deslauriers, D., & Baylor, G. W. (1991). Emotions in dream and waking event reports. *Dreaming*, 1, 287–300.
- Nielsen, T. A., & Chénier, V. (1999). Variations in EEG coherence as an index of the affective content of dreams from REM sleep: Relationships with face imagery. *Brain and Cognition*, 41, 200–212.
- Nir, Y., & Tononi, G. (2010). Dreaming and the brain: From phenomenology to neurophysiology. *Trends in Cognitive Sciences*, 14, 88–100.
- Noë, A., & Thompson, E. (2004). Sorting out the neural basis of consciousness. *Journal of Consciousness Studies*, 11, 87–98.
- Noreika, V. (2011). Dreaming and waking experiences in schizophrenia: How should the (dis)continuity hypotheses be approached empirically? *Consciousness* and Cognition, 20, 349–352.
- Noreika, V., & Windt, J. M. (2008). On the importance of 19th century dream research: Progress in dream research between Aristotle's work on dreaming and the discovery of REM sleep. *Sleep Medicine*, 9, 919.
- Noreika, V., Valli, K., Lahtela, H., & Revonsuo, A. (2009). Early-night serial awakenings as a new paradigm for studies on NREM dreaming. *International Journal of Psychophysiology*, 74, 14–18.
- Noreika, V., Valli, K., Markkula, J., Seppälä, K., & Revonsuo, A. (2010a). Dream bizarreness and waking thought in schizophrenia. *Psychiatry Research*, 178, 562–564.

- Noreika, V., Windt, J. M., Lenggenhager, B., & Karim, A. A. (2010b). New perspectives for the study of lucid dreaming: From brain stimulation to philosophical theories of self-consciousness. *International Journal of Dream Research*, 3, 36–45.
- Noreika, V., Windt, J. M., Arstila, V., Falter, C. M., Kiverstein, J., & Revonsuo, A. (2010c). The subjective and the objective duration of static NREM sleep dreams. *International Journal of Dream Research*, 3(Suppl. 1), 6–7.
- Noreika, V., Jylhänkangas, L., Móró, L., Valli, K., Kaskinoro, K., Aantaa, R., Scheinin, H., & Revonsuo, A. (2011). Consciousness lost and found: Subjective experiences in an unresponsive state. *Brain and Cognition*, 77, 327–334.
- Noreika, V., Falter, C. M., Arstila, V., Wearden, J. H., & Kallio, S. (2012). Perception of short time scale intervals in a hypnotic virtuoso. *International Journal* of Clinical and Experimental Hypnosis, 60, 318–337.
- Noreika, V., Falter, C. M., & Rubia, K. (2013). Timing deficits in attention-deficit/hyperactivity disorder (ADHD): Evidence from neurocognitive and neuroimaging studies. *Neuropsychologia*, 51, 235– 266.
- Noreika, V., Falter, C. M., & Wagner, T. (2014). Variability of duration perception: From natural and induced alterations to psychiatric disorders. V. Arstila & D. Lloyd (Eds.), Subjective Time: The Philosophy, Psychology, and Neuroscience of Temporality (pp. 529-555). Cambridge, MA: MIT Press.
- Northoff, G. (2004). *Philosophy of the Brain: The Brain Problem*. Amsterdam, Netherlands: John Benjamins.
- Nummenmaa, L., Glerean, E., Hari, R., & Hietanen. J. K. (2014). Bodily maps of emotions. *Proceedings of the National Academy of Sciences*, 111, 646-651.
- Ohayon, M. M., Priest, R. G., Caulet, M., & Guilleminault, C. (1996). Hypnagogic and hypnopompic hallucinations: Pathological phenomena? *British Journal of Psychiatry*, 169, 459–467.
- Okuma, T., Sunami, Y., Fukuma, E., Takeo, S., & Motoike, M. (1970). Dream content study in chronic schizophrenics and normals by REMP-awakening technique. *Folia Psychiatrica et Neurologica Japonica*, 24, 151–162.
- O'Regan, J. K., & Noë, A. (2001). A sensorimotor account of vision and visual consciousness. *Behavioral and Brain Sciences*, 24, 939–973.
- Orlinsky, D. E. (1962). *Psychodynamic and Cognitive Correlates of Dream Recall* (Unpublished doctoral dissertation). University of Chicago, Chicago, IL.
- Overgaard, M., Rote, J., Mouridsen, K., & Ramsøy, T. Z. (2006). Is conscious perception gradual or dichotomous? A comparison of report methodologies during a visual task. *Consciousness and Cognition*, 15, 700–708.

- Owen, A. D., Coleman, M. R., Boly, M., Davis, M. H., Laureys, S., & Pickard, J. D. (2006). Detecting awareness in the vegetative state. *Science*, 313, 1402.
- Palmer, T. D., & Ramsey, A. K. (2012). The function of consciousness in multisensory integration. *Cognition*, 125, 353–364.
- Peremen, Z., Hilo, R., & Lamy, D. (2013). Visual consciousness and intertrial feature priming. *Journal* of Vision, 13(5):1. doi:10.1167/13.5.1.
- Pessoa, L., Japee, S., & Ungerleider, L. G. (2005). Visual awareness and the detection of fearful faces. *Emotion*, *5*, 243–247.
- Petitot, J., Varela, F. J., Pachoud, B., & Roy, J.-M. (Eds.) (1999). Naturalizing Phenomenology: Issues in Contemporary Phenomenology and Cognitive Science. Stanford, CA: Stanford University Press.
- Piolino, P., Desgranges, B., Belliard, S., Matuszewski, V., Lalevée, C., De La Sayette, V., & Eustache, F. (2003). Autobiographical memory and autonoetic consciousness: Triple dissociation in neurodegenerative diseases. *Brain*, 126, 2203–2219.
- Popper, K. R., & Eccles, J. C. (1977). *The Self and Its Brain*. New York, NY: Springer-Verlag.
- Raz, A. (2005). Attention and hypnosis: Neural substrates and genetic associations of two converging processes. International Journal of Clinical and Experimental Hypnosis, 53, 237–258.
- Raz, A., Shapiro, T., Fan, J., & Posner, M. I. (2002). Hypnotic suggestion and the modulation of Stroop interference. *Archives of General Psychiatry*, 59, 1155–1161.
- Raz, A., Schweizer, H. R., Zhu, H., & Bowles, E. N. (2009). Hypnotic dreams as a lens into hypnotic dynamics. International Journal of Clinical and Experimental Hypnosis, 58, 69–81.
- Rees, G., & Frith, C. (2007). Methodologies for identifying the neural correlates of consciousness. In M. Velmans & S. Schneider (Eds.), *The Blackwell Companion to Consciousness* (pp. 553–566). Malden, MA: Blackwell Publishing.
- Revonsuo, A. (1995a). Conscious and nonconscious control of action. *Behavioral and Brain Sciences, 18,* 265–266.
- Revonsuo, A. (1995b). Consciousness, dreams, and virtual realities. *Philosophical Psychology*, 8, 35–58.
- Revonsuo, A. (1999). Binding and the phenomenal unity of consciousness. *Consciousness and Cognition, 8,* 173–185.
- Revonsuo, A. (2000). Prospects for a scientific research program on consciousness. In T. Metzinger (Ed.), *Neural Correlates of Consciousness* (pp. 57–75). Cambridge, MA: MIT Press.
- Revonsuo, A. (2001). Can functional brain imaging discover consciousness in the brain? *Journal of Consciousness Studies*, 8, 3–23.

- Revonsuo, A. (2006). Inner presence. Consciousness as a Biological Phenomenon. Cambridge, MA: MIT Press.
- Revonsuo, A. (2010). Consciousness: The Science of Subjectivity. Hove, East Sussex: Psychology Press.
- Revonsuo, A., & Salmivalli, C. (1995). A content analysis of bizarre elements in dreams. *Dreaming*, 5, 169–187.
- Revonsuo, A., & Tarkko, K. (2002). Binding in dreams. The bizarreness of dream images and the unity of consciousness. *Journal of Consciousness Studies*, 9, 3– 24.
- Revonsuo, A., Kallio, S., & Sikka, P. (2009). What is an altered state of consciousness? *Philosophical Psychology*, 22, 187–204.
- Ricci, R., & Chatterjee, A. (2004). Sensory and response contributions to visual awareness in extinction. *Experimental Brain Research*, 157, 85–93.
- Ruby, P., Blochet, C., Eichenlaub, J.-B., Bertrand, O., Morlet, D., & Bidet-Caulet, A. (2013). Alpha reactivity to first names differs in subjects with high and low dream recall frequency. *Frontiers in Psychology*, 4:419. doi:10.3389/fpsyg.2013.00419
- Salminen-Vaparanta, N., Koivisto, M., Noreika, V., Vanni, S., & Revonsuo, A. (2012). Neuronavigated transcranial magnetic stimulation suggests that area V2 is necessary for visual awareness. *Neuropsychologia*, 50, 1621–1627.
- Sartorius, N., Shapiro, R., & Jablensky, A. (1974). The international pilot study of schizophrenia. *Schizophrenia Bulletin*, 1, 21–34.
- Schneider, G. (2010). Monitoring anesthetic depth. In G. A. Mashour (Ed.), *Consciousness, Awareness, and Anesthesia* (pp. 114–130). New York, NY: Cambridge University Press.
- Schönauer, M., Geisler, T., & Gais, S. (2014). Strengthening procedural memories by reactivation in sleep. *Journal of Cognitive Neuroscience*, 26, 143–153.
- Schooler, J. W. (2002). Re-representing consciousness: Dissociations between experience and metaconsciousness. *Trends in Cognitive Sciences*, 6, 339– 344.
- Schredl, M., Funkhouser, A. T., Cornu, C. M., Hirsbrunner, H. P., & Bahro M. (2001). Reliability in dream research: A methodological note. *Consciousness and Cognition*, 10, 496–502.
- Schredl, M. (2002). Questionnaires and diaries as research instruments in dream research: Methodological notes. *Dreaming*, 12, 17–26.
- Schredl, M. (2003). Continuity between waking and dreaming: A proposal for a mathematical model. *Sleep* and Hypnosis, 5, 26–39.
- Schredl, M., & Doll, E. (1997). Autogenic training and dream recall. *Perceptual and Motor Skills*, 84, 1305– 1306.

- Schredl, M. & Hofmann, F. (2003). Continuity between waking activities and dream activities. *Consciousness* and Cognition, 12, 298–308.
- Schredl, M., Atanasova, D., Hörmann, K., Maurer, J. T., Hummel, T., & Stuck, B. A. (2009). Information processing during sleep: The effect of olfactory stimuli on dream content and dream emotions. *Journal of Sleep Research*, 18, 285–290.
- Schredl, M., Brennecke, J., & Reinhard, I. (2013). Does training increase NREM dream recall? A pilot study. International Journal of Dream Research, 6, 54–58.
- Schubert, R., Blankenburg, F., Lemm, S., Villringer, A., & Curio, G. (2006). Now you feel it—now you don't: ERP correlates of somatosensory awareness. *Psychophysiology*, 43, 31–40.
- Schwartz, S., & Maquet, P. (2002). Sleep imaging and the neuro-psychological assessment of dreams. *Trends in Cognitive Sciences*, 6, 23–30.
- Searle, J. (1984). *Minds, Brains and Science*. London, UK: British Broadcasting Corporation.
- Searle, J. R. (1992). *The Rediscovery of the Mind*. Cambridge, MA: MIT Press.
- Searle, J. R. (2007). Biological naturalism. In M. Velmans & S. Schneider (Eds.), *The Blackwell Companion to Consciousness* (pp. 325–334). Malden, MA: Blackwell Publishing.
- Sebel, P. S., Bowdle, T. A., Ghoneim, M. M., Rampil, I. J., Padilla, R. E., Gan, T. J., & Domino, K. B. (2004). The incidence of awareness during anesthesia: A multicenter United States study. *Anesthesia and Analgesia*, 99, 833–839.
- Sergent, C., & Dehaene, S. (2004). Is consciousness a gradual phenomenon? Evidence for an all-or-none bifurcation during the attentional blink. *Psychological Science*, 15, 720–728.
- Shannon, C. E., & Weaver, E. (1949). The Mathematical Theory of Communication. Urbana: University of Illinois Press.
- Shapiro, A., Goodenough, D. R., & Gryler, R. B. (1963). Dream recall as a function of method of awakening. *Psychosomatic Medicine*, 25, 174–180.
- Siclari, F., LaRocque, J. J., Postle, B. R., & Tononi, G. (2013). Assessing sleep consciousness within subjects using a serial awakening paradigm. *Frontiers in Psychology*, 4:542. doi:10.3389/fpsyg.2013.00542
- Silverman, D., Masland, R. L., Saunders, M. G., & Schwab, R. S. (1970). Irreversible coma associated with electrocerebral silence. *Neurology*, 20, 525–533.
- Skrzypińska, D., & Szmigielska, B. (2013). What links schizophrenia and dreaming? Common phenomenological and neurobiological features of schizophrenia and REM sleep. Archives of Psychiatry and Psychotherapy, 2, 29–35.
- Smart, J. J. C. (1959). Sensations and brain processes. *Philosophical Review, 68*, 141–156.

- Smit, H. J., & Rogers, P. J. (2000). Effects of low doses of caffeine on cognitive performance, mood and thirst in low and higher caffeine consumers. *Psychopharmacology*, 152, 167–173.
- Spanos, N. P., & Coe, W. C. (1992). A social-psychological approach to hypnosis. In E. Fromm & M. Nash (Eds.), *Contemporary Hypnosis Research* (pp. 102–130). New York, NY: Guilford Press.
- Spiegel, D. (2003). Negative and positive visual hypnotic hallucinations: Attending inside and out. International Journal of Clinical and Experimental Hypnosis, 51, 130–146.
- St. Jean, R., & MacLeod, C. (1983). Hypnosis, absorption, and time perception. *Journal of Abnormal Psychology*, 92, 81–86.
- Stompe, T., Ritter, K., Ortwein-Swoboda, G., Schmid-Siegel, B., Zitterl, W., Strobl, R., & Schanda, H. (2003). Anxiety and hostility in the manifest dreams of schizophrenic patients. *Journal of Nervous and Mental Disease*, 191, 806–812.
- Stumbrys, T., Erlacher, D., & Schredl, M. (2013). Testing the involvement of the prefrontal cortex in lucid dreaming: A tDCS study. *Consciousness and Cognition*, 22, 1214–1222.
- Tart, C. T. (1972). States of consciousness and statespecific sciences. Science, 176, 1203–1210.
- Tart, C. T. (1975). *States of Consciousness*. New York, NY: E. P. Dutton.
- Tart, C. T. (1988). From spontaneous event to lucidity. A review of attempts to consciously control nocturnal dreaming. In J. Gackenbach, & S. LaBerge (Eds.), *Conscious Mind, Sleeping Brain: Perspectives on Lucid Dreaming* (pp. 67–104). New York, NY: Plenum Press.
- Thompson, E., & Varela, F. J. (2001). Radical embodiment: Neural dynamics and consciousness. *Trends in Cognitive Sciences*, 5, 418–425.
- Titchener, E. B. (1902). Experimental Psychology: A Manual of Laboratory Practice (Vol. 1). New York, NY: MacMillan & Co., Ltd.
- Tonner, P. H., & Bein, B. (2006). Classic electroencephalographic parameters: Median frequency, spectral edge frequency etc. Best Practice & Research Clinical Anaesthesiology, 20, 147–159.
- Tononi, G. (2001). Information measures for conscious experience. Archives Italiennes de Biologie, 139, 367– 371.
- Tononi, G. (2004). An information integration theory of consciousness. BMC Neuroscience, 5:42. doi:10.1186/1471-2202-5-42
- Tononi, G. (2007). The information integration theory of consciousness. In M. Velmans & S. Schneider (Eds.), *The Blackwell Companion to Consciousness* (pp. 287– 299). Malden, MA: Blackwell Publishing.
- Tononi, G. (2008). Consciousness as integrated information: A provisional manifesto. *Biological Bulletin*, 215, 216–242.

- Tononi, G. (2009). Sleep and consciousness. In M. Klockars & T. Porkka-Heiskanen (Eds.), *The Many Aspects of Sleep* (Acta Gyllenbergiana VIII, pp. 63–72). Helsinki, Finland: Signe and Ane Gyllenberg Foundation.
- Tononi, G. (2010). Information integration: Its relevance to brain function and consciousness. Archives Italiennes de Biologie, 148, 299–322.
- Tononi, G. (2012). Integrated information theory of consciousness: An updated account. Archives Italiennes de Biologie, 150, 293-329.
- Tononi, G., & Sporns, O. (2003). Measuring information integration. BMC Neuroscience, 4:31. doi:10.1186/1471-2202-4-31
- Tononi, G., & Massimini, M. (2008). Why does consciousness fade in early sleep? Annals of the New York Academy of Sciences, 1129, 330–334.
- Treisman, A. (1996). The binding problem. *Current Opinion in Neurobiology, 6,* 171–178.
- Tulving, E. (1985). Memory and consciousness. *Canadian Psychology*, *26*, 1–12.
- Turner, J. (2010). Ontological pluralism. Journal of Philosophy, 107, 5–34.
- Turner, J. (2012). Logic and ontological pluralism. *Journal* of *Philosophical Logic*, 41, 419–448.
- Tysk, L. (1983a). Time estimation by healthy subjects and schizophrenic patients: A methodological study. *Perceptual and Motor Skills*, 56, 983–988.
- Tysk, L. (1983b). Estimation of time and the subclassification of schizophrenic disorders. *Perceptual and Motor Skills*, *57*, 911–918.
- Urban, B. W., & Bleckwenn, M. (2002). Concepts and correlations relevant to general anaesthesia. *British Journal of Anaesthesia*, 89, 3–16.
- Vaitl, D., Birbaumer, N., Gruzelier, J., Jamieson, G. A., Kotchoubey, B., Kübler, A., Lehmann, D., Miltner, W. H., Ott, U., Pütz, P., Sammer, G., Strauch, I., Strehl, U., Wackermann, J., & Weiss, T. (2005). Psychobiology of altered states of consciousness. *Psychological Bulletin*, 131, 98–127.
- Valli, K., Revonsuo, A., Pälkäs, O., Ismail, K. H., Ali, K. J., & Punamäki, R.-L. (2005). The threat simulation theory of the evolutionary function of dreaming: Evidence from dreams of traumatized children. *Consciousness* and Cognition, 14, 188–218.
- Varela, F. J. (1996). Neurophenomenology: A methodological remedy for the hard problem. *Journal* of Consciousness Studies, 3, 330–349.
- Velmans, M. (2007). An epistemology for the study of consciousness. In M. Velmans & S. Schneider (Eds.), *The Blackwell Companion to Consciousness* (pp. 711– 725). Malden, MA: Blackwell Publishing.
- Verhagen, J. V. (2007). The neurocognitive bases of human multimodal food perception: Consciousness. Brain Research Reviews, 53, 271–286.

- Viana, M., Sprenger, T., Andelova, M., & Goadsby, P. J. (2013). The typical duration of migraine aura: A systematic review. *Cephalalgia*, 33, 483–490.
- Vita, M. G., Batocchi, A. P., Dittoni, S., Losurdo, A., Cianfoni, A., Stefanini, M. C., Vollono, C., Della Marca, G., & Mariotti, P. (2008). Visual hallucinations and pontine demyelination in a child: Possible REM dissociation? *Journal of Clinical Sleep Medicine*, 4, 588–590.
- Vogeley, K., & Kupke, C. (2007). Disturbances of time consciousness from a phenomenological and a neuroscientific perspective. *Schizophrenia Bulletin, 33*, 157–165.
- Voss, U., Holzmann, R., Tuin, I., & Hobson, J. A. (2009). Lucid dreaming: A state of consciousness with features of both waking and non-lucid dreaming. *Sleep*, 32, 1191–1200.
- Voss, U., Schermelleh-Engel, K., Windt, J., Frenzel, C., & Hobson, A. (2013). Measuring consciousness in dreams: The lucidity and consciousness in dreams scale. *Consciousness and Cognition*, 22, 8–21.
- Vul, E., MacLeod, D. I. A. (2006). Contingent aftereffects distinguish conscious and preconscious color processing. *Nature Neuroscience*, 9, 873–874.
- Wagstaff, G. F. (1981). *Hypnosis, Compliance and Belief.* New York, NY: St. Martin's Press.
- Wahl, O. F., & Sieg, D. (1980). Time estimation among schizophrenics. *Perceptual and Motor Skills*, 50, 535– 541.
- Walsh, V. (2003). A theory of magnitude: Common cortical metrics of time, space and quantity. *Trends in Cognitive Sciences*, 7, 483–488.
- Wartofsky, M. (1979). *Models: Representation and the Scientific Understanding*. Dordrecht, Netherlands: D. Reidel.
- Watson, J. B. (1913). Psychology as the behaviorist views it. Psychological Review, 20, 158–177.
- Wearden, J. H. (1992). Temporal generalization in humans. Journal of Experimental Psychology: Animal Behavior Processes, 18, 134–144.
- Wearden, J. H., Edwards, H., Fakhri, M., & Percival, A. (1998). Why "sounds are judged longer than lights": Application of a model of the internal clock in humans. Quarterly Journal of Experimental Psychology. B, Comparative and Physiological Psychology, 51, 97–120.
- Weber, E. H. (1834). *De Pulsu, Resorptione, Auditu et Tactu. Annotationes Anatomicae et Physiologicae.* Leipzig, Germany: Koehler.
- Weitzenhoffer, A. M., & Hilgard, E. R. (1962). Stanford Hypnotic Susceptibility Scale: Form C. Palo Alto, CA: Consulting Psychologists Press.
- Wertheimer, M. (1923). Untersuchungen zur Lehre von der Gestalt II. Psycologische Forschung, 4, 301–350.

- Wheeler, M. A., Stuss, D. T., & Tulving, E. (1997). Toward a theory of episodic memory: The frontal lobes and autonoetic consciousness. *Psychological Bulletin*, 121, 331–354.
- Williamson, A. M., Feyer, A.-M., Mattick, R. P., Friswell, R., & Finlay-Brown, S. (2001). Developing measures of fatigue using an alcohol comparison to validate the effects of fatigue on performance. Accident Analysis & Prevention, 33, 313–326.
- Wilson, T. D. (2003). Knowing when to ask: Introspection and the adaptive unconscious. *Journal of Consciousness Studies*, 10, 131–140.
- Windt, J. M. (2010). The immersive spatiotemporal hallucination model of dreaming. *Phenomenology and* the Cognitive Sciences, 9, 295–316.
- Windt, J. M. (2012). Dreaming: A Conceptual Framework for Philosophy of Mind and Empirical Research (Doctoral dissertation). Johannes Gutenberg-Universität Mainz, Mainz, Germany.
- Windt, J. M., & Metzinger, T. (2007). The philosophy of dreaming and self-consciousness: What happens to the experiential subject during the dream state? In D. Barrett & P. McNamara (Eds.), *The New Science of Dreaming* (Vol. 3, pp. 193–247). Westport, CT: Praeger Perspectives/Greenwood Press.
- Windt, J. M., & Noreika, V. (2011). How to integrate dreaming into a general theory of consciousness — A critical review of existing positions and suggestions for future research. *Consciousness and Cognition*, 20, 1091–1107.
- Winget, C., & Kramer, M. (1979). *Dimensions of Dreams*. Gainesville, FL: University Presses of Florida.
- Wittmann, M., Leland, D. S., Churan, J., & Paulus, M. P. (2007). Impaired time perception and motor timing in stimulant-dependent subjects. *Drug and Alcohol Dependence*, 90, 183–192.
- Wright, L. (1973). Functions. *Philosophical Review, 82*, 139–168.
- Wundt, W. (1862). Die Geschwindigkeit des Gedankens [The swiftness of thought]. Die Gartenlaube, 17, 263– 265.
- Xuan, B., Zhang, D., He, S., & Chen, X. (2007). Larger stimuli are judged to last longer. *Journal of Vision*, 7, 1–5.
- Zanasi, M., Calisti, F., Di Lorenzo, G., Valerio, G., & Siracusano, A. (2011a). Oneiric activity in schizophrenia: Textual analysis of dream reports. *Consciousness and Cognition*, 20, 337–348.
- Zanasi, M., Calisti, F., Di Lorenzo, G., Valerio, G., & Siracusano, A. (2011b). Reply to Valdas Noreika's commentary on Zanasi, M., Calisti, F., Di Lorenzo, G., Valerio, G., & Siracusano, A. (2011). Oneiric activity in schizophrenia: Textual analysis of dream reports. *Consciousness and Cognition*, 20, 353–354.
- Zeki, S. (1991). Cerebral akinetopsia (visual motion blindness). A review. *Brain, 114*, 811–824.

- Zeki, S. (1993). A Vision of the Brain. Oxford, UK: Blackwell Scientific Publications.
- Zeki, S. (2003). The disunity of consciousness. *Trends in Cognitive Sciences*, 7, 214–218.
- Zeki, S. (2007). A theory of micro-consciousness. In M. Velmans & S. Schneider (Eds.), *The Blackwell Companion to Consciousness* (pp. 580–588). Malden, MA: Blackwell Publishing.
- Zeki, S., & ffytche, D. H. (1998). The Riddoch syndrome: Insights into the neurobiology of conscious vision. *Brain*, 121, 25–45.
- Zeki, S., & Bartels, A. (1999). Toward a theory of visual consciousness. *Consciousness and Cognition*, *8*, 225–259.
- Zipke, M., Ehri, L. C., & Cairns, H. S. (2009). Using semantic ambiguity instruction to improve third graders' metalinguistic awareness and reading comprehension: An experimental study. *Reading Research Quarterly*, 44, 300–321.